

Investigating the impact of various Corporate Governance Mechanisms on the Accounting Performance of Non-Financial Firms Listed on the Amman Stock Exchange

By **Ammar Z. Ramadan**

Supervised by: Prof. Eleri Jones

Submitted in partial fulfilment for the award of the degree of Doctor Business administration

University of Wales Trinity Saint David

2020



Table of	Contents	Page #
Chapter 1	Introduction	1
1.1	Background of the study	1
1.2	Theoretical framework adopted in the study	4
1.3	Research aim	4
1.4	Research objectives	5
1.5	Research questions	5
1.6	Rationale of the study	8
1.7	Definitions of the key terms	9
1.8	Thesis outline	9
1.9	summary	11
Chapter 2	Literature review	12
2.1	Jordanian economic sectors & Jordanian regulators	12
2.1.1	Industry and services sector in Jordan	17
2.1.2	Telecom and IT	18
2.1.3	Energy	18
2.1.4	Transport	19
2.1.5	Media and advertising	20
2.2	Regulatory institution & CG development in Jordan	21
2.2.1	The Jordanian capital market	25
2.2.2	Jordan securities commission (JSC)	27
2.2.3	Amman stock exchange (ASE)	28
2.2.4	Securities depository centre (SDC)	29
2.2.5	Corporate governance development in Jordan	30
2.3 2.4	Definition of corporate governance Theoretical framework	32 34
2.4.1		3 4 35
2.4.1	Agency theory Stewardship theory	33 40
2.4.2	Resource dependence theory	40
2.4.5	Conceptual framework	44
2.6	CG issues in developing countries	46
2.7	CG International principles & practices	47
2.8	CG and firm performance	50
2.8.1	Board of directors	52
2.8.2	Board of directors' sub-committees	55
2.8.3	Board size	60
2.8.4	CEO duality	65
2.8.5	Non-executive directors	68
2.9	summary	72
Chapter 3	Research methodology	74
3.1	Research philosophy	74
3.2	Research methods	76
3.2.1	Panel data	78
322	Pooled regression	70



3.2.3	Fixed effect	79
3.2.4	Random effect	80
3.2.5	GLS estimator	82
3.2.6	Sample	84
3.3	Variables and measurement of variables	87
3.3.1	Performance variables	87
3.3.2	Control variables	92
3.3.2.1	Sales growth (SGH)	92
3.3.2.2	Firm size (FSE)	93
3.3.2.3	Capital expenditure (CAPX)	94
3.3.2.4	Leverage (LEE)	95
3.3.2.5	Research & Development expenditure (R&D)	96
3.3.2.6	Liquidity (LIY)	97
3.3.3	Corporate governance variables	98
3.3.3.1	Board size (BSE)	98
3.3.3.2	CEO duality (CEOD)	99
3.3.3.3	Non-executive directors (NEDs)	100
3.3.3.4	Presence of board sub-committees (BSCS)	101
3.3.4	Corporate governance index (CGI)	102
3.4	Data analysis	107
3.5	summary	110
Chapter 4	Analysis and discussion	112
4.1	Results and discussion	112
4.1.1	Descriptive statistics	113
4.1.2	Specification test results	116
4.1.3	Correlation	118
4.1.4	Results of multiple regression analysis	121
4.1.4.1	Results of control variables	123
4.1.4.1.1	Sales growth (SGH)	123
4.1.4.1.2	Firm size (FSE)	124
4.1.4.1.3	Capital expenditure (CAPX)	125
4.1.4.1.4	Leverage (LEE)	126
4.1.4.1.5	Research and development (R&D)	127
4.1.4.1.6	Liquidity (LIY)	129
4.1.4.2	Results of corporate governance variables	130
4.1.4.2.1	Board size (BSE)	130
4.1.4.2.2	Non-executive directors (NEDs)	132
4.1.4.2.3	CEO duality (CEOD)	134
4.1.4.2.4	Establishment of board committees	139
4.1.4.3	Results for multiple regression for CGI	141
4.2	summarv	146



Chapter 5	Conclusion and recommendations	147
5.1	Introduction	147
5.2	Research findings	147
5.3	Limitations of the study	151
5.4	Research contribution	152
5.5	Further studies	153
5.6	summary	156
References		157
Appendix 1	Tables	204
Table 1	Descriptive statistics	204
Table 2	Correlation	205
Table 3	Multiple regression results for ROA	207
Table 4	Multiple regression results for ROE	208
Table 5	Multiple regression results for TOO	209



DECLARATION SHEET

This sheet MUST be signed and included within the thesis

This work has not previously been accepted in substance for any degree and is not being concurrently submitted in candidature for any degree.

Signed Ammar Z. Ramadan (student)

Date 15/07/2019

STATEMENT 1

This thesis is the result of my own investigations, except where otherwise stated. Where correction services have been used the extent and nature of the correction is clearly marked in a footnote(s). Other sources are acknowledged by footnotes giving explicit references. A bibliography is appended.

Signed Ammar Z. Ramada (student)

Date 15/07/2019

STATEMENT 2

I hereby give consent for my thesis, if accepted, to be available for photocopying and for inter-library loan, and for the title and summary to be made available to outside organisations.

Signed Ammar Z. Ramadan (student)

Date 15/07/2019

STATEMENT 3

I hereby give consent for my thesis, if accepted, to be available for deposit in the University's digital repository.

Signed Ammar Z. Ramadan (student)

Date 15/07/2019



Dedication

This thesis is dedicated to the spring of tenderness of my precious mother, who always urged me, supported me and wished me to get a doctorate degree In accounting major. But with a sad heart, she died before I graduated.

My father, who always encouraged me to complete doctoral degree - accounting major - where he was supporting me financially and morally.

My brothers and sisters for their continued support and encouragement in the last four years to complete different phases of the research.

My precious wife, and my dear sons and daughters for their continued support and encouragement, and for providing me with the appropriate opportunities to complete the thesis.

To all my friends and everyone who supported me to finish this thesis.

Ammar Z. Ramadan



List of Abbreviations

AC Audit Committee

AFM Amman Financial Market

ASE Amman Stock Exchange

BOD Board of Directors

BSCS Board Sub-Committees

BSE Board Size

CAPX Capital Expenditure

CBJ Central Bank of Jordan

CEO Chief Executive Officer

CEOD CEO Duality

CG Corporate Governance

CGI Corporate Governance Index

CR Current Ratio

EMH Efficient Market Hypothesis

ECGI European Corporate Governance Institute

FDI Foreign Direct Investment

FTA Free Trade Agreement

FEAS Federation of Euro-Asian Stock Exchanges

FSE Firm Size

GDP Gross Domestic Product

GLS Generalised Least Squares

HKJ Hashemite Kingdom of Jordan

IASs International Accounting Standards

IFRSs International Financial Reporting Standards

JACPA Jordanian Association of Certified Public Accounting

JCGC Jordanian Corporate Governance Code

JIB Jordanian Investment Board

JSC Jordanian Securities Commission

LEE Leverage
LIY Liquidity

MENA Middle East and North Africa region

NC Nomination Committee



NEDs Non-Executive Directors

OECD Organisation for Economic Cooperation and Development

QIZ Qualifying Industrial Zones

TOQ Tobin's Q

RC Remuneration Committee

ROA Return on Assets
ROE Return on Equity

R&D Research and Development Expenditure

SD Standard Deviation

SDC Security Depository Centre

SGH Sales Growth

UNDP United Nations Development Programme

WFE World Federation of Exchanges



List of Tables

Table 2-1	Comparative analysis of economic indicators
Table 2-2	Jordan corporate governance compliance with OECD principles
Table 2-3	Ownership and board structure for Jordanian companies
Table 2-4	ASE market performance
Table 2-5	Conceptual model
Table 2-6	Features of insider and outsider corporate governance systems
Table 3-1	Non-financial sectors of Amman stock exchange
Table 3-2	Firms related to different industries
Table 3-3	Composition of corporate governance index
Table 4-1	Descriptive statistics for ROA, ROE and TOQ
Table 4-2	Descriptive statistics for control variables
Table 4-3	Descriptive statistics for corporate governance variables
Table 4-4	Panel model tests
Table 4-5	Specification test results
Table 4-6	Correlation for dependent variables
Table 4-7	Correlation for control variables
Table 4-8	Correlation for corporate governance variables
Table 4-9	F-statistics and R-square
Table 4-10	Impact of sales growth on firm performance
Table 4-11	Impact of firm size on firm performance
Table 4-12	Impact of capital expenditure on firm performance
Table 4-13	Impact of leverage on firm performance
Table 4-14	Impact of R&D expenditure on firm performance
Table 4-15	Impact of liquidity on firm performance
Table 4-16	Impact of board size on firm performance
Table 4-17	Impact of non-executive directors on firm performance
Table 4-18	Impact of CEO duality on firm performance
Table 4-19	Impact of board sub-committees on firm performance
Table 4-20	Multiple regression by measuring firm performance on ROA
Table 4-21	Multiple regression by measuring firm performance on ROE
Table 4-22	Multiple regression by measuring firm performance on TOQ



Abstract

There is a long history of corporate scandals and failure in accounting performance around the globe as a result of weak government mechanisms overseeing companies due to the absence of clear government laws and legislation governing the relationship between shareholders (owners, investors) and executive managers (internal management) in a firm. Weak government mechanisms reinforce the greed of the dominant and influential groups within the various companies. This weakens the accounting and financial performance of a firm and leads to it being unable to generate sufficient profits to satisfy its shareholders, as well as being unable to attract new investors or encourage them to invest within the firm.

As a result, various researchers are investigating the impact of corporate governance (CG) mechanisms on firm performance. The majority of the research regarding corporate governance and its impact on firm performance has been conducted in developed countries, especially in the US and UK. Relatively less evidence is available in the Middle East, particularly in Jordan. Thus, the aim of the current study is to investigate the impact of various corporate governance mechanisms on firm performance for Jordanian non-financial companies listed on the Amman Stock Exchange during the period 2012 – 2018. Different hypotheses have been generated through agency theory to explore the relationship between the corporate governance mechanisms and the accounting performance of Jordanian nonfinancial companies. Agency theory is employed because it explains the agency problems between the corporate manages and shareholders, which have a negative impact on the value maximisation objective. It has been claimed that various features of the board of directors' act as effective corporate governance mechanisms for solving the agency problem between corporate managers and shareholders.

This study uses multiple regression panel data analysis to analyse the data.

A fixed effect model is used to investigate the impact of corporate governance mechanisms on firm performance. Secondary data is collected



for 95 Jordanian non-financial companies listed on the Amman Stock Exchange. The data is collected from three different sources: the Amman Stock Exchange website, the annual reports of the selected companies and from the DataStream database.

A mixed set of results is observed from the empirical investigation. The findings of the study reveal a significant negative impact of board size on firm performance, which indicates that large board size tends to be inefficient due to poor coordination and communication. The findings of the study also reveal that non-executive directors (NEDs) have a significant negative impact on firm performance. The findings are not in line with the hypothesis generated from agency theory, which states that NEDs play a vital role on the board by monitoring the firm's performance and providing valuable suggestions to the executive directors, as they are experts in their field and have years of experience. However, CEO duality has a significant positive impact on firm performance, which reveals that Jordanian firms perform better if the roles of CEO and Chairman are performed by one individual. These findings are also inconsistent with the Jordanian CG code, which emphasises the separate of the role CEO and Chairman for implementing the strict check and improve the balance of the board of directors. The findings indicate that three board committees (audit, remuneration and nomination) have a significant positive impact on the firm performance. These findings are consistent with the empirical results and Jordanian CG code because establishment of board committees tends to streamline various business operations, which has a positive impact on firm performance.

A corporate governance (CG) index has been generated based on the Jordanian corporate governance code requirements. Compliance with the corporate governance index has been checked for the Jordanian non-financial firms. The findings indicate that the corporate governance index also has a positive and significant impact on firm performance.



The findings of the current study will help the Jordanian regulators to amend the current Jordanian corporate governance code because the impact of different CG mechanisms on the firm performance has been checked through real data and statistical tests. Further, the findings can assist the Jordanian corporate boards and managers to improve their companies' corporate governance system by implementing the significant CG mechanisms that are explored by this study.



Chapter 1 Introduction

1.1 Background of the study

This study investigated the impact of corporate governance (CG) on firm performance in Jordan. CG provides a mechanism to streamline the operating affairs of the business and help it to achieve the objective of shareholder wealth maximisation. Previous studies conducted by researchers shed light on the significance of the CG mechanisms and highlighted how CG mechanisms assisted in overcoming the agency problem faced by public limited companies around the globe (Shabbir and Padgett, 2008; El-Faitouri, 2014; Akbar et al., 2016; Alqatan and Hussainey, 2019; Al-Ahdal et al., 2020). The agency problem is the conflict of interest between the corporate managers and corporate investors in the case of public limited companies. It has resulted in the bankruptcy of various companies, such as Enron (2001), WorldCom (2003) and recently Carillion (2018), Patisserie Valerie (2019) and Thomas Cook (2019) (Kollewe, 2019).

Such conflicts can be overcome through effective management of the firm's affairs under the umbrella of corporate governance. Effective corporate governance ensures proper utilisation of the firm's resources, which increases investor confidence (Denis and Denis, 1994). The mechanism of corporate governance helps firms to manage internally and externally; first, it assists them in managing internal matters, and, secondly, it helps them respond to the external environment and external conditions (Gregory et al., 2005). Various researchers such as Rwegasira (2000), Nam et al. (2004) and Akbar et al. (2016) have reported in their studies that the mechanism of corporate governance protects a firm's assets from misuse by managers, which ensures safe and efficient decision making. Such measures help to utilise resources properly, which enhances the performance of the firm.

Although corporate governance phenomena are under investigation globally, the majority of the work has been carried out in Western countries which are developed nations, especially in the UK and US, and the Eastern point of



view has not been greatly investigated. Only a few studies have been conducted that explore the impact of CG mechanisms and firm performance in Middle Eastern countries. Middle Eastern countries have a unique culture (the majority of the population are Muslim and follow the Islamic guidelines in daily life) and economic factors (financial markets (capital and money market) are not developed, and non-existence of bond market) that are different from Western developed economies. Researchers have adopted those CG mechanisms in their research that are implemented by the regulators in those countries. This study is focused on the CG mechanisms that are implemented by Jordanian regulators. Another unique feature of this study is that the impact of various CG mechanisms has been explored in a standalone and collective manner by generating a CG index based on the requirements of the Jordanian CG code.

During the 90s the Jordanian government made efforts to motivate investors to invest in Jordan. The government provided incentives to Jordanian citizens based overseas and to international investors to invest in Jordan by launching the Jordanian Economic Growth Stimulus Programme to achieve sustainable development and enhance the competitiveness of the economy (Kabariti, 2019). This initiative helped Jordan to integrate its economy with the world at the global level. Under this initiative, the capital markets of Jordan were liberalised, and CG mechanisms were restructured. In addition to this, an institutional framework was also developed, and new institutes were established. It helped to strengthen the regulatory environment and ensured transparency and accountability in financial matters. These institutes were the Securities Depository Centre (SDC), Jordanian Securities Commission (JSC) and Amman Stock Exchange (ASE). As a result of the above governmental efforts, the country's economic conditions improved, and the Jordanian capital markets flourished. The gross domestic product (GDP) of Jordan increased to 2.3% and per capita GDP rose to 2,909 Jordanian Dollars in 2018, which was the highest level it reached during 2014 to 2018 (Central Bank of Jordan, 2020).



The main objective of this study is to investigate the impact of CG mechanisms on the firm performance of Jordanian non-financial listed companies on the ASE. Secondary data is collected for 95 non-financial companies for a period of six years ranging from 2012 to 2018. The study has adopted the CG mechanisms that were proposed by the Jordanian CG code 2011. Jordanian companies began to implement the requirements of the CG code 2011 in 2012. For this reason, the data is collected from 2012 to 2018. The focus of this study is also on non-financial companies because the reporting style of financial companies is different from that of non-financial companies. Various statistical tools, such as descriptive analysis, correlation and multiple regression analysis, are used to explore the impact of CG mechanisms on firm performance. This study will examine the effects of various CG mechanisms in a standalone and combined manner for the Jordanian companies, and it is the first study of its kind in the context of Jordan.

The study in hand is keen to identify and understand the existing literature about variables such as CG and organisational performance, particularly in Jordan. There is a visible literature gap that will be identified and answered by the current study to enhance the existing knowledge in this field of study. This study is equally essential for Jordanian corporate managers because they can implement significant CG mechanisms in improving their firms' performance. Empirical results have shown that the implementation of CG mechanisms is helping corporate managers to streamline their operating activities and enhancing accountability and transparency mechanisms. Effective operations, sound accountability and a transparent system enable firms to improve their performance (Mkheimer, 2018).

This study is beneficial for the regulators because regulators around the globe are continuously updating the CG mechanisms (European Corporate Governance Institute (ECGI), 2020). Jordanian regulators developed the first CG code for Jordanian banks in 2006, a CG code for Jordanian companies in 2007 and the CG code was updated in 2011. The outcomes of this study will help the regulators to understand the real impact of various CG



mechanisms that are proposed by the Jordanian regulators through CG codes.

1.2 Theoretical framework adopted in the study

The theoretical framework is an essential and integral part of any given research design. It is quite necessary to understand the theoretical stances of every piece of research so that its visible contribution to the specific field of literature can be highlighted explicitly. The study in hand is based on agency theory, and the research questions and hypotheses are generated based on agency conflict. Agency theory deals with the conflicts of interest between principals and agents. In public companies, the shareholders are the principals who hire agents so that the agents can achieve the shareholders' wealth maximisation objectives. The managers of the company are the agents who need to work to achieve the shareholders' wealth maximisation objectives. Unfortunately, however, they work to achieve their own personal goals; this conflict of interest between managers and shareholders is known as an agency problem.

Therefore, the aim of this study, based on agency theory and agency conflict, is to explain the relationship between the CG mechanisms and their impact on the firm's financial performance. It is worth mentioning that agency theory and its associated hypotheses are tested, and this theory testing enriches the subject knowledge accordingly. Further details are provided in Chapter 2 of this thesis.

1.3 Research aim

This research aims to investigate the impact of corporate governance mechanisms on the firm performance for the non-financial Jordanian firms listed on the Amman Stock Exchange.



1.4 Research objectives

The research aim is divided into the following objectives:

- To conduct a critical literature review in the field of corporate governance and firm performance
- To collect the secondary data for various CG mechanisms and firm performance for Jordanian non-financial firms from 2012 – 2018
- To analyse the data by using various statistical tools, such as descriptive analysis, correlation and multiple regression analysis
- To present the findings of the study and compare and contrast the results of the study with previous studies
- To provide a brief conclusion and recommendations in light of the analysis of this study

1.5 Research questions

The board of directors is a critical CG mechanism, and previous research has identified the independent director as a useful feature/character of the board of directors (Gillan, 2006; Liu and Fong, 2010; Alqatan and Jussainey, 2019). Aggarwal et al. (2011) and Akbar et al. (2016) have stated that the prime duty of board members is to protect the interests of investors and ensure that agents (managers) are performing their duties to maximise the wealth of shareholders. Thus, the framework of corporate governance provides strategic guidance which helps to monitor the performance of firms through the board (Organisation for Economic Cooperation and Development (OECD), 2004). The board of directors (BOD) also plays its role to minimise the conflicts between the agent and principals. Accordingly, the BOD keeps an eye on top management to ensure alignment between the managers and



shareholders. This alignment will help to reduce the conflict and increase the firm's performance, ultimately protecting the interests of shareholders (Braendle, 2019). The BOD has a significant impact on the firm's performance, so the first research question of this study is:

Research question 1

How does the BOD (such as size, Chief Executive Officer (CEO) duality and Non-executive directors (NEDs)) show their impact on performance of a company in Jordan's non-financial listed firms

Board sub-committees can also help to increase the performance of the BOD (Vafeas, 1999). Companies usually tend to establish various committees which monitor the affairs related to audit (audit committee), remuneration (remuneration committee), and hiring for vacant posts at the board level (nomination committee). These sub-committees assist the board of directors (Chhaochharia and Grinstein, 2009; El-Faitouri, 2014). Anderson and Bizjak (2003) have stated that the remuneration committee helps to determine the compensation of executives working within the firm.

Generally, executives and independent directors of listed firms are nominated and controlled by the board (Vafeas, 1999). It can be very costly to decide on a collective nomination if the firm has a very disbursed ownership structure. Thus, the board undertakes the responsibility to nominate and appoint executives and independent directors for the firm. Selection of an appropriate person for an executive position is crucial for a firm and this job is performed by the nomination committee, which ensures the quality of nominations. Additionally, this nomination committee works as a separate unit from the managerial team, which ensures the selection of the most appropriate person for an executive role (Vafeas, 1999). On the other hand, if responsibility is delegated to the managerial team for the selection of executives, it will not ensure quality. It is possible that the executive directors would select another executive director based on their personal affiliations, which could protect their interests rather than those of the shareholders.



Offering an attractive remuneration package to the executives will motivate them to act in the best interests of the shareholders and to align the interests of the shareholders with those of the managers. Anderson and Bizjak (2003) have stated that the board of directors delegates the task of designing wages and compensation to the remuneration committee. This committee also provides its consultancy role to the board regarding the payment to the executive directors. Vafeas (2003) and Akbar et al. (2016) have stated that the financial scandal around the globe at the start of the 21st century highlighted the need to implement regulations for strengthening the role of the compensation committee.

Young et al. (2008) and El-Faitouri (2014) have stated that the audit committee is a crucial part of a corporate governance system which helps to restrict managers from being involved in financial crimes while managing their firms' affairs. The fundamental function of an audit committee is to review the financial activities and evaluation of financial statements of the firm with the collaboration of internal and external auditors. Additionally, the audit committee also reviews internal controls to evaluate the performance of companies (Aggarwal et al., 2005; Yawson, 2006; Wulf, 2007). Thus, considering the significance of various board committees, the second research question of this study is:

Research question 2

Do the board committees have an impact on the firm performance and the organisation efficiency of non-financial listed Jordanian companies?

It is essential to mention that the committees such as audit, remuneration and nomination play a pivotal role in the performance of the firm. Furthermore, the investigation in the current study focuses on examining the impact of the level of compliance on corporate performance by adopting two approaches, by using a standalone corporate governance mechanism to



investigate their effects on firm performance and by using the governance index. Various researchers have taken the index-based approach to quantify the impact of CG index on firm performance and found significant results (Adams, 2003; Padgett and Shabbir, 2008; El-Faitouri, 2014; Akbar et al., 2016). Hence, in line with considering the significance of using the governance index, so the third research question of this study is:

Research question 3

Does the corporate governance index (CGI) have an impact on the performance of the non-financial listed Jordanian organisations?

1.6 Rationale of the study

Firstly, this study might be helpful to enhance our understanding of reducing agency conflict by using different corporate governance mechanisms in the context of Jordanian non-financial companies. This investigation is dealing with non-financial listed firms and will provide essential insights to understand the behaviour of non-financial firms. These findings will enhance the existing body of knowledge in the field of CG and performance. Secondly, Jordan is a developing country. Thus, outcomes of this study could also help other developing nations with a similar background to Jordan in various aspects such as related to political, cultural, environmental and economic conditions, especially in the context of the Middle East and North Africa (MENA) region.

Thirdly, a significant improvement has been witnessed regarding the economic environment of Jordan and increasing listing of firms on the ASE from 2000 to 2016. Thus, this study seeks to provide help to understand firm performance through improvement in the economic environment. Hence, assessing the impact of foreign investment in Jordanian firms and its comparison with the other comparable MENA markets can also provide essential insights. This liberalisation has attracted a bulk of foreign investment in Jordan, and its capital market has a high ratio of attracting



foreign investment around the globe (OECD, 2006). All the above-cited facts highlight the significance of the study, which motivated the researcher to carry out this study.

1.7 Definitions of the key terms

This section provides the definitions of the key terms used in the current study.

CEO duality is the hiring of one individual person for the two positions at the board level, such as CEO and Chairman of the company.

Executive directors (EDs) are the directors who manage the operating activities of the company, and they are responsible for achieving the shareholders' wealth maximisation objective.

Non-executive directors (NEDs) are the directors who are responsible for improving the governance system of the company by creating a sound monitoring system.

Corporate governance index is the index generated by the researcher based on the corporate governance requirements of the Jordanian corporate governance code 2011.

1.8 Thesis outline

Chapter one introduces the current study and explains its significance, along with providing the specific research questions with their potential insights.

Chapter two reviews the background of the Jordanian economy and its industrial sectors. It also provides the details of the country's financial and regulatory system. This chapter focuses on the theoretical background, and provides a review of existing literature in the area of corporate governance mechanisms and their features and firm performance.



Chapter three deals with the data and its extraction sources used in this study. Sample selection criteria and details of variables from performance and corporate governance perspectives are also given in this chapter. The philosophical orientation, the methodology and statistical tests applied in this study will also be discussed in this chapter.

Chapter four deals with the results of this study. The results of the descriptive statistics are provided in the first section. The statistical inference has been drawn based on findings of the research. Results are effectively presented in tables so that interpretation can be made easily about the study interventions. The results are discussed and compared and contrasted with those of previous studies.

Chapter five of this thesis provides a conclusion along with different recommendations. The study findings are discussed in specific alignment to research questions set in the study. Research limitations and potential avenues for future research studies are also provided.



1.9 summary

This chapter divided into eight sections, the first section shed light on the background of the study, the second section deals with the theoretical framework adopted in the study, as well as this section explained that the study in hand is based on agency theory, and the research questions and hypotheses are generated based on agency conflict, the third section deals with the aim of research which is to investigate the impact of corporate governance mechanisms on the firm performance for the non-financial Jordanian firms listed on the Amman Stock Exchange.

The fourth section clarify that the research aim divided into five objectives which are conduct a critical literature review in the field of corporate governance and firm performance; collect the secondary data for various CG mechanisms and firm performance for Jordanian non-financial firms from 2012 – 2018; analyse the data by using various statistical tools, such as descriptive analysis, correlation and multiple regression analysis; present the findings of the study and compare and contrast the results of the study with previous studies; provide a brief conclusion and recommendations in light of the analysis of this study.

The fifth section of this chapter deals with the research questions that the researcher developed to the study in hand, while the sixth section deals with the rational of the study. whereas the seventh section deals with the definitions of the key terms, the final section of this chapter deals with the Thesis outline.



Chapter 2 Literature review

2.1 Jordanian economic sectors and Jordanian regulators

Being a small country, Jordan has limited natural resources from minerals with the majority of the country being a desert landscape. Jordan emerged as Transjordan in 1921, after the Arab revolt against the Ottoman Empire. Jordan's geographical position gives the country substantial strategic importance from political and economic perspectives. Jordan is situated at the crossroads of what the world's communities such as Christians, Jews and Muslims call the Holy Land. Knowles (2005) stated that The Hashemite Kingdom of Jordan (HKJ) has proved to be one of the world's most vulnerable countries to external political, economic and security events. From 1940, Jordan has remained under pressure due to a large variety of crises. Occupation of coastal parts of Palestine, the Iran-Iraq War, America's invasion of Iraq, the Arab Spring in 2011, all these factors have had an impact on the country's political and economic situation. The recent Syrian crisis has also hampered the steady pace of Jordan's economic growth.

Citizens of Jordan have an upper-middle income level, and their average per capita income was 4,330.3 USD in 2019 (The World Bank, 2020). Almost 80% of the population lives in urban areas. Thirty-eight per cent of the population is comprised of young people aged 14 or less, suggesting that Jordan is the youngest nation (The World Bank, 2018). The major exports of country are potash and phosphates. This scarcity is not limited with natural resources relevant to minerals; also there are limited water resources in Jordan which make Jordan low production related to agriculture products. Such scarcity of natural resources put pressure on Jordan status and due to these factors Jordan is placed at number 04 in poorest country according to water resources. Due to such hardships because of natural resources, almost 75% of jobs in Jordan are due to services sector. Thus 70% of Jordan's Gross Domestic Product (GDP) is produced through service sector (World Bank, 2018).



Jordan faces many economic challenges, such as poverty, high unemployment and high inflation rate. The budget deficit is also a significant challenge for the Jordanian government. To tackle such issues and problems, King Abdullah II Bin Al Hussein of Jordan has introduced many economic reforms. These reforms are opening of the trade regime, privatisation of state-owned companies and reducing subsidies on various government-sponsored projects. Such efforts have attracted foreign direct investment, ultimately creating a pool of jobs for the Jordan people. Despite these efforts from the government of Jordan, global economic pressures have also depressed the country's economic growth (World Bank, 2018). However, government agencies are keen to address these issues and are continually working to resolve the problems.

In order to improve the living standards of the poor and middle class, the government introduced two economic relief packages during 2011. These two measures were considered as a budgetary supplement. However, during that time, Jordan's financial position was disturbed because the country had to use more expensive fuel for electricity generation as the supply of natural gas from Egypt was cut off due to attacks on the gas pipeline. Despite such pressures, the government was able to attract foreign direct investment (FDI) and aid from Gulf countries, which helped to ease the budgetary expenditures. However, the budget deficit remained at 10% of the Gross Domestic Product (GDP) despite such help (United Nations Development Programme (UNDP), 2017). This budgetary deficit was managed through assistance from the neighbouring countries during 2012. Also in 2012, the financial crisis hit globally; however, limited exposure to foreign capital markets helped Jordan to survive this global financial crisis. Jordan is not isolated from its neighbouring nations, and it has strong ties with its Gulf neighbours through trade, remittances, FDI and tourism. The Jordanian government is trying to make the utmost efforts to devise the best policies for achieving economic benefits. As a result of these efforts, the reliance on fuel-based electricity generation was shifted to nuclear power generation to tackle power shortfalls (OECD, 2013).



By following methods of regional integration, Jordan has made itself vulnerable to various volatilities pertaining to political, economic and social aspects. After 2010, like other countries of the Middle East, Jordan also absorbed the aftershocks of various waves of disturbances, and such issues have had significant impacts on the country's economic status. The stronger citizen voice and accountability has also put pressure on Jordan's government. A general increase in the prices of commodities, an increase in import bills, a decrease in foreign remittances, and low tourism activities were observed in 2011 (OECD, 2013). Disturbance in the adjacent countries has also negatively affected Jordan's economy. Interruption of the natural gas supply from Egypt caused an additional \$2.4 billion to the taxpayers because the government used costly fuels to generate electricity in 2012. However, such financial blows have not hampered Jordan's good governance policies, and the government of Jordan is spending heavily on the human development of its nation.

Jordan is spending a high proportion of its development budget on education, health, retirement benefits and social safety benefits; the country is spending more than 25% of its GDP consistently on the above sectors. In addition to this, Jordan provides gender parity to individuals across various public policies. During 2003, Jordan introduced a comprehensive modernisation programme across the country, aiming to change the education system to develop a knowledge-based economy (World Bank, 2018). Such policies pertaining to expenditure and improvement in the education system have helped Jordan to increase school enrolment, which is high compared to similar income-level nations. However, it is becoming difficult for the government to maintain the pace of development in the health and education sector because of the growing population rates.

During the last decade, Jordan has introduced critical reforms in various sectors, and it has made notable progress in the areas of education and health. This situation portrays a good picture of Jordan's Social, Education and Health systems. However, sound economic policies are needed to overcome the negative impacts of global crises. In addition to the above



factors, the high unemployment rate and over-dependence on remittances are still significant economic problems for Jordan.

In 2011, the country went through the "Arab Spring revolutions" movement which forced the government to introduce political reforms along with economic governance. Jordan's government tackled this movement through gradual improvements in the judicial system to improve public accountability; these changes were approved through parliament by constitutional amendments. To improve the economic situation, the government introduced austerity measures to ensure transparency and accountability in public matters. The government also focused on improvements in public sector development by introducing reforms in budget and debt management. Such measures have ensured a better economic future for Jordan at a gradual pace; additionally, these measures have provided a supportive regional and external environment (The World Bank, 2018). However, it is perceived that, in the future, Jordan will have to face fierce competition from its neighbouring countries because these countries are also attracting foreign investment. If the government overcomes the above challenge, then it would provide employment opportunities to the masses and thus reduce poverty.

Although businessmen across the world consider Jordan as the best place to invest in this region, due to the security and safety that Jordan enjoys unlike most of the region countries, nevertheless investment opportunities are not being utilized properly. The government provides equal opportunities to local and foreign investors due to the Investment Promotion Law of 1995. This law also provides guarantees against expropriation. The Investment Promotion Law of 1995 provides some incentives to investors such as exemption from customs duties and even tax holidays. In addition to this, investors can also obtain the benefits pertaining to the unrestricted transfer of capital and profits, as explained by the International Labour Organization (2013).

Jordan's government has introduced disclosure regulations which bind public companies to disclose various financial information. These disclosure regulations ensure financial security. This disclosure is helping to develop



the country's financial sector. These guidelines have been prepared to keep in mind the cultural dimensions of Jordan. Companies are bound to disclose board members' ownership and top management's salaries. Any concealment of a fact by companies results in penalties. At present, various penalties have been imposed on firms which have violated the disclosure regulations, ranging from charging of fines to stopping the firms from circulating on the ASE.

The popular movement that most of Jordan's neighbouring countries are witnessing, due to the poor economic and political environment in these countries makes the influx of refugee and emigration moving towards Jordan is high. This influx hampers the delivery of public services in the country and disturbs labour market conditions, posing a considerable threat to the economy. Despite these challenges, which are not easy to resolve apparently, Jordan has managed to maintain a steady pace of economic growth with visible presence. As per the report of The World Bank (2016) covering the economic perspectives of countries (Economic Monitoring Report), Jordan's GDP growth rate was 3.5% in 2015 and was expected to touch 3.9 % in 2016.

Although Jordan has limited resources as compared to its neighbouring nations, the contribution of services and goods exports in GDP in the year 2014 was more than 40%. Other significant contributors to the GDP are foreign remittances, Gulf subsidiaries and development grants that contribute up to 19% in GDP. The following shows the ease of doing business for the country with other allies explicitly (The World Bank, 2018).



Table 2-1 Comparative analysis of economic indicators

Indicator (rank)	Jordan	MENA lowest	MENA best	Regional average	Global best
Starting a Business	106	173 (Kuwait)	32 (Oman)	119.05	1 (New Zealand)
Dealing with construction permits	109	157 (West Bank and Gaza)	4 (UAE)	91.48	1 (New Zealand)
Getting Electricity	48	172 (Djibouti)	4 (UAE)	97.19	1 (Korea, Rep.)
Registering Property	96	168 (Djibouti)	11(UAE)	91.71	1 (New Zealand)
Getting Credit	185	185 (Jordan)	82 (Saudi Arabia)	138.14	1 (New Zealand)
Protecting Minority investors	165	187 (Sudan)	9 (UAE)	127.62	1 (New Zealand*)
Paying Taxes	79	186 (Mauritania)	1 (UAE)	83.48	1 (UAE)
Trading across borders	50	184 (Sudan)	50 (Jordan)	128.67	1 (10 Economies*)
Enforcing Contracts	124	184 (Djibouti)	25 (UAE)	115.76	1 (Korea, Rep.)
Resolving Insolvency	142	161 (Syria)	58 (Tunisia)	129.43	1 (Finland)

Source Almustafa (2017)

2.1.1 Industry and services sector in Jordan

Jordan's industrial sector is comprised of manufacturing, construction, mining, and power. These four Jordanian industrial sectors contributed 26% to its GDP in 2014. During that time, almost 21% of Jordan's labour force was engaged in the industrial sector. The major industrial products of the industrial sector are potash, phosphates, pharmaceuticals, cement, clothing and fertilisers. The construction industry is the top industrial sector. This growing trend in the construction sector is supported by the US-Jordan Free Trade Agreement (FTA). This agreement between the US and Jordan indicates that Jordan is one of the most important markets for the US. However, the US has also engaged with other MENA countries through other trade agreements. The Agadir Agreement, with the EU, is one of a number of such contracts which have the potential to reap benefits for Arab countries. Jordan is also offering other attractive options to investors to start their business in the country. Jordan has skilled labour, which also attracts foreign investors to start their business in the country, and ultimately it will promote economic development (OECD, 2013).



Although the availability of natural resources in Jordan is limited, the country's abundant reserves of potash and phosphates are contributing to the industrial production. Both potash and phosphates are used in the production of fertilisers and were worth almost \$1bn in 2018. (Central Bank of Jordan, 2020). Furthermore, the textile sector is also a significant contributor to Jordan's economic development, and its net worth was almost \$1.19 billion in 2017 (Central Bank of Jordan, 2020).

2.1.2 Telecom and IT

The telecommunication sector is also a major contributor to Jordan's GDP, and its share is 13.5% of the country's total GDP. During 2001, Jordan introduced telecom liberalisation which made the country's Information Technology (IT) sector the most developed in the Arab region. Three telecommunication companies are operating in Jordan and offering their services to the country's mobile phone users. Zain (Kuwait based) holds 39% of the market share, Orange (owned by France Telecoms) holds 36% and Umniah holds 25%. The competitive environment of the mobile sector has resulted in a price war to retain the market share (Zu'ubi, 2013).

The Information Technology Association (int@j) was established to prepare Jordan for the implementation of the latest technological advancements in the field of telecommunication. This association helps the country's telecommunication sector to compete with that of other countries. The improvements in the IT and telecommunication sector have played their part in enhancing Jordan's GDP by almost 14%. Keeping in mind the growth potential in the IT sector, the Jordanian government is keen to provide training opportunities for the workforce to reduce the unemployment rate (Zu'ubi, 2013).

2.1.3 Energy

Like other developing nations, Jordan also faces challenges in the energy sector. The demand for energy remains a major concern for the government.



Lack of domestic investment in the energy sector has attracted \$14 billion of foreign investment to meet the country's energy needs. This investment programme was aimed to reduce dependency on fuel imports by 2020. Additionally, it was aimed to increase the proportion of nuclear energy production by 60% by the end of 2035 (The World Bank, 2018).

Jordan's neighbours are rich in petroleum resources, but Jordan imports oil from its neighbours to meet its own energy needs. Jordan had offered oil supply routes to Iraq and, in return, purchased discounted oil from Iraq, but, after the US invasion of Iraq, the oil supply to Jordan was disrupted. After this invasion, Jordan shifted its primary and secondary sources through the port of Al Aqabah by using tankers. Currently, Saudi Arabia is the primary supplier of oil to Jordan, but oil is also imported from Kuwait and the United Arab Emirates (UAE) (Jordanian Ministry of Energy and Mineral Resources, 2013).

It was estimated that Jordan has average natural gas reserves of approximately 6 billion cm³, but it is believed that natural gas reserves are much higher than this estimation. During 2003, 390 million cm³ of natural gas was extracted and consumed in Jordan because it is a domestic source of energy. For this purpose, the primary source is the Risha gas field. This gas field is situated in the eastern part of the country. Arab Gas line is the primary pipeline which provides natural gas to Jordan from Egypt and passes underwater to Al Aqabah. This pipeline starts from Egypt and is linked to two major power stations. It provides approximately one billion cm³ of natural gas per year to Jordan (Jordanian Ministry of Energy and Mineral Resources, 2013).

2.1.4 Transport

The transportation sector is considered vital for the economic development of Jordan. This sector provides almost 10% of the country's total GDP, which was approximately \$2.14 billion in 2007. These statistics indicate the service-based economy of Jordan, and thus it is believed that the transportation



sector holds a vital position in the country's economy. Due to such potential in this sector, the government of Jordan planned to revolutionise the transportation sector in 2008. For this purpose, a new national transport strategy was formulated and implemented to improve and modernise the transportation sector. Privatisation was also part of this policy. The uncertain situation and security crisis in Iraq promoted Jordan's transportation sector. Large numbers of tourists also use different modes of transportation. Jordan's government followed realistic measures to improve the transportation sector by relocating Aqaba's main port and developing a railway system. But the increasing trend in fuel prices hampered the development of the transportation sector by increasing the operational costs of transport. Despite these challenges, uncertainty in fuel prices also tends to offer a lucrative incentive to investors, motivating them to invest in transportation (Jordan Ministry of Transport, 2013).

2.1.5 Media and advertising

In the modern contemporary world, there is no doubt that the media plays a very critical role in setting the image (face) of any social entity such as Jordan. Media is considered to be one of the essential components of society, and has an impact on the people as well as the community. Although the state mainly influences Jordan's media, this sector has witnessed significant privatisation and liberalisation during the last few years. Data shows that, on average, almost \$280 million was spent on advertisement in Jordan's media in 2017. Jordan Television (JTV) remains the sole broadcaster in the country, but other mediums are also available for the public to access information, such as blogs, websites and news portals. These growing mediums have the potential to attract investments.

Jordan's Media and Advertising Industry expanded by 260% between 2000 and 2017 (Jordan Media and Advertising, 2018). Jordan Media and Advertising (2013) stated that the majority of spending on advertisements was made by the telecoms sector during 2017, with 205 contributions. After the telecom sector, the banking and finance sector had the highest spending



on advertising (12%), whilst the services, real estate and automotive industries spent 11%, 8% and 5% respectively. In order to obtain the best possible output from the media and advertising sector, Jordan needs a vocational training facility to explore its emerging potential in the specific industries.

2.2 Regulatory institution and CG development in Jordan

Jordan is representative of the mixed system of the Ottoman Empire (based on French law), British common law, and Islamic law. In the case of corporate governance mechanism, Jordan developed its corporate governance mechanism on an 'insider-oriented' corporate governance system. However, Jordan's government has been improving the country's legal and organisational structure. For this purpose, the government has introduced critical economic legislation, such as the Company Law (1997) and the New Securities Law (2002). These two frameworks provide necessary legislation regarding corporate governance roles and concepts to improve and enhance the country's investment climate.

The first decade of the 21st century has seen tremendous changes around the globe regarding liberalisation and globalisation. Being an emerging economy, Jordan's government assessed the country's worth and adopted various policies to cope with the emerging trends in the MENA region.

Various vital players implement economic reforms in Jordan, and out of these the corporate compliance authority is the key player. This authority follows company law to ensure economic reforms and developments by implementing corporate governance provisions. The Central Bank of Jordan (CBJ) issued a Director's Handbook of Corporate Governance in 2004. In addition to this, the Central Bank of Jordan is continuously exerting efforts to improve the country's corporate governance mechanism. Further, the Central Bank of Jordan is also performing to enhance the banking system through a Corporate Governance Code (Al-Amarneh, 2014).



Developments in the mechanism of corporate governance are directly linked with the shareholders' rights. "In Jordan, shareholders enjoy a considerable right in terms of access to secure methods of registering a property, the strength of legal rights and legislation enforcement, getting credit from local banks and financial institutions, sell or transfer shares, obtaining relevant information or actions on a timely basis, and other firm activities like voting and board elections (World Bank, 2015)". The weighted average CG compliance score for the Jordanian CG code was 3.91. The code was compared with the subcategories of the OECD principles that are presented in Table 2-2 below.

5 4 3.67 3.67 4 3.5 3 2 1 0 Rights **Treatment** D&T Role **Board** Source Almustafa (2017)

Table 2-2 Jordan CG compliance with OECD principles

An assessment was made by Shanikat and Abbadi (2011) regarding the corporate governance of Jordan and its compliance with OECD principles. Key findings of their research indicate some interesting points, which are as follows:

- Shareholders can take part in the decision-making process, except in significant cases such as the sale of assets.
- Practically, shareholders are not treated as equal to managers, although regulatory authorities take action to control insider trading.
- Stakeholders have active roles and rights in the implementation of the corporate governance system because their rights are protected by Jordan's company laws and regulations.



- Disclosure and transparency were observed to a large extent because Jordan's reporting standards follow the International Financial Reporting Standards (IFRSs) and International Accounting Standards (IASs).
- Boards largely fulfil their responsibilities, as these are extensively defined by law and regulation.

In the MENA region, the financial development of Jordan is comparatively high in comparison with other countries (see the below table). According to Creane et al. (2007), the government of Jordan has imposed minimal restrictions on the property rights. In addition to this, in Jordan the banking system is well developed and efficient as compared with other countries of the region. During the last few years, the government of Jordan has increased the efficiency of the banking sector through prudential measures and regulations. This has been ensured through the establishment of a corporate governance code for the banking sector that was implemented by the Jordanian banking industry in 2007.

For shareholding companies, the code of corporate governance was issued in 2008. Additionally, this code was supposed to help business firms to secure their rights. This code provides guidelines regarding various corporate governance issues such as board size and composition along with ownership structure. In an attempt to update guidelines regarding the corporate governance mechanism, the government of Jordan with the collaboration of market participants and the Ministry of Industry and Trade developed a new code in 2011. This new code specifies distributions of rights and responsibilities of various entities related to firms, such as shareholders, BODs, and top management. This new code also provides guidelines regarding effective decision-making processes in firms. Table 2-3 illustrates ownership and BOD structure from 2008 to 2014 of non-financial listed firms.



Table 2-3 Ownership and board structure for Jordanian companies

	Obs	Mean	Median	SD	Min	Max
Government ownership	790	0.01	0.00	0.07	0.00	0.60
Family ownership	790	0.24	0.13	0.27	0.00	0.99
Foreign ownership	790	0.03	0.00	0.13	0.00	0.99
Institutional ownership	790	0.36	0.33	0.27	0.00	0.99
Concentration (Total)	790	0.61	0.64	0.23	0.06	0.99
Board size	790	8.22	8.00	2.38	3.00	14.00
Board independence	790	0.56	0.57	0.16	0.14	0.91
CEO-duality	790	0.20	0.00	0.40	0.00	1.00

Source Almustafa (2017)

Economic crises during 2008 have put many economies under pressure and caused the failure of many firms around the globe. Like other countries, Jordan has also attempted to enhance its financial security and improve its economic environment in response to these economic crises. Despite the war and instability in neighbouring countries, Jordan has maintained its steady economic growth during the last few years. This sustained economic growth has witnessed an increased volume of trade and market capitalisation, which has increased the number of firms listed on the Amman Stock Exchange (ASE) (ASE, 2012). This state of affairs shows that Jordan has been working continuously from 1990 to up till now to improve its economic liberalisation and corporate governance.

Various measures have been adopted by the Jordanian government in this regard to improve financial stability and create a better economic environment. One such measure was the establishment of the ASE to trade public securities. An additional criterion was the establishment of the Securities Depository Centre (SDC) to protect investors; further, the Jordan Security Commission (JSC) was established to keep control over the equity market. These measures have helped to ensure disclosure and transparency, which also protects investors. Although such practices have made Jordan an advanced country in terms of corporate governance in MENA countries, both the World Bank and the IMF (2004) assessed that

corporate governance in Jordan was insufficient and it still needs to improvement (ROSC, 2004). These suggestions also endorsed the findings of Glaeser et al. (2001), regarding economic liberalization that tend to produce short-term economic growth in case of developing economies.

2.2.1 Jordanian capital market

Amman Financial Market (AFM) was set up in 1976, but it was restructured in 1994 to enhance its size, liquidity, disclosure of reliable information and transparency. Later, the AFM was replaced with three primary entities. These entities were the Amman Stock Exchange (ASE), Jordan Security Commission (JSC) and Securities Depository Centre (SDC). The ASE remained a well-developed and credible regulated market for many years with a market capitalisation of approximately US\$5 billion. In addition to this, the ASE is an outstanding emerging stock exchange (Central Bank of Jordan, 2015).

Such developments in Jordan's financial system brought significant positive changes in the country, and the number of listed firms increased dramatically from 1999 to 2015. In 1999, only 152 firms were listed on the ASE whereas in 2015 there were 228 firms listed. The market capitalisation also increased remarkably during the same period. The average market capitalisation of listed companies in MENA countries was far less in comparison to that of Jordan (109%) during 1990 – 2014; for example, it was 34.34 in Egypt and 60.85 in Israel. Various research studies have shown that effective financial markets ensure economic growth and development in both a developed and a developing economy (Bayraktar, 2014).

The Jordanian capital market has played a vital role in the country's economic development. The country's financial sector has dominated the market along with the industrial sector that holds 41% market capitalisation. The service and insurance sector proportions are much less, at 8% and 2% respectively. The above statistics highlight that the ASE has performed far better than other financial markets in the MENA region. Statistical data has



also pointed out that the investment of the public sector in various ASE companies is 18%, investment of individuals is 30%, and rest is invested by private institutional investors (Bayraktar, 2014)

Jordan's government is attracting overseas investors by allowing them to invest in different projects and they can start a public company by holding a 50% stake in it. Jordan has exempted the dividends income and capital gains from tax so overseas investors can invest in Jordanian financial markets. Such lucrative options have provided a competitive edge to the companies listed on the ASE. The three regulatory bodies, ASE, JSC and SDC, are currently involved in monitoring, regulating and supervising the country's financial affairs. With the implementation of the Securities Law (2002) and Accounting and Auditing Standards, the regulators have controlled the economic issues of the Jordanian corporate sector (Almustafa, 2017).

Table 2-4 ASE market performance

Year	No. of listed Co.	Market Cap. (JD Million)	Market Cap. as a% of the GDP	General Weighted	Value Traded (JD million)
	C0.	(3D WIIIIOII)	a70 of the GD1	Price Index (point)	(3D minion)
2005	201	26,667.10	326.6	8191.5	16,871.00
2006	227	21,078.20	233.9	5518.1	14,209.90
2007	245	29,214.20	289	7519.3	12,348.10
2008	262	25,406.30	216.7	6243.1	20,318.00
2009	272	22,526.90	149.6	5520.1	9,665.30
2010	277	21,858.20	122.7	5318	6,690.00
2011	247	19,272.70	102.7	4648.4	2,850.20
2012	243	19,141.50	93.5	4593.6	1,978.80
2013	240	18,233.49	83	4336.7	3,027.30
2014	236	18,082.62	75.8	4237.6	2,263.40
2015	228	17,984.67	70.7	4229.9	3,417.10

Source Almustafa (2017)



2.2.2 Jordan Securities Commission (JSC)

The JSC was established in 1997 to regulate the capital market. This commission was established under the Securities Law No. 23 and it reports directly to the country's prime minister. The aim of establishing the JSC was to protect investors. Additionally, it was created to regulate the capital markets by implementing various accountability and transparency mechanisms. This commission also acts to protect the capital markets from potential financial crimes. In addition to the above, the JSC is performing its role in creating awareness about the operations of financial markets in Jordan so investors can take part in the country's economic activity (Jordan Securities Commission, 2020).

The JSC is run by a board and this board is comprised of five full-time commissioners. These commissioners have vast experience in the operations of the security market, and the Council of Ministers appoints them for five years. This board prepares laws to monitor and regulate the security market. Additionally, it also lays down regulatory instructions for the country's capital markets. The board also issues a licence to firms which offer different financial services and run mutual funds for their clients. This commission is also responsible for adopting the accounting standards which are introduced and implemented in Jordan (Jordan Securities Commission, 2020).

To keep an eye on recent developments at the international level, the JSC tends to foster its relationships and cooperation with both Arab and non-Arab regulatory bodies. Collaboration with other regulators has resulted in the exchange of ideas for enhancing the efficiency of Jordanian capital markets. However, the current government is continuously focusing on this opportunity to further capitalise it, following the mission of providing leadership to Jordan's corporate sector (Jordan Securities Commission, 2020).

The JSC has maintained its focus on attracting both local and overseas investors to the Jordanian capital markets to expanding the investor base. Various activities are targeted for this purpose, such as increasing public



awareness through published material, delivering lectures and arranging visits of university students. The JSC, along with other stakeholders, is attempting to devise a mechanism for implementation of international corporate governance principles. It is also maintaining its collaboration with media and legislative authorities to protect investors in the capital market. The JSC also offers various training programmes internally and externally to increase the capabilities of its employees to safeguard the Jordanian capital markets (Jordan Securities Commission, 2020).

2.2.3 Amman Stock Exchange (ASE)

The ASE was established in 1999; it is a private institute and is responsible for managing the affairs of the security market in Jordan. The ASE holds administrative powers and has financial autonomy to regulate the trading of securities. The organisational structure of the ASE shows that it has seven members who act as the board of directors. It has 68 brokerage houses which facilitate the exchange of information with the board and monitor and report to the board. The ASE utilises regulatory and monitoring measures to ensure fairness and transparency in the securities market and protect the rights of investors. In addition to these measures, the Amman Stock Exchange has also issued various directives for the smooth running of the security market's operating affairs. The above mechanisms implemented by the ASE have provided a safe environment for investors in which they feel that their investment is safe. It also comes under the jurisdiction of the ASE to devise processes and methods to ensure transparent trading of securities on the stock market. The ASE also disseminates relevant information to all the key players of the security market to enhance the transparency and accountability within the security market (Amman Stock Exchange, 2020).

The Amman Stock Exchange and Jordan Securities Commission work closely to ensure proper implementation of local and international regulations for the smooth running of the market. These institutions also engage with other domestic and foreign regulators to improve the performance of the ASE



and build close relations with other entities to adopt the latest developments in the field. The ASE is a member of and collaborates with the World Federation of Exchanges (WFE). Additionally, the ASE is also working as a member of the Federation of Euro-Asian Stock Exchanges (FEAS). These collaborations are helpful to encourage international investors to the ASE. The Finance Ministry is very keen and excited to launch investment-focused seminars and workshops to motivate investors by providing the required information and knowledge. This helps in boosting the morale of investors and overall facilitate investing in Jordan (Amman Stock Exchange, 2020).

2.2.4 Securities Depository Centre (SDC)

Jordan's SDC is a public utility institution, and it was established in 1997 under Securities Law No. 23 of 1997. The SDC, which has a legal personality with financial and administrative autonomy, commenced operation in May 1999 and is the only entity in Jordan that is legally empowered by Securities Law No. 18 of 2017 to oversee the registration and deposit of securities, transfer of ownership and safekeeping of securities, and clearance and settlement of securities transactions in Jordan. For the SDC to perform its operations, it was necessary to establish a central registry and depository of authenticated shareholders along with a primary settlement process. The SDC ensures that the shareholders' registers of all public shareholding companies are held and maintained at the SDC in electronic form (Securities Depository Centre, 2020).

The SDC also works under the supervision of the JSC, and its role is to boost the confidence of investors in securities. It has financial and administrative autonomy in Jordan. Another primary function of the SDC is to develop an electronic database of registered shareholders. The SDC is one of the significant institutions in the Jordan capital market as it holds the ownership registers of all issued shares. It has been assigned, in cooperation with the JSC and the ASE, the task of developing the Jordan capital market (Securities Depository Centre, 2020).



2.2.5 Corporate governance development in Jordan

Over time, Jordanian investors were demanding a secured regulatory environment in which they felt that their investment was safe. This fact raised the need to implement a corporate governance code in Jordan so that the system used to govern Jordanian companies could be improved. For this purpose, the regulators developed the Jordanian corporate governance code (JCGC) that brings harmony to the Jordanian capital markets. The implementation of the JCGC ensures that the local market is following the essential criteria for enhanced transparency and accountability. This code has provided a sense of security to Jordanian investors and spread a positive message at the global level regarding the country's capital markets. As discussed, Jordan's financial markets have attracted investors and businesspeople in the past decade to invest in the country. The primary cause of this is the implantation of the JCGC by the Jordanian regulators (Central Bank of Jordan, 2020).

Mallin (2016) has stated that the Cadbury Report (1992) and OECD Principles of Corporate Governance (2004) laid the foundations of corporate governance around the world. The recommendations of the Cadbury Report (1992) were used by the regulators to develop the corporate governance codes for their countries. The focus of the CG codes was mainly linked with firms' board structure and ownership structure. Keeping in mind the guidelines of various international corporate governance codes, Jordan formulated its corporate governance code for banks in 2006. The Jordanian code was based on the best practices, OECD principles of corporate governance and the guidelines of the Basel Committee to improve the corporate governance mechanism for the Jordanian banking sector.

The JSC and ASE developed the rules and regulations required for the implementation of the JCGC. The JCGC provides a clear framework for improving the relationship between the managers and the shareholders for the elimination of agency conflict. The code also defines the duties, rights



and responsibilities of all the stakeholders. However, the Central Bank of Jordan has also played a crucial role in devising corporate governance mechanisms for the Jordanian banking sector. It has also issued a handbook of corporate governance to facilitate the directors (Jordan Securities Commission, 2020).

Al-Jazi (2007) stated that various laws which are relevant to CG have been issued and implemented in Jordan. The JCGC (2006) issued by the Central Bank of Jordan has mainly covered issues such as the structure of the board, AGMs, financial disclosures, accounting and auditing practices of companies, and ownership structure. Jordanian regulators issued a corporate governance code for public companies in 2007 that was amended in 2011. The main guidelines of the JCGC (2011) are that the size of the board should be between five and 13 directors, at least 1/3 of the directors should be NEDs, and CEO duality should be observed (Jordan Corporate Governance Code, 2011).

In order to improve the investment outlook, a disclosure department is working under the JSC and is accountable for implementing rules (JSC, 2017). In addition to this, Jordan's government has signed separate agreements with developed and developing economies to encourage foreign investors. These agreements have been signed with the UK, France, the US, Germany, Italy, Malaysia, Romania, Tunisia, Turkey, Algeria, Yemen, Bulgaria, Austria, China, Spain, Syria, Poland, Kuwait and Singapore (Jordan Investment Board, 2013). Further, Jordan has gone under various financial and economic reforms to ensure transparency and accountability in order to enhance investor confidence. Jordan is included in the top three nations in the MENA region that have successfully attracted foreign investment (Al-Muhtaseb, 2010). The next section of the thesis will provide the definition of corporate governance, corporate governance models, and the theoretical background of the study.



2.3 Definition of corporate governance

CG has gained popularity during recent years from all stakeholders, such as academia, professional bodies and government agencies. It has gained interest from around the world due to corporate crimes. Different researchers and professionals have defined CG differently, but the term is separated into two main perspectives, narrow and broader. In the narrow perspective, CG is defined as a tool that is helpful to protect the stakes of shareholders. In the broader perspective, it is a tool to protect the stakes of all the stakeholders of the business, such as investors, owners, creditors, suppliers, customers and government agencies (Gillan, 2006; Tricker, 2019).

Gillan and Starks (1998) and Zingales (1997) defined CG in narrow terms, and the emphasis of their definition was on protecting the stakes of shareholders. Gillan and Starks (2000), Solomon (2016), Millan (2016) and Tricker (2019) defined corporate governance in a broader sense and stated that CG mechanisms are helpful for companies to implement a sound monitoring system for the managers, so they operate in the best interests of all the stakeholders.

Aldamen et al. (2012) stated that managers should put their efforts into achieving the objectives of shareholders, so they feel that their investment is agency relationship exists protected. An between managers and shareholders. If managers are achieving the shareholder wealth maximisation objective, then shareholders get an impression that there is no conflict of interest between them and the managers. On the other hand, if the investors are not receiving a high return on their investment, then shareholders will get an impression that the managers are not working in their interest, and this phenomenon is known as agency conflict. This situation leads to the implementation of a system to monitor and control managers. The cost of implementing this system is borne by the shareholders, and it is knowns as agency cost. Keasey et al. (2005, p.251) defined CG as:



"The set of mechanisms – both institutional and market-based – that induce the self-interested controllers of a company (those that make decisions regarding how the company will be operated) to make decisions that maximise the value of the company to its owners (the suppliers of capital)."

The above definitions elaborate the broad view of corporate governance. The implementation of CG mechanisms is helpful to streamline business operations. Streamlined business processes help managers to achieve the shareholders' wealth maximisation objective. Furthermore, all the stakeholders appreciate the managers if they enhance the performance of the company (Solomon, 2014).

Firm performance holds a central position in governance mechanisms, and agency theory provides the best illustration to provide a framework to conceptualise the relationships between firm performance and structure of the organisation. Denis and McConnell (2003) stated that, in the absence of agency conflict, the performance of the company can be enhanced. But a lot of real-world examples have witnessed that managers have tried to gain personal benefits from the investment of shareholders, such as at Enron, WorldCom, etc. For this reason, hypotheses will be developed based on the agency theory and from the perspective of agency conflict that is adversely affecting the performance of companies (Tricker, 2019).

Agency theory deals with the issues of agency problems and the owner's interests with the purpose of maximising the shareholders' wealth. Generally, it is believed that, if the agency problem is decreased, then the performance of the firm is increased. This conceptualisation provides a way to increase the shareholders' wealth. Hence, both the definitions of CG have limitations, and agency theory provides hypothetical clarifications to elaborate variables and associations between the variables to test the relationship.



2.4 Theoretical framework

The theoretical underpinning of this study is primarily based on agency theory to investigate the relationships among study constructs, i.e. corporate governance practices and firm performance. Agency theory deals with the shareholders and managers in the corporate world.

Conceptualisation provided by agency theory helps to focus on only two parties, i.e. the owners (principal) and managers (agents). This one-dimensional perspective helps to carry out analysis very easily (Watson and Head, 2016). According to agency theory, it is assumed that conflict of interest increased due to managers being opportunistic from the shareholders' perspective. These explanations provide a powerful and robust basis for explaining and testing the relationships between the firm performance and CG. In addition to this, agency theory provides and suggests remedial measures by reducing the conflicts/problems between shareholders and agents (Fama and Jensen, 1983). Scant literature in this field of study suggests some sources of agency problems, such as available free cash flows, under-investment/over investment and not considering the rule of optimistic grid present worth (Dhumale, 1998; Jensen and Murphy, 1990; Tricker, 2019).

Performance of the firm depends upon the policy formulation and implementation by the management team. Furthermore, increasing the motivation of managerial personnel to handle the affairs of a firm is also crucial for top management, and, in the case when the objectives of managerial staff are misaligned due to improper monitoring, this can result in severe consequences for firms and owners (Liu and Fong, 2010). Thus, agency theory provides a monitoring mechanism to bring alignment between the objectives of the firm and its managerial staff, which protects the interests of shareholders/owners and ultimately increases the firm's performance (Mallin, 2019).



The ownership structure of any organisation/firm also has an influence on the affairs of the firm, and the board of directors is the fundamental characteristic which can mitigate the rising conflict among owners and managers in order enhance the firm's performance. Other factors which can increase the firm's performance are board size, the duality of CEO and chairman on one hand, and the ratio of non-executive directors (NEDs). The underlying phenomena behind the functioning of the executive and non-executive directors in managing the firm's affairs are explained by stewardship theory as well as resource dependency theory. Besides this, these two theories also provide explanations regarding how directors influence the firm's performance. However, there is overlap between some aspects of agency theory and these other two theories (Solomon, 2014).

Agency theory tends to provide alignment between the interests of agents (managers) and principals (owners) by proposing that this alignment can be created through introducing incentives for the agents so that they could perform better to safeguard the owners' interests and enhance shareholders value. In this regard, a statement by Jensen and Meckling (1976) provides guidelines and elaborates that introducing incentives can motivate the agents to improve their efforts so positively reflected in firm performance. This alignment can result in reduced conflict between the two parties, ultimately reducing agency problems and increasing firm performance.

2.4.1 Agency theory

Usually, in public listed firms, the organisational structure is devised in such a way that there is a distinction between ownership and control of agents and their principals. In this relationship, principals (owners) hire agents (managers) to manage the affairs of the firm in the owners' best interests and, in return for the managers' efforts, the owners pay them financial compensation in the shape of a salary and incentives (Sappington, 1991). This relationship can generate a conflict of interest due to a divergence of interests between these two parties. This conflict of interest which can impact



the firm performance has been investigated previously by most researchers through agency theory (Fama and Jensen, 1983; Mallin, 2019).

The theory revolves around the fact that conflicts of interest between the principals and agents arise due to the opportunistic nature of managers (agents). However, this premise considers managers as a conscious entity but also at the same time as opportunistic. Thus, under agency theory it is assumed that managers or agents tend to maximise their personal wealth/value instead of the shareholders' wealth/value (Demsetz, 1983). In addition, agency theory holds the assumption that due to less available information to the principals/owners, could not assess properly the efforts imparted by managers to increase the firm's value. (Mallin, 2019).

According to the conceptualisation of Jensen and Meckling (1976), agency costs are comprised of monitoring and bond costs as well as remaining losses. To monitor the activities of managers, a cost has occurred, which is termed the monitoring cost. On the other hand, the bonding cost occurs in developing the systems and structures, and this cost can be in the shape of financial or non-financial cost. The residual losses occur when managers tend to promote their own interests; this cost can also occur due to improper monitoring or bonding activities. According to Fama and Jensen (1983), residual loss occurs when the monitoring and bonding costs increase in comparison to the profit generated.

Agency theory considers the connection between investors and supervisors to be the essential relationship established with a specialist, in which the owners seek directors to manage the company to the most significant advantage of the former (Sappington, 1991).

Jensen and Meckling (1976) featured the primary connection with the specialist. They studied the arrangement of the ownership of the organisation, in particular, the work of responsibility for value as a component to regulate the administrator's enthusiasm to that of the owners. Moreover, Fama and Jensen (1983) described the work of senior



management by verifying the potential advantage of official supervisors in expanding companies. In this way, the office hypothesis mainly concerns the institutional action paths that influence the organisation's comparisons. The highlights of the operator's vital worldview are that s/he recommends clarifications and answers for different types of office problems and provides an approach to avoid discussions by creating motivating force agreements and a commitment approach to create management components (Mallin, 2019).

The office hypothesis suggests that the Non- Executive director (NEDs) take an essential job in controlling and administering the officers (Fama and Jensen, 1983). In this way, NEDs could increase the value of companies due to their external information and capabilities and their ability to verify (Fama, 1980). In this sense, both the organisation hypothesis and the trust in the activities anticipate a causal and positive connection between the company's performance and the proximity of the NEDs. In contrast, the management hypothesis claims that internal executives might be more inclined to take over managers than NEDs because of their better learning in relation to business tasks (Baysinger and Hoskinsson, 1990).

Furthermore, the administration hypothesis argues that the low maintenance/ formal position of NEDs largely represses their ability to verify and makes their commitment to essential leadership unimportant (Bozec, 2005). In this sense, instead of speculation about organisation and dependence on assets, the management hypothesis holds that NEDs will probably have a negative influence on the execution of the company.

NEDs can also add to the expansion of the board measure, which has the benefit of a broader set of skills which further add to the lack of essential leadership and correspondence that is reflected in the moderate performance of the sheets (Lipton and Lorsch, 1992; Jensen, 1993). As the size of the board increases, the problems of coordination and correspondence also increase (Eisenberg et al., 1998). Furthermore, the hypothesis proposes the division between the director and the managing



director from a similar position; the combination of these works in a single individual can cause the expansion of the organisation's problems, weakening the feasibility of the observation of the CEO (Jensen, 1993). However, the stewardship hypothesis recommends that a powerful administration depends on the level of managerial solidarity, so it is appropriate for the administrator and the CEO to have a position similar to that indicated from this point of view (Dalton and Kesner, 1987; Donaldson and Davies, 1991).

Agency theory proposes that specialists are less inclined to work. To reduce this difference in interests, investors must use internal quality control systems to select managers and, in this sense, balanced supervisors must act to satisfy their ability to increase investor esteem and improve the performance of society. This primary factor should be complemented by intentional effort to select and control supervisors, with CG tools that recognise any feasible problem. Assuming that organisational cost guarantees that the manager does not seek personal responsibility, ignoring the interests of investors, office expenses reduce the office problem and contribute to improving the company's performance (Pike et al., 2018).

The manager is an agent within the firm who acts on behalf of the owners (principals). For instance, a salesperson acts on behalf of the other person, namely, the principal. This principal can be a manufacturer and the agent can sell goods to others on behalf of the manufacturer. In other cases, a stockbroker on the stock exchange can be an agent of the client (principal) and can perform the activities of selling and buying on behalf of that principal. Thus, agents act by using the name of the principal and proceed with agreements and carry out transactions on behalf of that principal (Mallin, 2019).

According to company law, directors also act as agents of the company. Individual directors, as well as Board of directors (BODs) as a whole, perform various activities on behalf of owners and usually have the power to make contractual agreements with the other dealing parties. Hence, due to their



dominant position in the company, directors have the authority to decide what a firm can achieve. However, this power is not free from evil. Thus, this authoritative position of directors within the firm raises some questions regarding the usage of power in the case when owners (shareholders) and directors are different individuals. This raises two questions:

- How it can the principals (shareholders/owners) be assured that directors/managers, being agents, are acting in accordance with the interests of shareholders, and they are pursuing their interests?
- In a case when directors are not performing in accordance with the shareholders' viewpoint, then how can shareholders/owners motivate them to pursue their goals?

Being agents of the company, directors have various duties, and one of these is to perform the fiduciary duty. Alternatively, it can be called a duty of trust. Hence, to maintain trust, directors should work on behalf of principals in such a way that it should promote total good faith. Further, directors should not prioritise their interests at the cost of shareholders/owners' interests. Hence, if a director is found guilty of a breach of their fiduciary duty (duty of faith and trust), then they must be held liable in law, in the case when the firm is willing to initiate legal proceedings against them. The rest of the BODs could undertake this legal action against such a director or directors.

Agency theory deals with the relationship between the owners of a company and its managers; here, managers are called agents and owners are called principals in this relationship. The conceptualisation of agency theory holds that, when a firm is first established, its owners perform the managerial duties and they are also the managers of that firm. Over time, as the firm or company grows, its owners tend to hire managers to perform the managerial duties due to its increased size. Hence, owners expect that these managers will perform in the best way and will protect the owners' interests. Thus, due to this, there exists an agency relationship between the owners and managers of that company (Tricker, 2019).



2.4.2 Stewardship theory

Stewardship theory focuses on the mental and sociological strategies for supervision, as opposed to the monetary (financial) tools of the organisation hypothesis. According to Davis et al. (1997), this holds that hierarchical individuals have some kind of positive aggregate character that induces positive behaviour. Muth and Donaldson (1998) agree that the monetary advantage is not the sole engine of administrative behaviour, and supervisors require some vigilance to monitor the affairs of investors adequately. Subsequently, the separate property is not considered a deficiency of the administration hypothesis, since the useful practices are deemed to be the inactive/natural behaviour of the leaders (Davis et al., 1997), and are responsible for a variety of intentions despite the monetary benefit (Muth and Donaldson, 1998; Solomon, 2016).

Fama and Jensen (1983) found that the supervisors of the internal part of the board are almost certain that the external leaders form significant associations because of the deep understanding of the hierarchical exercises delighted by the previous ones. The administration hypothesis states that the concern for the professional movement prevents operators from acting against the interests of investors. The administrators' commitment to organisational performance is identified with the environment in provision of socio-social and mental elements (Clarke, 2004; Mallin, 2019). For instance, it is believed that garments have better performance with a more prominent reinforcement and the realisation of a job, which is a mental aspect. Publicly, managers usually recognise themselves as hierarchical agents and therefore believe that the power the directors give them is a means to train the association and several representatives to reach the authorised goals (Pike et al., 2018).

Donaldson and Davis (1994), in terms of a situational perspective, write that managers are expected to behave optimally in a participatory way. Furthermore, the duality between CEO and president will guide and control,



especially with regard to the more consistent decision-making process and strategy (for example, investments), which should add to higher success. As executives have increasingly comprehensive and in-depth information on daily activities within their companies, their options are better informed (Tricker, 2019).

2.4.3 Resource dependence theory

The hypothesis of the dependence of the activities indicates, the CG structures, for example, the management, influence the entry of the company into the primary activities for the execution of the company (Pfeffer, 1973). The hypothesis of the trust in the resources supports, in particular, the structures within a large NED organisation, due to the greater competence and information they can provide, as well as to the better management of the systems (Haniffa and Hudaib, 2006). In this way, NEDs can promote improving the management of systems with external partners, including customers, governments and various organisations (for example, banks, suppliers and buyers). In this way, NEDs improve access to assets (Nicholson and Kiel, 2007), which makes access to less expensive sources of information possible and, in this sense, strongly influences the performance of companies (Mallin, 2019).

Pfeffer (1972) and Pfeffer and Salancik (1978) argued that variety in the board's opinion and the basis of external leaders are vital components to meet the organisation's needs for any subsequent capital or to monitor the possibility of condition. Aljifri and Moustafa (2001) and Pearce and Zahra (1992) believe that stable bonds help them to secure their commercial advantages in case of ecological vulnerability. They further explained this phenomenon with the help of some useful data from a variety of sample organisations around the globe.

Furthermore, the theory of heritage dependence clarifies the strategies that organisations use to access money resources. Regarding dissolution problems, organisations are exceptionally informed to select delegates for



the money establishments on their pages (Mizruchi and Stearns, 1988). In any case, if the company has an abnormal amount of bank bonds, it will likely appoint a chief bank loan official during the meeting to encourage access to the bank. At the end of the day, it is a more straightforward method of obtaining credits (Thompson and McEwen, 1958; Pike et al., 2018).

Furthermore, Kaplan and Minton (1994) recognised that organisations often desire to assign budget leaders on the board if the costs of the actions or the performance of an organisation are divided. Furthermore, it is prescribed that internal managers be supplanted by experienced external managers when a company's performance worsens (Hermalin and Weishbach, 1988). The hypothesis of resource dependency uses external links of the council (Nicholson and Kiel, 2007). Hitt et al. (2000) argued that the development experiences sector negative effects of low capital availability, mind-boggling expenses, related to improperly created money markets and the unpredictability of financial improvement. In this sense, organisations are forced to discover innovative approaches to obtain benefits from the external connections of the board of directors. In the creation of nations, organisations must have ties with external resources (Tricker, 2019).

The theory of the dependence on activities maintains that the prepared condition of the company is reflected in the structure of its board of directors (Hillman et al., 2000; Boyd, 1995; Pfeffer, 1972), which implies that leaders are chosen for their aptitude to encourage the right of entry to the necessary capital. In this sense, it should be likely to recognise the rigid conditions of the board. For example, the closeness of financiers to management recommends that organisations seek modest access to capital, from which they derive that they plan large companies or find themselves in monetary problems (Hillman et al., 2000). In general, it may be required that a board with different individuals with connections changed to external resources has more significant access to these assets, which improves the performance and the esteem of the company (Pike et al., 2018).



As discussed in the above section that research questions are generated on the basis of agency theory. Agency theory is explaining the relationship between the corporate managers and corporate owners. But there are lot of real-world examples in which the companies were bankrupted due to agency conflict between managers and shareholders (Tricker, 2019). Various corporate governance rules and practices are adopted by the companies to eliminate the agency conflict between the managers and shareholders. This research will use various corporate governance mechanisms that are adopted by the companies to overcome the agency conflict, such as board size, CEO duality, non-executive directors and presence of board subcommittees.

Board size is an essential corporate governance mechanism because an effective board is vital to control the operational and governance system of the company (Lehn et al., 2009). CEO duality is providing an opportunity to the Chairman and non-executive directors to generate an effective monitoring and check and balance system on the CEO and executive directors in running the operations of the company (Arosa et al., 2012). The presence of non-executive directors on the board are ensuring that an effective monitoring is implemented within the company as well as executive directors can get vital professional advice from the non-executive directors because they are expertise in their area (Mallin, 2019). UK corporate governance code (2003) has recommended that all the public companies should establish audit, nomination and remuneration committees on the board so these areas can be executive and non-executive directors. Further, UK corporate governance code (2008) has stated that all the members of these committees should be non-executive directors for limiting the involvement of executive directors from audit, recruitment of board vacancies and setting the remuneration of executive directors. The above corporate governance mechanisms are adopted by the companies to overcome the agency conflict, which in turn helpful to enhance the firm performance. For this reason, the impact of the above corporate governance mechanisms will be studies in this research study.



The following section will provide the conceptual framework for the study that is adopted by the researcher. The reason to discuss the conceptual framework is to provide information that what corporate governance mechanisms and control variables will be used in the current study. Further, the section will provide information about various statistical tools that will be used to analyse the impact of corporate governance and control variables on the firm performance for the Jordanian listed companies.

2.5 Conceptual framework

It is the written presentation which provides explanations for the variables of the study, either in the shape of graphics or in narrative form. Further, this conceptual framework provides study factors/variables and hypothesised relationships among the study constructs (Wooldridge, 2002). Statistical tools will be used for the analysis. The correlation will be used to explore the relationship between the variables. Multiple regression will assist in examining the impact of different independent variables on the dependent variable. The following section provides details of the conceptual framework that is adopted for the study.



Table 2-5

Conceptual model

Dependent variable

Corporate performance measured by

- "Return on assets (book value measure)
- Return on equity (book value measure)
- Tobin Q (market value measure)"

Independent variables

Control variables

- Sales growth
- Capital expenditure
- Leverage
- Firm size
- Research and development expenditures
- Liquidity

Corporate governance variables

- "Non-executive directors
- Board size
- Duality
- Audit committee
- Remuneration committee
- Nomination committee
- Corporate governance index"

Source: Self-generated

The framework seeks to investigate the main premises of the study in hand and has been discussed thoroughly in the current scenario. Further, in-depth details regarding the variables, their measurement and the study hypotheses are provided in the last section of this chapter.



2.6 Corporate governance issues in developing countries

According to Allen (2005) and Oman et al. (2004), CG in developing markets had recently received much attention due to the increasing popularity of this field of management. The emerging markets will have a broad budget base for physical creation, including stock exchanges, commercial banks and central banks. However, they will have fewer procedures and frameworks for accounting, administration, guidelines, and other monetary bases and markets, and will be less productive with a smaller amount of liquidity than the world's most progressive cadres. These distinctions direct to a more obvious vulnerability and danger and improve the conceivable results of global improvements for financial specialists from all nations (Kearney, 2012; Mallin, 2019).

Many problems are affecting developing economies, such as uncertainty and risk, political insecurity, fragile production, high amount of government mediation and low-security dimensions for financial specialists (Tsamenyi et al., 2007). It is necessary to adopt valid CG structures. Furthermore, the lack of CG and a welcoming connection between banks, companies and government is one of a number of serious problems that have caused the 'associated free society' (Singh and Zammit, 2006). Nenova (2009) points out that the significant challenges linked to CG for creative nations are: the exchange of esteem (of investors or partners that do not control) to govern large investors; weak legal system; and problems related to audit (Gutterman, 2019).

It has been observed that different parts of developing markets are of fundamental importance in the impact of decisions made regarding the management of a company, for example, the ownership structure, the progress of the budget market and the nature of the administration (Fan et al., 2011; Ararat and Dallas, 2011; Tricker, 2019). Furthermore, the concept of a follow-up board in the interests of investors and to evaluate directors in a viable way is of urgent importance for an organisation in developing markets where CG components, in general, will be weak (Douma et al., 2006).



In developing nations, companies are usually established as family businesses where the family dominates the board of directors and often the decision to appoint executive and non-executive directors is based on nepotism and personal interests rather than on the basis of experience competence (La Porta et al., 1999). The closeness of relatives in an organisational meeting, in particular the creator, has been correlated to better levels of performance within specific nations; connections can be of great importance with close cooperation between high-level companies within countries, for example, Thailand. On the other hand, an increasingly effective result after the proximity of the marginalised appeared in several markets, for instance, in the Republic of Korea (Fan et al., 2011; Mallin, 2019).

It is believed that there should be an abnormal state of administrative freedom within a council, with unofficial heads that shape most individuals and positions such as director and executive director, which is why the oversight the board may improve, and office problems may decrease (Shleifer and Vishny, 1997; Fama and Jensen, 1983). Ararat and Dallas (2011) argued that they might end up being inadequate because there is not enough useful analysis for the investors they control. Investor control can be inclined to look for plans that have almost no advantage for investors, with poor critical leadership that has a negative effect on the organisation (Ararat and Dallas, 2011; Tricker, 2019).

2.7 Corporate governance: international principles and practices

Previous research (Franks and Mayer, 2001; Short et al., 1999; Solomon, 2014) recognised two main CG models: the sample of the marginalised (or Anglo-Saxon), which is used in the United States, United Kingdom, Australia, Canada and New Zealand; and the internal (or continental) information screen, which is used in the Netherlands, Germany, Switzerland, France, Austria, Sweden, Finland and Denmark. The highlights of the internal and external models are shown in Table 2.6.



The internal GC model is organised by a high dependence on the banking fund, a fragile legitimate insurance of minority investors, weak exposure, concentrated ownership, a part of the rule for property and council members in organisations, and the limited opportunity to combine or win (Rosser, 2003). Furthermore, Solomon (2013) argued that these investors could be a small meeting of investors (for example, loan banks), individuals of organisations (for example, establishing families) and the state. Furthermore, Solomon (2014) drew attention to the fact that insider information alluded to relationship-based frameworks due to the welcoming connection between the companies and their predominant investors.

From the first moment, without a doubt, an intimate connection between managers and investors would limit the problem of the office (agency); there is little effort to adapt the organisation's rewards, managers and investors to the possibility that they are mainly similar people. However, other GC problems seem to arise in such situations, for example, regarding the reduced size of the possession and manage partition (especially in family organisations) (Solomon, 2014). There may be a seizure of the prize by alternative investors in light of the issue of data irregularity, as alternative investors are incapable of increasing data about the organisation's activities due to a lack of clarity. In such circumstances, ambiguous exchanges related to money and abuse of services are regular (Solomon, 2016; Mallin, 2019).



Table 2.6 Features of insider and outsider CG systems

Model	Insider	Outsider	
Owners	Insider shareholders	Outsider shareholders	
Separation of ownership and control	Little	Separated	
Control over management	Insider shareholders	Managers	
Hostile takeover activity	Rare	Frequent	
Protection of investors	Weak	Strong	
Shareholders' rights	Potential for abuse of power	Potential for shareholder by majority shareholders democracy	
Shareholders voting	Majority of shareholders tend to have more voice in their investee companies	Shareholders characterized more by exit than by voice	

Source Mallin (2019)

Instead of focusing on employees, the untouchable model is represented by a strong dependence on the value of money, reliable and legitimate secure investors (especially minority shareholders), and dispersed ownership, and is reduced to working representatives, credit institutions and a different partner, solid settlement models, and considerable opportunities to combine and obtain a substantial need for dissemination (Rosser, 2003; Tricker, 2019). But outside (Anglo organisations) demonstration is owned by outside investors. The official problems that arise due to this posting have already been clarified. Various specialists (Fukuyama, 1992; Villalonga, 2006; Rosser, 2003; Mallin, 2019) argue that globalisation plays a central role in the distinctive combination of CG models throughout the world and their absorption by the Untouchable Anglo model. In this regard, one can count on all local and final products for exchange with the marginalised Anglo-Saxon or marginalised authoritarian settings.

Singh (2003) announced that most developing markets are defective. He noted that these areas of activity are experiencing the negative effects of



asymmetric information, simple accounting, administration and degradation in a much more obvious way than in sectors of the nation's able created activity. Bruner et al. (2002) argued that one of the disadvantages among the most generous purposes behind financial emergencies in developing markets is the lack of GC evidence. Singh and Zammit (2006) stated in previous inquiries that defect-making companies are developing bad rivalry; poor CG; and an intimate connection between government, business and banks. As a result, Singh (2003) was confident that improving his GC models was crucial for developing markets. Klapper and Love (2004) prescribe that these corporate sectors urge organisations to accept extensive GC studies.

Jordan has experienced extensive monetary and political changes from the 1990s to the 2000s, showing that Jordanian organisations are administered everywhere. Furthermore, an effort was ready to relate the GC standards in Jordanian organisations. This fact prompted the Jordanian Securities Commission (JSC) to release the JCGC in 2006 (further subtleties in this regard are discussed in the attached sections). The JCGC has updated various standards and CG meters that currently exist worldwide in universal codes. The proposals of the JCGC were separated from those of the OECD and the Cadbury Report (1992). The JCGC was prejudiced by the Cadbury Report (1992) and OECD (2004) standards for consultative groups composed of senior management, investor rights, transparency and transparency, obligations, and the intensity of the trustee and partition review commissions between the administrator and the managing director (Pike et al., 2018).

2.8 Corporate governance and firm performance

The classic principal-agent model proposes that managers being an agent of shareholders tend to pursue their own interests at the expense of shareholders, which tends to increase the agency problems. This agency problem can be due to several issues which managers face within organisational circuits, such as job security and compensation. Managers as agents tend to shape their behaviours according to the firm size despite the



firm's performance. Principals or shareholders bear agency cost in the shape of monitoring the activities of agents or managers in order to reduce information asymmetry as well as assessing the managerial efforts in enhancing the firm's performance. Thus, the critical component of such agency cost pertains to the monitoring cost, which occurs while gathering information regarding the actions and efforts of managers. On the other hand, the manager being an agent of the firm also bears a bonding cost, and this cost might reduce the agency conflict, although assessment of this bonding cost is not an easy task for principals (Jensen and Meckling, 1976; Mallin, 2019).

From the existing body of literature, agency theory provides useful insights to deal with and mitigate agency problems in corporate governance mechanisms. As evident from existing literature, agency theory also justifies performance-based compensation and managerial ownership in shares (Fama and Jensen, 1983). Agency theory also provides the necessary mechanism to reduce agency problems by increasing the role of externals for monitoring control (Jensen and Meckling, 1976). Corporate governance mechanisms provide various ways to deal with agency problems through an agency model with the core purpose being to bring harmony and alignment among the interests of principals and agents accordingly (Fama and Jensen, 1983; Tricker, 2019). Several studies have documented the internal governance mechanisms and their role in enhancing the firm's performance. Particularly, board and ownership structures have remained critical areas of interest in order to decrease misalignment among the principals and agents so that overall firm performance could be increased. Thus, alignment between the interests of principal and agent brings common interest which further ensures enhanced firm performance and greater value maximisation.

Nevertheless, this chapter elaborates on the mechanisms which can be fruitful in reducing agency problems. The chapter also seeks to posit insights regarding increasing managerial incentives for the common interests of principals and agents. More specifically, this chapter deals with board structure (e.g. the board size, CEO duality and the presence of NEDs)



control variables which have been used in this study. Additionally, this study investigates the role of foreign investors in predicting the performance of a company and hence contributes to the existing body of knowledge in the field of corporate governance in developing economies such as Jordan.

2.8.1 Board of directors (BODs)

The board of directors plays a crucial role in managing the managerial affairs of firms as well as in managing the potential conflicts among shareholders and managers. In the principal-agent relationship, the board of directors play its monitoring role in the case when the common interest of both agents and principals is misaligned. There is always the probability of agency problems as managers, being agents, tend to pursue their own interests instead of protecting the interests of shareholders. To tackle such situations, principals usually appoint the board of directors to enhance the firm's performance and reduce conflicts. Thus, this divergence of interests needs better monitoring of the firm's affairs, which may result in enhanced agency costs in the shape of monitoring cost as well as residual losses (Jensen and Meckling, 1976; Al-Ahdal et al., 2020). This situation results in such costs, which are borne by the shareholders. Thus, the reduction of agency costs remains a prime focus to enhance the shareholders' wealth.

On the other hand, the board of directors is the apex body in the hierarchy of corporate control systems. Liu and Fong (2010) and Alqatan, Chbib and Hussainey (2019) stated that more power and control exercised by the board of directors ensure maximisation of shareholder value as managers as agents found themselves in a position to align with the principal's interest. Thus, it can be argued that the board of directors is an essential instrument in the mechanism of corporate governance to protect the interests of shareholders (Jensen and Meckling 1976). Hence, an independent board of directors is considered favourable and seeks to bring an effective and efficient governance mechanism by independent directors who can oversee the activities of managers or agents in a more transparent and unbiased manner (Liu and Fong, 2010; Shu and Chiang, 2020).

The power inherited by the board of directors enables significant decisions in favour of the organisation (Gillan, 2006; Booth et al., 2002; Detthamrong, Chancharat and Vithessonthi, 2017). The board of directors also ensures that managers are protecting the interests of shareholders by pursuing aligned interests. Fama (1980) stated that the BODs is considered a crucial instrument which helps to inspect the decisions of managers. According to the assumption of agency theory, the board of directors is an essential tool which helps to ensure the implementation of corporate governance in the true spirit, which further tends to reduce the agency conflict among shareholders and managers (Fama, 1980). From the viewpoint of resource dependency theory, BODs act as a co-optative mechanism, and this phenomenon helps to calibrate the organisation with the requirements of external environments (Aguilera and Cuervo-Cazurra, 2009; Razzaque and Mether, 2020).

According to the perception of Solomon (2014), there should be some fundamental observation regarding the selection of the board of directors; such board members should hold frequent meetings, and there should be effective communication among the board members and owners/investors. Board members should have the ability to develop consensus and harmony in accepting the suggestions of other members; and they should have a high level of integrity and concern regarding taking financial risks. Walker (2005) stated that there are two main key points which should be kept under consideration in selecting the board's members, one is to choose appropriate members, and other is the financial compensation. However, Ingely and Walt (2002) and Tricker (2019) concluded that there should be diversity in the board members, such as board members should be from both genders, i.e. male and female, as well as diversity should also exist in terms of their experience. The worth and effectiveness of a board member are assessed based on the value addition in the firm. Such suggestions are also considered in the UK combined code, which elaborates that:



"The board's role is to provide entrepreneurial leadership of the company within a framework of prudent and effective controls which enables risk to be assessed and managed. The board should set the company's strategic aims, ensure that the necessary financial and human resources are in place for the company to meet its objectives and review management performance. The board should set the company's values and standards." (UK Combined Code, 2006, p. 3; Mallin, 2019)

Responsibilities of the board of directors are classified from three perspectives: control perspective, services perspective, and resource dependence perspective. The board of directors is responsible for appointing or terminating the managers (taking care of the organisation's day-to-day operations) and the CEO if they are not protecting the shareholders' interests (Monks and Minow, 1995; Braendle, 2019). The second perspective relates to the services role, which elaborates that they should provide a counselling facility to the CEO regarding various administrative and managerial issues. Further, the BODs also have the power to impart their advice regarding policy formulation and its implementation (Johnson and Daily, 1996). The third perspective concerns the resource dependency theory. According to this theory, it is the fundamental purpose of the board to assist the firm in the utilisation of internal and external resources to improve the firm's performance (Pfeffer and Salancik, 1978; Anderson et al., 2020).

However, such roles performed by the board of directors in a firm result in conflict between the board and the CEO. To mitigate such conflicts, Fama and Jensen (1983) recommended that most of the board members should be non-executive directors so that these members should mediate in case of disagreements. Hermalin and Weisbach (1988) and Bhaumik et al. (2019) also stated that the presence of non-executive directors on the board improves the check and balance system on the executives and enhances board independence. They further stated that in the boardroom there always exists a significant conflict among board member and CEOs (Mallin, 2019).



Regarding board structure and its functioning in day-to-day operations of the firm and the presence of non-executive directors, it is still the primary research questions. Keeping in mind this issue in the context of the Jordanian CG Code (JCGC, 2006), suggestions are available which support the perception that there should be a reasonable board size ranging from five to 13 members, each with a background of necessary skills and experience level. In addition to this, the roles of CEO and chairman should be separate, i.e. duality must be avoided, and 1/3 of the members should be non-executive directors. Thus, due to such stipulations present in the JCGC (2006), this study considered board size, CEO duality and percentage of NEDs (JCGC, 2011).

The effectiveness of the board depends upon its function related to its capability to monitor management so that misaligned activities can be eradicated. Resultantly, the quality of decisions taken by the board has the potential to affect the performance of managers and firms. Thus, better monitoring by board members could force management to act in the interests of shareholders, ensuring value maximisation and reducing the agency conflict. From here onwards, the literature review seeks to posit the point of view about various board-related CG mechanisms and their role in predicting the performance of firms (Akbar et al., 2016).

2.8.2 Board of directors' sub-committees

Board sub-committees play a critical role, and these committees enhance the efficiency of the corporate board (Jiraporn et al., 2009; Tricker, 2019). According to Harrison et al. (1994), Cifci et al. (2019) stated that board sub-committees are classified into two major types as per their functioning: one example is related to monitoring or oversight functioning, and the other is based on providing support to the management. The operating board committees provide their expertise and support to the firm's management regarding strategic decisions. Such committees are intended to protect the interests of shareholders by keeping an eye on the activities of corporate executives. Agency theory also proposes monitoring activities to monitor the



activities of executives through auditing services (Fama and Jensen, 1983). Further, the proper appointment of directors and senior management is also related to the monitoring function of audit committees (Chhaochharia and Grinstein, 2009; Mallin, 2019).

The Cadbury Report (1992) proposed that, generally, board committees are an additional tool to manage the financial affairs of the firm as well as to ensure accountability to protect the interests of shareholders (Cadbury, 1992). According to Harrison et al. (1994), successful management of board sub-committees can induce responsible behaviours among the board members. Thus, board committees ensure legitimacy and enhance the credibility of corporate governance mechanisms. Hence, if such committees perform their functioning well, then the conflict among shareholders and managers can be reduced, which can result in decreased costs and increased shareholder wealth (Weir et al., 2002; Shahid and Abbas, 2019).

According to the observation of Harrison et al. (1994), the role of board committees has been increased since the 1980s, and most of the available corporate governance codes recommend such committees. These codes advocate establishing three types of committees, such as nomination, remuneration and auditing committees. However, in spite of the plentiful literature available regarding recommendations regarding board committees and their prevalence in the corporate system, the role of monitoring committees on firm performance is not clear. However, some studies have claimed that such committees have the potential to influence the firm's performance in a positive manner (e.g. Harrison et al., 1994; Sun and Cahan, 2009; Qurashi, 2018).

Operating committees of the board of directors are usually constituted and dominated by an insider, while non-executive directors often form monitoring committees, and thus these NEDs become a source to protect the interests of minority shareholders (Klein, 1998; Vefeas, 1999). Researchers who favour the small size of such committees argue that these committees function very well with a small size because they can arrange meetings



frequently, which helps to analyse the available information and so is very convenient for decision making (Karamanou and Vefeas, 2005; Shehata, 2015). In addition to this, the presence of NEDs on the board helps to integrate and strengthen the knowledge and expertise of externals in the firm's decision-making process, and the board can concentrate on the issues of strategic interest (Harrison, 1987). Weir et al. (2002) stated that board sub-committees thus help to promote the corporate governance mechanism by increasing its credibility and legitimacy. It is essential to mention that the importance of these committees in the modern-day organisation is clear when we talk about the efficient and effective implementation of the corporate governance mechanism (Jacoby et al., 2019).

Out of these committees, the audit committee usually performs its duty by holding regular meetings with internal and external auditors to make the audit processes transparent as well as ensuring robust internal accounting controls. Such a review of accounts and financial information helps to reduce the information asymmetry through timely and verified disclosure of financial information to stakeholders, which further decreases agency costs (Klein, 1998). A timely and robust audit through an audit committee ensures transparency, and the chance of fraud becomes minute, which boosts investor confidence as well as increasing the firm's value. Thus, it is essential to understand the internal control evaluation process to assess the negative behaviours and errors which have become dangerous for the firm's integrity (Caplan, 1999; DeZoort, 1998; MacAulay et al., 2020).

Weir and Laing (2000) and Razzaque and Mather (2020) argued that the remuneration committee helps to reduce the agency conflict between management and shareholders by selecting the remuneration/compensation of senior management and this brings alignment among the interests of shareholders and agents. Thus, in the case when the compensation is not fair, the senior management can conspire with the CEO to obtain benefits. NEDs can bring harmony among the remuneration committees and management by ensuring that remuneration will increase performance of the management as well as improving the firm's value (Gregory, 2002; Monks,



2001). However, in some cases, external/independent directors tend to consult with external experts regarding remuneration decisions, which can be undermined in some instances if not appropriately managed (Monks, 2001; Al-Ahdal et al., 2020).

Similarly, the nomination committee helps to reduce the conflict between shareholders and agents through the appointment of qualified non-executive/independent directors (Vefeas, 1999). The Cadbury Report (1992) stated that nomination committees play a crucial role in strengthening the board of directors by selecting a suitable composition of board members. Thus, the ability of NEDs to perform their monitoring role well depends upon the recruitment process (Vafeas, 1999) as well as their independence on the board. In the absence of nomination committees, outside directors can misjudge the performance of CEOs, which can result in higher compensation to the CEO and increasing costs for the firm (Westphal and Zajac, 1995). Hence, such agency conflicts can be either removed or reduced through improving the functioning of a firm's nominating committee (Jensen, 1993; Anderson et al., 2020).

On the other hand, despite such positive comments from previous researchers, some have criticised the functioning of board committees and have postulated that board committees can hamper the financial performance of firms. Some proponents of such arguments state that board committees are a financial burden on companies (Vefeas, 1999). Other authors state that such committees require a lot of managerial supervision, thus reducing the time for other day-to-day activities (Conger et al., 1998). Such committees can replicate the overall functioning of board members and lead to an overlap in the duties and responsibilities of board members. Lastly, such committees increase the heterogeneity among the board members, which can result in a lack of consensus and cohesiveness among board members regarding essential issues. The literature is still debating the pros and cons of all these committees (Detthamrong, Chancharat and Vithessonthi, 2017).



Some empirical studies have documented the impact of audit committees. For instance, a study by Wild (1994) investigated the market reactions in 260 listed USA firms for a period of 15 years from 1966 to 1980. He investigated the influence of audit committees before and after their constitution. Statistically, a significant improvement was observed in share returns after the establishment of an audit committee, strengthening the argument that audit committees tend to enhance the managerial accountability, which ensures higher performance (Shehata, 2015).

A similar study was conducted by Vefeas (1999) in 606 USA listed firms and concluded that nomination committees increase the quality of the board as well as enhancing its effectiveness. A similar study was conducted by Main and Johnston (1993) on a sample of 220 UK listed companies to assess the effectiveness of remuneration committees. The outcomes of the study revealed the contradictory situation that these committees reduce shareholders' value while increasing executives' pay. Klein (1998) conducted a study on 486 US listed firms covering two years for the span 1992-1993 to ascertain the link among the establishment of audit, compensation and nomination committees and firm performance. The empirical results of this study could not show a statistically significant association among these variables (Mallan, 2019).

Thus, every board committee performs some specific functions, and if all a firm's committees perform their duties very well regarding monitoring and operating functions then the firm's performance may be increased. The present study investigates the impact of board committees of non-financial listed companies in the context of Jordan. For this purpose, data has been collected from various electronic databases and annual financial reports of the firms, keeping in mind the guidelines laid out by the Jordanian Corporate governance code of 2006 regarding the selection of members of the audit, remuneration and nomination committees from the board of directors. This code was implemented from the start of 2007 (Gutterman, 2019).



2.8.3 Board size

Generally, it is assumed that a larger board size will bring diverse knowledge and expertise that will help to increase the firm's performance. However, empirically it has been found that larger board size can carry serious consequences such as lack of coordination and commination, which can hamper the effectiveness of the board (Eisenberg, 1989). In addition to this, it has been observed that, in larger boards, directors do not criticise the top management regarding improvement in performance, and they do not take performance-related issues seriously (Lipton and Lorsch, 1992; Tricker, 2019).

According to the perception of Jensen (1993), large-size boards (increasing from seven to eight members) result in heavy monitoring and operating costs due to less effective functioning. The agency model supports this argument. In large-size boards, the CEO becomes the sole party who has control, instead of the board having control and monitoring the firm's management. Such a situation provides space for the managers, and they tend to pursue their interests at the cost of shareholder value by increasing the agency cost and reducing the firm's performance (Hermalin and Weisbach, 1998; Solomon, 2014). Similar findings have been reported by Kholief (2008): that a large board size does not induce cohesiveness among the board members, and it becomes challenging for the board members to develop a consensus over important issues. Hence, large boards become slow in their monitoring and operating functioning and efficient decision making is hampered, increasing the agency cost for the firm through increased agency conflicts, resulting in worse firm performance (Gutterman, 2019).

Ahmed et al. (2006) posit the view that it becomes difficult in the presence of a large number of members to formulate and adopt new ideas because a large board lacks cohesiveness and consensus, which results in hampered board functioning where novel ideas are not encouraged. Thus, such situations indicate that conflicts among the board members themselves do not bring them to the same page, i.e. the interests of shareholders. Instead,



they pursue their interests, which increases the agency problem. In the opinion of Cascio (2004), there is still a need to find out a sufficient and suitable board size in firms so that firm's affairs can be managed through its board.

Simply, it can be stated that there is no specific tool to measure a reasonable and suitable board size in the firm. Similarly, there is no formula which can suggest the number of directors from inside or outside the firm. However, some studies support the argument that a smaller board size is good, while others have argued in favour of a large board size. For instance, in the opinion of Yermack (1996), large boards have less cohesiveness and lack proper communication. Such a situation brings agency issues and results in decreased firm performance. Thus, a large board size can induce agency problems and director free-riding problems, which can cause conflicts among the board members resulting in poor performance of the firm (Jacoby et al., 2019).

However, in a situation where the board is large in size, the CEO dominates the decision-making process because s/he enjoys a powerful position in the firm, which increases agency problems and resultantly lowers firm performance (Miller, 2003). It is important to mention that a relatively large board tends to be more effective in modern-day organisations in terms of decision making and improving a firm's overall efficiency (Pfeffer, 1972, Goodsten et al., 1994; Tricker, 2019).

This phenomenon tends to support the large board size because large boards have more diversity and effective cohesion, which can help board members to transcend challenging situations more efficiently (Goodstein et al., 1994). Large boards can function smoothly in drastic economic conditions or during financial turbulence and can mitigate the agency conflicts in such circumstances (Mintzberg, 1983). However, in smaller boards, lack of diversity can result in poor strategic development by reducing the firm's ability to cope with such situations (Goodsteing et al., 1994; Shu and Chiang, 2020).



Literature provides limited support regarding the argument that larger boards provide diversity and expertise to the firm through their broad background (Arosa et al., 2010; Dalton et al., 1999) and outside linkage (Haniffa and Hudaib, 2006; Yawson, 2006). In the perception of Dalton et al. (1999), larger boards help the members to attract more highly qualified opinions. Large boards also have the potential to improve the decisions and their related outcomes, sharing ideas and contributions, which strengthens new knowledge and expertise for the management, thus reducing the agency conflict and increasing the firm's performance (Lehn et al., 2009; Mallin, 2019).

Previous studies have not reported a clear picture regarding the relationship of board size and firm performance, and in an attempt to find such relation Anderson et al. (2004) conducted a study to assess the association among board size and firm performance, which was found to be negative. They reported that financial markets react in positive ways to the announcement of board downsizing in a firm. Similarly, a firm's announcement regarding an increase in the number of board directors' results in decreased equity value. They further stated that this pattern is not observed in all types of companies because it is not a linear function in many cases. They postulated that small and medium-size firms are affected negatively with the announcement regarding increasing the board size while large firms do not suffer from such announcements (Shahid and Abbas, 2019).

Yermack (1996) reported a negative association among the board size and firm performance where firm performance was measured based on Tobin's Q in a sample of 452 large US firms for a period of nine years covering the span 1984 to 1991. In his study, he did not consider the utility and financial companies because such firms follow government regulations and their board of directors is bound to adopt these regulations in making decisions. His study reported that firms with small boards show improved financial performance in comparison to firms with large boards. He further stated that incremental cost would observe an increasing trend with the increase in



board size and firms would obtain higher market value with smaller board size. Findings of his study supported through empirical results proved that firms had obtained more value in capital markets through diversification (Detthamrong, Chancharat and Vithessonthi, 2017).

Previous studies have shown that small boards produce more favourable results in regard to firm performance in comparison to large boards, as the efficiency of the board decreases with the increase in board size because large-size boards have barriers in coordination and processes (Al-Khouri, 2006; Guest, 2008; Haniffa and Hudaib, 2006). However, Eisenberg et al. (1998) stated that Yermack's (1996) study, which was conducted on large firms, could not be generalised on small-size firms because the cultures of large and small firms differ. A study conducted by Eisenberg et al. (1998) on a sample of 879 small and medium-sized firms over a period of three years covering the span of 1992 to 1994, showed a negative association among board size and firm financial performance where financial performance was measured based on return on assets (Solomon, 2014; Braendle, 2019).

Various studies have documented the impact of board size on firm performance (Bozec, 2005; Cheng et al., 2008; Guest, 2008) and proved that small board size is linked with reduced agency cost, which leads to improved financial performance. However, other researchers have also documented the impact of large board size and firm performance and reported a positive relationship between firm performance and large board size. Al-Muhtaseb (2010) stated that the market responds positively to an increase in board size. They argued that large board sizes show better monitoring of firms. However, a large board size shows poor operating performance, and the main reason behind this poor performance is the diversity of board size and communication skills inherited by diverse board members.

A study conducted by Sanda et al. (2005) on a sample of 93 Nigerian companies for a period of four years covering the span of 1996 to 1999, reported a positive association among board size and firm financial performance measured through return on equity. Their results supported the argument that large board size provides improved access to the external



environment, which enables them to attract financing and raw materials conveniently, as compared to firms with small boards. This argument is aligned with the resource dependency theory, which postulates that a large board offers greater access to critical resources such as raw materials and finance. These findings are in alignment with the studies of Kiel and Nicholson (2003), Beiner et al. (2006) and Coles et al. (2008). Thus, it can be assumed that this divergence might be due to the perception of shareholders and management regarding large board size and its relationship with the knowledge-enhancing capability (Mallan, 2019).

According to Haniffa and Hudaib (2006), large board size has the potential to improve business decisions by reducing the agency conflict among shareholders and agents, which reduces agency problems. Similar results were also reported by Mangena and Tauringana (2007). In their study, they investigated the relationship of board size and firm performance of 72 firms which were listed on the Zimbabwean stock exchange for a period of three years, covering a span of 2002 to 2004. More sophisticatedly, their results remained unchanged despite adjusting the inflation factor. This situation shows that large boards facilitate effective monitoring even in an uncertain situation, either economic or political; this reduces the agency conflicts and increases the firm's performance. Ho (2003), Topak (2011) and Cifci et al. (2019) reported that financial performance and board size had no relationship.

The above mixed results show that there is a lack of consensus among the researchers regarding board size, either large or small, and firm performance. Thus, the issue of large or small board size is related to controlling and management activities of managers. Therefore, it can be assumed that, if monitoring is carried out properly, it can amend the behaviours of managers in alignment with the reduced agency conflicts and increased firm performance. Under strict monitoring and control, it becomes complicated for the managers to forego the interests of shareholders, and they conduct improved decision making to maximise the shareholders' wealth.

The Jordanian corporate governance code was introduced in 2006 to develop the capital market; it states that:

"The administration of the Company is entrusted to a board of directors whose members shall be not less than five and not more than thirteen, as determined by the Company's memorandum of association." (JCGC, 2006; JCGC, 2011)

The legislator in Jordan has proposed that a board should consist of five to 13 members. However, some firms do not follow the instructions of the JCGC, which might be due to variation in firm size and nature.

2.8.4 CEO duality

The duality of the CEO has the potential to influence the firm's agency problems. Agency theory provides the necessary support behind the idea that enhances the board's independence from the activities of management. This separation ensures improved performance as a result of proper monitoring and overseeing (Jensen, 1993). Similarly, stewardship theory postulates that there should be no separation in the control and monitoring activities of the CEO, and there should be duality. In a case when a single person is obliged to make decisions and responsibilities, this might have the potential to facilitate the integration of knowledge regarding firm operation (Adams and Mehran, 2005; Adams et al., 2007; Arosa et al., 2012; Peng et al., 2010; Jacoby et al., 2019). The CEO also lists all the relevant issues on the meeting agenda which might be discussed at a board meeting. The CEO generally provides an administrative facility to the firm and is accountable for the necessary implementation of organisation-wise decisions and policies (Fama and Jensen, 1983; Akbar et al., 2016).

According to Mallette and Fowler (1992), if the roles of CEO and chairman are combined into a single person, then it can result in increased overall control and, on the other hand, it can minimise the power of the board. Alternatively, duality in a single person can control the ability of independent



directors regarding their ability to monitor and control the governance role. Under such a circumstance, the conflict between shareholders and agents will further hamper the firm's performance negatively. Thus, to ensure the independence of the board, the roles of chairman and CEO must be separated so that there should be a proper balance of managerial behaviour (Ehikioya, 2009; Van den Berghe and Levrau, 2004). This situation can also be helpful to prevent managers from acting in their own benefit and interests. Fama and Jensen (1983) stated that separation of the roles of CEO and chairman establishes a clear boundary among the control and monitoring functions of non-executive directors. The UK Combined Code hence also recommends separation of roles, stating that:

"There should be a clear division of responsibilities at the head of the company between the running of the board and the executive responsibility for the running of the company's business. No one individual should have unfettered powers of decision." (UK Combined Code, 2006, p. 4)

Conversely, stewardship theory postulates in favour of duality. It states that duality has the potential to influence the performance of the company in positive ways due to various factors such as knowledge regarding the firm and investment opportunities, which can ensure improved decision making (Weir et al., 2002). As per the findings of Adams and Ferreira (2007), duality can bring a positive performance in a firm because the chairman, being the CEO, can provide their expertise and knowledge to the board directors. This facilitation can also help the board members more effectively, and thus duality can influence the firm's performance in a positive manner. More specifically, the potential conflict between CEO and chairman can be eradicated by fixing the responsibilities of CEO and chairman in a single person (Baliga et al., 1996; Harris and Helfat, 1988; Bhaumik et al., 2019).

If the argument developed based on stewardship theory is accepted, then it can be assumed that the CEO is actively engaged and driven to lead the company (Boyd, 1995). Thus, it can be argued that CEO duality will benefit



the firm even in challenging circumstances (Chahine and Tohme, 2009). According to Tsamenyi et al. (2007), the phenomenon of duality is generally observed in small-size companies because such firms have concentrated ownership structures which indicate the integration of various roles into a single person. According to Vafeas and Theodorou (1998), CEO duality might have the potential to help in removing agency conflict and reducing agency costs if further supervision is offered. Further, duality can increase the accountability of a company because, in the case of duality, the poor performance of the firm can be linked with the CEO easily (Bozec, 2005; Abor, 2007; Sheikh and Wang, 2012; Bhaumik et al., 2019).

An investigation carried out by Rechner and Dalton (1991) on a sample size of 141 firms from the Fortune 500 list for a period of six years covering the span of 1978 to 1983 reported that firms with separate control of CEO and chairman perform better than those firms who have duality. According to a study by Dahya et al. (1996) which was carried out on a sample of UK listed firms, it was observed that the stock market responds in a positive way to those firms where the roles of chairman and CEO are split from each other. Similar findings were reported by Haniffa and Hudaib (2006) in their study of 347 Malaysian listed firms, and they postulated that separation of chairman and CEO role ensures improved firm performance.

Boyd (1995), Elsayed (2007) and Shahid and Abbas (2019) reported a positive association between the duality and financial performance of firms. According to the perception of Boyd (1995), if the two roles are combined into a single person it can result in improved decision making. Similarly, Davis (2001) and Donaldson and Davis (1991) reported that CEO duality ensures a unified leadership which can facilitate the integration of knowledge into the firm. These results are in alignment with the argument that CEO duality increases the chances of improved decision making, which results in improved financial performance. Bozec (2005) surveyed a sample of 25 companies for a period of 25 years covering the span of 1976 to 2000 and did not find any significant impact of CEO duality on the indicators of performance.

The above discussion indicates that scant literature shows mixed and contradictory findings regarding the dual role of CEO and chairman. Both perspectives (agency and stewardship theory) provide different insights regarding the role of CEO as chairman. In some instances, duality can increase firm performance while, in other cases, it can hamper the decision-making process, resulting in agency conflict and reduced firm performance. According to the recommendations of the JCGC of 2006, the roles of CEO and chairman should be separated from each other.

2.8.5 Non-executive directors

Some corporate governance codes recommend a combination of both of executive and non-executive directors in the board. According to agency theory, there is a need to ensure sufficient monitoring mechanisms to protect the interests of shareholders, and for this purpose the presence of non-executive directors on the board is necessary. Thus, it can be assumed that the presence of a higher number of NEDs will ensure reduced agency cost through proper monitoring (Shleifer and Vishny, 1994; Mallin, 2019).

Raheja (2005) reported that the presence of executive directors could ensure better performance due to their specific knowledge and intrinsic motivation. Also, NEDs provide separate and independent monitoring which results in improved performance, despite their potential to increase the firm's performance. They also have less knowledge relevant to the firm. The emergent consensus is that a board must be diverse, which should have a positive impact on the overall value of the firm so that accurate strategic decision making can take place (Gabrielsson, 2007). The presence of a large number of NEDs on a board decreases the CEO's power over the board, which is also supported by agency theory. Thus, the presence of NEDs on the board can safeguard the interests of shareholders (Gabrielsson and Winlund, 2000; Tricker, 2019).



The presence of NEDs increases the reputation and legitimacy of the company (Pfeffer, 1973; Pfeffer and Salancik, 1978). It can also help to overcome the issue of shortfall related to human resources (Daily and Dalton, 1993). Despite this, the presence of non-executive directors also has the potential to mitigate the conflict among board members and can put them on the track of improved financial performance. However, the critical nature of non-executive directors remains at its place in explaining the overall corporate governance mechanism in contemporary firms around the globe (Mallin, 2019).

However, stewardship theory postulates in favour of executive directors because NEDs have limited access to the specific or firm-related knowledge and information of the firm's internal process. In the perception of Bozec (2005) and Jiraporn et al. (2009), NEDs work as part-time workers for the firm, and they cannot monitor managers because they do not have enough information. Hence, in the presence of a large number of NEDs, decision making will be very weak, and decision quality will not be up to the mark, which will result in reduced performance for firms. Similar findings have been reported by Weir and Laing (2000), and they argued that, due to NEDs only having part-time contracts, they are not familiar with the day-to-day operations of the firms (Higgs and Britain, 2003; Solomon, 2014).

Lawrence and Stapledon (1999) stated that it is not easy for NEDs to enhance firm performance because of several issues; first a link might exist between the CEO and the NEDs apart from their genuine relations in the context of that firm. On the other hand, the resource dependence perspective argues that the internal strength of the firm enables it to gain a competitive advantage which cannot be achieved through NEDs (Shehata, 2015). Despite agency theory arguing that the presence of NEDs can improve the financial performance of a firm, previous studies based on empirical findings in the existing literature show mixed findings (Baranchuk and Dybvig, 2009; Gordini, 2012).



Gordini (2012) stated that there is a positive association among NEDs and the financial performance of a firm, and NEDs have the potential to increase the firm's performance through their skills and experiences. Khan and Awan (2012) found a positive and significant association between the NEDs and the financial performance of a company in which financial performance was measured based on ROA and ROE and market performance was measured based on Tobin's Q. It might be due to the close monitoring mechanism and contribution to the firm.

These results/findings it concluded that NEDs ensure an effective monitoring function to develop the managed and disciplined behaviour of the managers. Contradictory to these findings, some researchers have reported a negative relationship for the relationship of NEDs and firm performance (Agrawal and Knoeber, 1996; Bozec, 2005; Shu et al., 2020).

Besides these results, some researchers have reported no relationship among these factors (Arosa et al., 2012; Kumar and Singh, 2012). Hence, keeping in mind the perspective of agency theory presence of NEDs on boards is mandatory, which can ensure proper control and monitoring activities resulting in reduced agency conflict and enhanced firm performance. This argument is also supported through resource dependency theory, which states that the presence of NEDs on a board incorporates their skills and expertise for improved networking and relationship. If NEDs play their role effectively, and they carry out the activities of monitoring and control efficiently, this can discourage the agents/managers from pursuing their own interests. Thus, the presence of NEDs can be regarded as a reliable regulatory mechanism (Mallin, 2019).

Stewardship theory supports the argument that the presence of NEDs will increase the agency problems due to information asymmetry because NEDs work part-time for the firm and they do not have accurate information and knowledge about the firm's day-to-day activities. Thus, the overall performance of the firm will be hampered. Jordanian firms have a one-tier board structure: NEDs and executive directors both sit on the same board.



As per the recommendations of the JCGC (2006), board size should range from five to 13 members. Additionally, the JCGC (2006) recommends that at least 1/3 of the board members must be non-executives.



2.9 Summary

Keeping in view all above mentioned facts, there is a visible evidence of presence of unique corporate governance mechanism in Jordan in order to provide deeper understandings of economic environment and corporate governance framework operating particularly in Jordan. This chapter firstly has reviewed Jordanian economic environment and seeks to provide general background of various economic sectors which are working in Jordan and contributing the economic development of Jordan. Major industries/sectors are also discussed specifically in this context. Like other nations, Jordan is also interested in attracting foreign investment; hence series of economic and financial reforms being planned are thoroughly discussed. A holistic picture of overall investment culture which is prevailing in the country has also been discussed

Corporate governance is a mechanism which provides a code of conduct to manage the affairs of firms so that investors could get best output of their savings (Shleifer and Vishny, 1997; Cadbury Committee, 1992). Although various definitions have been given by scholars into the previous literature, this chapter of thesis elaborates the phenomena of corporate governance from two perspectives one is shareholder's perspective while other is stakeholder's perspective. In alignment to the study objectives regarding relationship of corporate governance and organization performance a slight definition was more pertinent because this narrow definition provides more realistic and in-depth explanations. Although different theoretical frameworks have been discussed in this study, but agency theory is more relevant and explains proposed relationships. Aim of discussing relevant theories is explain the mechanism from the aspect of each theory. Also, this chapter presented a picture of issues pertaining to the mechanism of corporate governance from the perspective of developing nations.

This chapter also seeks to elaborate the various general themes regarding CG and its practices in the contemporary organizations in developing economics. "Internal mechanisms include the board of directors (e.g. board



size, board sub-committees, CEO duality and non-executive directors) and ownership structure (e.g. large shareholders or concentrated ownership, the identity of shareholders and managerial ownership)". Other board sub-committees have not been discussed here due to their scope beyond this study which is under discussion. More specifically this chapter has reviewed literature pertaining to the impact of foreign investors on firm performance by digging deep the relevant field of available knowledge in an emphatic fashion. On the basis of these mechanisms, existing study tends to propose a research framework and specific variables which need to be further tested relevant to the CG mechanisms. The onward section of this study highlights the environmental concerns explicitly particularly in local context.



Chapter 3 Research methodology

This study attempts to find out the impact of CG practices on the financial performance of Jordanian firms in the industrial and services sector (non-financial sector). More specifically, this study deals with issues of governance. The first section of this chapter elaborates on research philosophy and methods applied. This chapter also deals with the data analysis techniques and various regression assumptions, such as multicollinearity, heteroscedasticity and serial correlation. It also provides guidelines regarding detection of regression-related issues/problems and their rectification.

On the other hand, another section of this chapter deals with the CG mechanism and its effect on the financial performance of firms listed on the ASE for a period of seven years ranging from 2012 to 2018. Hence, this section provides a detailed illustration of data used in this study to investigate various CG-related mechanisms and financial performance. Initially, sources are illustrated which have been consulted for data collection. Next to this, the sample and its selection procedure are elaborated on. Study variables about corporate governance mechanisms (board structure and ownership structure) along with measures of financial performance (ROA/ROE) are explained, and, at the end, control variables are discussed. The data sources for various variables and their measurement are also considered.

3.1 Research philosophy

According to Burrell and Morgan (2005), with respect to the selection of suitable research, the paradigm is crucial in every research study. Philosophical orientation in every study pertaining to social science holds a pivotal role. As this study investigates the impact of the CG mechanism on firm performance through hypotheses, the positivism paradigm has been chosen as the most suitable research paradigm as it allows the researcher to investigate and test the hypotheses empirically (Saunders et al., 2019).



Burrell and Morgan (2005) stated that positivism helps to draw causal inferences among the variables and their relationship. The deductive approach is linked to positivism, which helps to illustrate the causal relationship among variables and, more specifically, it helps to generalise the results based on the sample (Saunders et al., 2016). Keeping in mind the assumed relationships/hypotheses developed earlier, this study utilises the deductive approach due to the reasons mentioned below:

- The deductive approach follows scientific principles in ascertaining the nature and outcome of the event.
- It helps to conclude based on hypotheses testing based on existing theories.
- This approach avoids developing or building a new theory.
- The deductive approach helps to infer causal relationships among constructs.
- Quantitative data is used in a deductive approach.
- It is a structured approach and devises clear parameters.
- This approach ensures the independence of the researcher. It makes
 the study free from researcher bias because analytical procedures are
 followed, and the personal opinion or observation of the researcher is
 not incorporated.
- Under this approach, it becomes easy to generalise the results due to the adequate sample size (Saunders et al., 2019).

Andres and Vallelado (2008) stated that, due to the scientific nature of the deductive approach, it follows a series of steps, which are as follows:

- Based on existing theories, the first testable hypothesis/hypotheses are developed.
- Clarification regarding the testing of hypothesis/hypotheses is made.
- The next stage requires operationalising of study variables, and their measurement is in quantitative terms.



- The next step requires testing the operationalised hypothesis/hypotheses and inferring causal relationship among variables;
- The final step requires testing the study results and based on these results and later providing confirmation/endorsement or even some modification in the theory based on results (Bell, Bryman and Harley, 2018).

According to Burrell and Morgan (1994), the deductive approach uses a functionalist paradigm and study objectives are set following the deductive approach keeping in mind the population of the study. Hence, in this study objectives have been developed, keeping in mind the notion that the impact of the corporate governance mechanism on the financial performance of the firm can be measured empirically through applying research analysis tools. Accordingly, positivism helps to explain change or occurrence, and the deductive approach describes the causal relationship among study constructs. Positivism also helps to identify and predict relationships in replicated scenarios. Thus, positivism can contribute better to designing a research strategy in testing hypotheses (Hussey and Hussey, 2009; Saunders et al., 2019).

In summary, this study has followed a deductive approach, which is more appropriate. It is essential to mention that most of the gurus of quantitative research are of the same point of view regarding testing the theoretical framework utilising quantitative tools. Moreover, a deductive approach is also suitable in cases where the researcher is not interested in introducing a new theory; instead, the researcher is quite keen to test the existing field of knowledge in that specific field of theoretical understanding.

3.2 Research methods

According to Punch (1998) on selection of appropriate research, the approach is crucial, keeping in mind the research issues. Widely, around the



globe, two research methods are used, one is quantitative and the other is qualitative. A qualitative research method usually presents a descriptive approach. It uses non-numerical data to represent a given phenomenon (Berg, 2004) while, according to the perception of Babbie (2012), the qualitative method is used to assess the variations in attitudes and behaviours occurring in social processes over a given period. Conversely, Hussey and Hussey (2009) and Bryman (2016) postulated that a quantitative approach uses statistical analysis based on numeric facts and figures and provides a stronger argument based on reliability and validity to generalise the results.

Additionally, a quantitative approach can use a large amount of quantitative data for a longer period, which can increase the credibility and generalisability of results (Berg, 2004). Alternatively, some researchers recommend using both methods for better results. However, some fundamental issues are attached to the qualitative research approach, which suffers from a number of problems, such as the issue related to small sample size, due to which it is quite inconvenient to draw results considering the whole population (Hakim, 1987). Also, there are problems related to measures of reliability in qualitative research methods as well as transparency (Berg, 2004). In addition, qualitative methods consume a massive amount of time, but might produce less suitable explanations of research phenomena (Berg, 2004; Cohen, Manion and Morrison, 2017).

Hence, due to the difficulties and weaknesses attached to qualitative data, the present study used a deductive positivism approach and hypotheses have been developed based on available and pre-tested theories. Empirical findings based on quantitative data demonstrate the acceptance or rejection of hypotheses. For testing the hypotheses, the present study employed regression as a major tool for data analysis (Ardalan, 2012). Hair et al. (2009) stated that regression is a suitable tool for data analysis in cases where a single outcome variable is linked with one or two predictors. This study selected multiple regression analysis as the primary instrument for hypotheses testing because it is amongst the most common methods and



previous researchers have endorsed it in testing the relationship of various CG mechanisms and their relationship with the financial performance of the firms (Claessens and Yurtoglu, 2013; Peoples, 2020).

3.2.1 Panel data

Cross-sectional, time series and panel data are various types of quantitative data used for empirical investigations. In cross-sectional data, different values of one or more variables are collected at the same point in time while, in time series data, observations or values for one or more variables are collected for more than one time period. On the other hand, panel data uses the same cross-sectional units, which are collected over a time period. Simply put, panel data uses time dimensions and spaces at the same time (Gujarati, 2003). Previous literature provides support regarding usage of panel data regression approaches (Gujarati, 2003; Greene, 2012; Hall and Wall, 2019). Using panel data provides various advantages which are outlined here:

- It uses prior data.
- It combines both time series and cross-sectional data.
- Variables showing high linearity can be identified easily and their removal from the data becomes easy.

In order to deal with the error term in panel data, two models are usually used, such as the fixed effects model and the random effects model. These two models follow different assumptions to deal with error terms. The fixed effect model assumes that the individual effect term is constant while the random effect model follows a different approach, and it assumes that an individual's effect is random. According to the perception of Greene (2008, p. 285), a simple panel data regression model can be given as:



$$\gamma_{it} = X'_{it}\beta + Z'_{i}\alpha + \varepsilon_{it}$$

Where:

Yit is the dependent variable.

Xit are predictors.

B and α are coefficients.

Z'_iα is an unobserved individual-specific effect.

i and t are the indices for individuals and time.

 \mathcal{E}_{it} is the error term.

 $Z'i\alpha$ denotes heterogeneity pertaining to individual effect in the above model while Z'i denotes a set of individual or group-specific predictors. These can be either observed variable or unobserved variable, which are considered constant over a specific time, t (Greene, 2008; Townsend and Saunders, 2018).

3.2.2 Pooled regression

If Z_i contains only a constant term, then OLS provides more consistent and efficient estimates regarding constant (α)/intercept coefficient/slope vector usually denoted by β . Hence, the original model reduces to:

$$Y_{it} = X'_{it}\beta + \alpha_i + \varepsilon_{it}$$

3.2.3 Fixed effect

The fixed effect model provides a consistent estimate in a situation when Z_i is unobserved but correlated with X_{it} because this situation leads to biased estimation through β , hence, the model reduces to:

$$Y_{it} = X'_{it}\beta + \alpha_i + \varepsilon_{it}$$



Where:

 Y_{it} = Outcome variable (here i = entity and t= time).

 β = Coefficient of the predictor.

X_{it=} Predictor.

 α_i (i=1.... n) = Unknown intercept for each entity in the model.

Where $\alpha_i = Z'_i \alpha$ embodies all observable and specified means. According to Greene (2012) and Pallant (2020), the fixed effect does not change. Each entity possesses its features which might influence independent variable/variables. It is the rationale behind the assumption of correlation among the error term and independent variable/variables. Hence, in this situation, the fixed effect helps to remove time-invariant characteristics from independent variables in panel data regression from the predictor variables.

3.2.4 Random effect

A random effect model is used in cases where unobserved individual heterogeneity can be uncorrelated with variables in the model, and this scenario model reduces to:

$$\gamma_{it} = X'_{it}\beta + \alpha + u_i + \varepsilon_{it}$$

Fixed effect and random effect models can be distinguished based on unobserved individual effects, simply by checking whether these effects are stochastic or not (Greene, 2012). Further, Bell, Bryman and Harley (2018) recommended that fixed and random effect models can be distinguished based on unobserved individual effect.

It is believed that in the fixed effect model the random effect reduces to zero and based on this argument the fixed effect model is considered to be a restricted version of the random effect model and using random effect is preferable due to its general nature. However, the larger number of parameters inherited in the random effect specification can result in loss of



efficiency in some cases when data does not support random effect. Hence, it is suggested that the random effect should be tested against a fixed effect. The advanced treatment of both models can help to distinguish their usability, and a tool called the Hausman test can be used in this scenario. In this case, assumptions can be followed to use/choose random effect estimation, i.e. unobserved heterogeneity in the model should not be correlated with predictors. Fixed effects models have been used in this study based on statistical results given in Chapter four.

According to Greene (2012), specifications of the Hausman test are followed by most researchers using random and fixed effect models. This test is usually used to detect violation of random effects modelling assumptions. "If there is no correlation between the independent variable(s) and the unit effects, then estimates of β in the fixed effects model (β_{RE})." Thus, the Hausman test statistic (denoted by H) is used to measure the difference between two estimates; an econometric model can be given as:

$$H = (\beta_{RE} - \beta_{FE})'[Var(\beta_{FE}) - Var(\beta_{RE})]^{-1}(\beta_{RE} - \beta_{FE})$$

Here

 β_{FE} = Coefficient of estimates and time-varying covariates.

 β_{RE} = Estimated coefficients of the random effects model.

 $Var(\beta_{FE})$ = Asymptotic variances of large sample size.

While testing the null hypotheses, chi-square with degrees of freedom is equal to the number of repressors present in the model. At p < 0.05 (denotes conventional levels of significance), two models show the difference, which is large enough to reject the null hypothesis. Here, the random effect model is rejected in favour of the fixed effects model.

Contrary to this, if the Hausman test shows a p-value >.05, then the situation indicates that there is no statistically significant difference and using random



effect will provide an unbiased estimation and using random effect over fixed effect is preferred under such circumstances. Simply, in most applications, the value of correlation among covariates and unit effects is not equal to zero. Hence, in the case when the null hypothesis is not rejected due to the Hausman test, there might be an exact correlation which is higher than zero and, and in this case, random effect estimation becomes biased. However, it can also be assumed that the Hausman test does not have necessary and sufficient statistical power. While, in the case of using the model of random effects, biasness will be still there in the estimation of coefficient (β), even if the Hausman test does not reject the null hypothesis. Admittedly, in many instances it will be better to use random effect (even biased estimator) against fixed effect (also unbiased estimator) if the random effect model provides a necessary and sufficient variance reduction against the fixed effect model (Bell, Bryman and Harley, 2018).

While using the random effect model, time-invariant variables can also be added into the model. Contrary to this, when using the fixed effect model, time-invariant variables are absorbed through the intercept. Using random effects assumes that error terms of models are not correlated with the independent variables, and this situation helps to imagine time-invariant variables as predictor variables. Hence, removing such variables from the model can lead to biasness in the estimation model and, based on random effects, it becomes very convenient to generalise the results beyond the sample size used (Kohler and Kreuter, 2005; Cohen, Manion and Morrison, 2017).

3.2.5 GLS estimator

In the regression model, when unknown parameters are estimated, the generalised least squares (GLS) technique is employed. This technique is used in the case when heteroscedasticity exists or there is directly present unequal variance in observations. GLS is also used in situations where a certain degree of correlation exists among observations. This fact corrects omitted variable biases even if there is autocorrelation and heteroscedasticity



in the pooled time-series data. This study used GLS to test random effects models because this technique provides the freedom to the researcher to investigate variation in pooled time-series data (cross-sectional observations over time) (Gaur and Delios, 2006; Peoples, 2020). For a given correlation matrix, Ω , GLS is given as:

$$\hat{\beta} = [X'\Omega^{-1}X]^{-1}[X'\Omega^{-1}y]$$

While Ω can be expressed as:

$$\Omega = \Sigma \otimes I$$

Where Σ is n x matrix $[\sigma_{ij}]$, then:

$$\Omega^{-1} = \ \Sigma^{-1} \otimes I = \begin{bmatrix} \sigma^{11}I & \sigma^{12}I & \cdots & \sigma^{1n}I \\ \sigma^{21}I & \sigma^{22}I & \cdots & \sigma^{2n}I \\ \vdots & \vdots & \vdots & \vdots \\ \sigma^{n1} & \sigma^{n2}I & \cdots & \sigma^{nn}I \end{bmatrix}$$

Where σ_{ij} denotes the $_{ij}$ element of Σ^{-1} :

$$\hat{\beta} = \left[\sum_{i=1}^{n} \sum_{j=1}^{n} \sigma^{ij} X'_{i} X_{j} \right]^{-1} \left[\sum_{i=1}^{n} \sum_{j=1}^{n} \sigma^{ij} X'_{i} y_{i} \right]$$

3.2.6 Sample

The present study used a sample of non-financial firms listed on the ASE, covering a period of seven years ranging from 2012 to 2018. This list was obtained from the website of the official page of the ASE website. The ASE deals with two sectors, one relates to financial firms while the other relates to non-financial firms. The financial sector generally covers those firms which offer financial services to companies such as banks, insurance companies, diversified financial services and firms dealing with real estate businesses (Hall and Wall, 2019). On the other hand, firms in the non-financial sector include industrial and service sector organisations. A snapshot of firms listed on the ASE as follow in table 3.1 below:

Table 3.1 Non-financial sectors on the Amman Stock Exchange

Description	Non-financial sector		Total
	Industrial	Services	TOtal
Number of firms	50	47	97
Percentage of	51.55%	48.45%	100%
firms	J1.JJ/0	40.43 /0	100%

Source: Self-generated

Secondary data is collected for the study instead of primary data or carrying out an empirical research. The main reason to collect secondary data is because it is mandatory for the public companies to prepare a full set of financial statements at the end of their accounting year and publish in its annual report. The full set of financial statements consists upon income statement, balance sheet, cash flow statement and statement of changes in equity (Melville, 2019). Accountants need to prepare the full set of financial statements and auditors are responsible to audit the financial statements before publication to ensure the faithful representation of the financial statements. These financial statements contains all the financial data related to the financial performance and financial position of the public companies.



The data required for this study is related to the financial performance, financial position and corporate governance mechanisms of the public companies. This data is available in the annual reports of all the public companies as it is mandatory for the public companies to publish it at year end, so the researcher has decided to rely on secondary data. The reliability of financial data is not an issued because auditors needs to audit the financial data before it is published for the general public. Another reason to rely on secondary data instead of primary data is that if researcher has contacted the companies to collect primary data then they should have provided the annual reports to the researchers. Furthermore, other researchers who have conducted the similar research in Western countries have collected the secondary data because the data is available and there is no reliability issue with the financial data. Due to the above reasons the researchers have decided to collect the secondary data instead of primary data.

Data used in this study was obtained from three sources: the first source is the Orbits database; the second source is firms' annual reports, while the official page of the ASE website was used as the third source of data in this study. Data pertaining to CG-related variables such as details regarding BODs was collected through the Orbits database, and the official page of the ASE website, while data pertaining to other financial variables was manually obtained through annual reports. According to Fraser et al. (2006), a firm's annual reports provide more accurate data as compared to other data sources pertaining to secondary data, because data used in annual reports is highly reliable, which ensures quality reporting. In order to reduce the errors related to data entry while copying data from the annual reports, the data entered was marked, and all the entries were checked again to ensure that was correct. Both the databases (i.e. Orbits database and annual reports) provide necessary financial information about firms. Additionally, these two databases also provide information regarding the names of auditing firms. A total of 97 firms (industrial and services-related firms) are listed on the ASE. Firms related to various industries and listed on the ASE are as follows in table 3.2 below:



Table 3.2 Firms related to different industries

Sector	Industry	Number of firms
Services	Health Care services	04
	Educational services	06
	Hotels and tourisms	09
	Transportation	10
	Technology and communication	02
	Media	01
	Utilities and energy	04
	Commercial services	11
Industrial	Pharmaceutical and medical	04
	Chemical Chemical	08
	Printing and packaging	01
	Food and beverage	09
	Tobacco and Cigarette	02
	Mining and extraction	11
	Engineering and construction	07
	Electrical industries	03
	Textiles, Leathers and Clothing	05
	97	

Source: Self-generated

The study is based on 95 companies because the data for two companies out of the total 97 companies was not available. The current study excluded firms from the financial sector; the reason for the exclusion of these firms is the presence of various government regulations and regulatory bodies, which provides a different controlling mechanism for financial sector firms (Abed et al., 2012). Hence, using the financial sector in this study was not a reasonable choice. Financial sector firms also follow a unique style in reporting their financial information, which was another reason for their removal from the sample size of this study (Claessens et al., 2006; Al-Najjar



and Taylor, 2008; Estrin et al., 2009; Al-Fayoumi et al., 2010; Bell, Bryman and Harley, 2018).

Keeping in mind the patterns of available literature (Yermack, 1996; Cheng et al., 2008), the present study employed the same criteria for sample selection. Yermack (1996), Cheng et al. (2008) and Saunders et al. (2019) stated that data for several subsequent years is required for panel data analysis. Further, financial data up to the year 2018 has been included in this study due to 2018 being the most recent year for which data was available at the time of data collection.

This section elaborates on the methodology and the methods applied due to the availability of pre-tested theories and hypotheses developed. For testing of hypotheses, multiple regression analysis was selected as a significant tool for the study; in order to capture the effects of firm and time-specific heterogeneities, panel data models can be specified as fixed effects or random effects. Moreover, this chapter seeks to investigate some specification tests which might tend to influence corporate governance variables, which can result in statistical problems related to predictors in the regression model.

3.3 Variables and measurement of variables

3.3.1 Performance variables

Various researchers have measured firm performance differently (Cochran and Wood, 1984; Ittner and Larcker, 2003). The majority of researchers have measured it by using Tobin's Q (TOQ) (Yermack, 1996; Weir et al., 2002; Akbar et al., 2016; Anderson et al., 2020), return on asset (ROA) (Yermack, 1996; Kiel and Nicholson, 2003; Alqatan, Chbib and Hussainey, 2019), return on equity (ROE) (Adjaoud et al., 2007), return on investment (ROI) (Adjaoud et al., 2007) and net profit margin (NPM) (Bauer et al., 2004).



A firm's performance can be assessed in two dimensions, i.e. one is based on market performance, and the other is based on accounting performance. According to the perception of Daily et al. (2003), accounting-based measures of the performance show the current accounting (financial) performance of the company, while market-based measures of performance indicate the investor's perceptions of the firm's potential market performance. Although both measures are used in practice, various researchers have criticised them; for instance, Hiffa and Hudaib (2006) mentioned that no best indicator of financial performance has been recommended by previous literature, and studies lack a consensus regarding the best measure of performance. Further, they stated that every measure has its pros and cons, and hence no particular measure can be considered as a proxy to measure the financial performance of the firm (Al-Ahdal et al., 2020).

Return on asset (ROA) was initially recommended by Demsetz and Lehn (1985) as an indicator to measure the year-to-year fluctuations in business return, which is based on actual business performance, contrary to this, stock market rate of return has also been recommended by various researchers which show the expected performance of the business. This idea has been used widely by the proponents of the corporate governance mechanism (Gompers et al., 2003; Haniffa and Hudaib, 2006; Braendle, 2019), and they have used this concept in their studies. Generally, the accounting-based measures of firm performance do not consider prospects; instead, they show comprehensive indicators of current accounting performance. Conversely, market-based measures cannot present a vivid and real picture of the financial performance of firms where the economy is in its developing phase or in emerging markets because under such circumstances most of the firms operate under debt-financing. Hence, market-based measures of performance do not represent the actual performance of the firm (Kumar, 2004; Detthamron, Chancharat and Vithessonthi, 2017).

According to Gompers et al. (2003), the market share price of companies indicates market value with the signal that the capital market is working



efficiently, as per the Efficient Market Hypothesis (EMH). Due to its status as an emerging market, the stock market in Jordan is yet to be developed in such a way that it could compete with developed markets. For example, if the information is publicly disclosed, it will result in and share prices will fluctuate in accordance with the available information. Contrary to this, accounting performance indicates greater internal control; however, managers have very little control over market valuation because market valuation and stock returns are related to each other (Grossman and Hoskisson, 1998; Bhaumik et al., 2019).

Black et al. (2006) stated that inside and outside mechanisms of CG have different impacts on firm performance. According to Wulf (2007), there is a direct relationship between accounting performance and the firm's strategies. For instance, almost 80% of researchers have used accounting-based measures (ROA and ROE) in their studies while measuring firm performance, while many researchers have used ROA and ROE as a measure of financial performance when investigating the impact of CG mechanism on firm performance (Kiel and Nicholson, 2003; Razzaque and Mather, 2020).

Tobin's Q is an indicator of the firm's market performance defined as the market value of equity divided by replacement cost (Yermack, 1996; Shu and Chiang, 2020). According to Haniffa and Hudaib (2006), Tobin's Q indicates the effectiveness of management regarding the proper utilisation of the firm's resources to maximise the shareholders' wealth. Various researchers have used different methods while measuring Tobin's Q; for instance, Yermack (1996) recommended using market value and replacement cost, while Booth and Deli (1996) recommended using a combination of market value and total assets. Using different measurement styles to measure the Tobin's Q will produce different results even for the same period due to change in estimation method and valuation of assets. Although Tobin's Q can be measured through various methods and various perspectives, difficulties are embedded in the measurement/calculation of Tobin's Q. For example,



calculating the replacement cost is not an easy task because companies do not report it in their annual reports (Shahid and Abbas, 2019).

Various researchers have recommended different advantages of measuring performance based on accounting-based measures. Simply, higher values of ROA and ROE indicate that the management of the firm is using its resources (assets and equities) effectively to maximise the shareholders' wealth. Measuring performance based on ROA and ROE handles the firm size very comfortably. Further, it becomes elementary to compare the performance of the firms based on ROA and ROE (Lev and Sunder, 1979). As recommended by Demsetz and Lehn (1985), year to year changes can be considered in an entirely better way based on ROA and ROE as compared to the stock market return. The reason behind this might be that the stock market rate of return indicates an expected change in firm performance while ROA and ROE indicate the more realistic and actual condition of the firm (Akbar et al., 2016).

Nevertheless, many researchers have criticised the usage of accountingbased measures of firm performance based on different aspects. For instance, Ross et al. (2008) stated that ROA and ROE have been used since long time, but the level of earlier earnings cannot show the real picture of future earnings/profits. Similarly, Krivogorsky (2006) and Jacoby et al. (2019) stated that ROA and ROE are based on the principles of historical cost accounting and hence, in this situation, they do not show the current condition of the equity market. Another concern, pointed out by Alexander et al. (2007), indicates an alarming situation as using different accounting methods and techniques can bring different results while using the same ROA and ROE in different years. In spite these, previous researchers have pointed out that accounting-based measures of firm performance (ROA and ROE) do not consider risk (Ross et al. (2008). Additionally, it is challenging to assess the organisational environment and industry differences while using ROA and ROE in determining firm performance because these measures do not reflect the relationship of the firm and customer satisfaction (Alexander et al., 2007; MacAulay, Dutta and Hynes, 2020). However, to reduce the

potential impact of these weaknesses and limitations, when measuring the performance of the company using accounting-based measures, we have considered some control variables in this study. These control variables help to justify the use of accounting-based measures in assessing performance (Shu and Chiang, 2020).

According to Demsetz and Lehn (1985) and Mehran (1995) ROE indicates the financial performance of the firm based on return on shareholders' investment and, based on this, it can be considered as an essential ratio which reflects the investors' concern. On the other hand, considering the ROA as a measure of financial performance considers the firm resources to carry out the firm activities. As per the contents of agency theory, there can exist a conflict as managers can misuse the resources of the company to protect their own interests, instead of securing the shareholders' interests. Hence, using ROA and ROE as a measure of the performance shows how managers are efficiently and effectively using the firm's resources, and a lower level of ROE and ROA will highlight an ineffective managerial team. Thus, ROA and ROE can be considered as reliable measures from the perspective of shareholders while measuring the firm performance. Keeping in view above lines/argument present study has considered ROE and ROA as measures of firm financial performance on the basis of accounting-based measures.

Return on assets (ROA) shows the degree of managerial efficiency in utilising the firm's resources/assets to generate profit/earning. In some cases, ROA is considered as Return on Investment (ROI). Mathematically, it is calculated as:

Return on Assets (ROA) = (Net Income) / (Total Assets)

On the other hand, Return on Equity indicates firm performance based on the amount invested by the owners/investors or how a firm is generating profit in comparison to the amount invested. This is also called Return on Net Worth. Mathematically it is calculated as:



Return on Equity (ROE) = (Net Income) / (Total Equity)

To calculate the ROA and ROE for sample firms, this study used the DataStream database.

3.3.2 Control variables

Control variables have been considered in this study to explain changes in firm performance. From the scrutiny of scant literature, it has been observed that various researchers have used different control variables. Hence, this study considered already used control variables such as sales growth, capital expenditure, firm size, leverage, research and development expenditure and liquidity (Daines, 2001; Gompers et al., 2003; Black, 2001; Anderson et al., 2020). Despite using the above-mentioned control variables, it can be expected that other factors also have the potential to impact the firm performance. However, the existing body of literature does not show any formula for considering the number of control variables in any study. Thus, keeping in mind the practice of previous studies, this research has adopted mainly previously used control variables

3.3.2.1 Sales growth (SGH)

According to Kim (2006), those firms that show higher growth tend to seize more opportunities, and market valuation of the company is increased with growth opportunities. Firms which aim to grow need a large amount of money/financing, either internal or external, which drives firms to follow improved governance practices in order attract investors (Beiner et at., 2006); this also reduces capital cost. Literature shows a direct and positive association between firm performance and growth opportunities, which are measured based on year-to-year sales growth (SGH) (Gompers et at., 2003, Drobetz et at., 2004, Cui et at., 2008, Henry, 2008; Alqatan, Chbib and Hussainey, 2019). Hence, this study considers sales growth as a control variable.



3.3.2.2 Firm size (FSE)

Although firm size has been considered as a control variable in various studies, there exists a lack of consensus regarding the relationship of firm size and firm accounting performance (Nenova, 2003; Durnev and Kim, 2005). According to the findings of Short and Keasey (1999), larger firms can create entry barriers for other firms due to access to specific resources; this situation can generate a better accounting performance. According to Jensen (1986), firm size can be used as a proxy to assess agency problems. According to these findings, people in managerial posts tend to increase the size of firms so that they can enjoy access to more power and large resources. According to the findings of Fama and Jensen (1983) and Boone et al. (2007), an increase in firm size brings diversification, which leads firms towards natural complexity. This increased complexity increases the need for more advice on the board.

Further, it is believed that larger firms follow complex operations so that their strategies can be better pursued. According to Serrasqueiro and Nunes (2008), large firms show a better performance because they have a better opportunity to raise funds, as well as following diversified strategies in their operations. One more attribute of larger firms is the presence of a wide variety of expertise which enables larger firms to make better decisions, which increases their performance (Black et al., 2006; Bhaumik et al., 2019).

Some researchers have also documented the difficulties which are faced by larger firms such as more inspections and scrutiny by the regulatory authorities, another stakeholder (Nenova, 2003; Garen, 1994). Under such difficulties, it becomes difficult for family-owned firms to extract private profits (Nenova, 2003). According to a study by Agrawal and Knoeber (1996), with the increase in firm size, i.e larger firms show decreased accounting performance. The reason for this reduced performance might be the reduced control by management due to the larger firm size, while small firms have improved control over operations due to less complexity and smaller size (Akbar et al., 2016).



According to Garen (1994), larger firms only have to face low costs in complying with CG codes, while small firms bear much higher costs due to their small size. However, larger firms can take higher costs relevant to the CG mechanism as larger firms have to face a higher level of media scrutiny as compared to smaller firms (Garen, 1994). Similarly, agency theory (Jensen and Meckling 1976) also indicates that larger firms tend to face increased agency costs due to their larger size because larger firms need more controls and checks to restrain the managers from being opportunistic. In addition to this, larger firms need a more significant amount of internal control to bring an alignment between management and shareholders, which also raises the monitoring cost (Jensen and Meckling, 1976). Keeping in mind the established literature, this study tends to assess the firm size based on taking the natural logarithm of total assets (Elsayed, 2007; Topak, 2011; Al-Matari et al., 2012; Lehn et al., 2009; Akbar et al., 2016). At the same time, calculations regarding total assets of firms were drawn directly from the balance sheet provided by the Orbis database.

3.3.2.3 Capital expenditure (CAPX)

In order to gain a competitive advantage, firms tend to invest in innovative initiatives to introduce new products and services (Back et al., 2004; Taufil-Mohd et al., 2013). Such new products and services then provide monopolistic positions to the firms and firms enjoy premium price profits due to investment in technology-oriented decisions (Barton and Wong, 2006). While investing in future projects to generate higher performance requires a lot of amounts, and hence financial performance for the current period is shattered due to such investments (Weir et al., 2002). Therefore, it becomes necessary for technology-oriented firms to follow the code of corporate governance mechanism to secure and safeguard their intangible assets (Durnev and Kim, 2005). keeping in mid the existing quantum of literature This study considered that investment opportunities measured in the shape of the ratio of capital expenditure to total assets (CAPX) tend to influence firm performance in adverse ways (Durnev and Kim, 2005, Black et at., 2006; Tariff, 2006; El-Faitouri, 2014).



3.3.2.4 Leverage (LEE)

The term is defined as the ratio of long-term debt to total assets. Some researchers have documented a negative impact of leverage on firm performance while others have reported a positive effect. The positive effects of leverage can be noted in the case when lenders involve themselves in the monitoring process. According to Jensen and Meckling (1976), leverage can play a crucial role in reducing agency problems as an internal corporate governance mechanism, particularly free cash problems.

According to the perception of Jensen (1986), if the level of external debt increases, it can bring a positive impact on the performance of the firm. A higher level of debt will minimise the chances for the management to make decisions according to their discretion, which will reduce the agency problems. For instance, according to the opinion of Jensen (1986), a higher level of debt might lead to a more disciplined management by disciplining the managers to use available free cash flows for non-profitable projects. Where the management is obligated to make periodic repayments of interest and principal, this will ensure proper usage of free cash flow, and managers will not become opportunistic. According to Stiglitz (1985), effective control over management can be ensured through lenders rather than depending upon owners. Moreover, Modigliani and Miller (1963) also favoured the presence of high debt in the firm's capital structure. They expected a positive relationship could be observed among firm performance and leverage through tax shield benefits. Similar findings have been reported by Agrawal and Knoeber (1996) regarding a positive relationship between firm performance and a higher level of debt.

Contrary to this, Myers (1977) stated that the presence of a higher level of leverage might negatively affect the performance of a company, because increasing the level of leverage will undermine the firm's ability to obtain another debt. Such a situation will lead to the firm losing its investment opportunity, which will reduce its capability to generate profit in the future. Similarly, Myers (1977) and Stulz (1988) argued that the presence of a



higher level of leverage can hamper firm performance due to higher financial risk. Additionally, they stated that a higher level of leverage would result in increased interest payments, and increased monitoring cost will ultimately reduce the firm's financial performance (Andrade and Kaplan, 1998; Dethamrong, Chancharat and Vithessonthi, 2017). Data related to firms' level of leverage was extracted directly from financial statements provided by the Orbits database.

3.3.2.5 Research and development expenditure (R&D)

Companies tend to invest in new products to gain a competitive advantage through these new products or services. Once these new products or services become available, companies can demand higher prices and generate profitability in the long term (Borghesi et al., 2007; Sheu and Yang, 2005). Additionally, a new invention can also be considered as a barrier for rivals, preventing them from entering the market and attracting new clients (Sharar, 2006). Following prior CG studies (see, for example, Vafeas and Theodorou, 1998, Dahya et al., 2007; El-Faitouri, 2014), this study uses R&D as a control variable.

Also, financially restricted R&D corporations bear larger risk compared with firms which are not restrained, and, because of this, they must halt their R&D ventures. Li (2011) analysed the affiliation among stock reversion, R&D investment and financial constrictions. Outcomes exposed the robust liaison of financial restraints and stock output. Contrary to the investment that is capital in nature, R&D investment is minor, elastic and flexible. It is a crucial task to assess finance by small firms, as evident from the literature, on the whole (Shankit and Abbadi, 2011). Some authors visualise the way and hardness of financial constraints as depicting institutional difference like SME Bank while lending face severe constraints when they are in distress while fee in the comfort zone in normal times (Baum, Schäfer, & Talavera, 2011; Akbar et al., 2016).



3.3.2.6 Liquidity (LIY)

According to Jose et al. (1996), a company's survival depends upon the liquidity of the firm because liquidity has a significant association with the firm's sales and growth. Liquidity also has an impact on the risk level of the firm because it reduces financial costs for the firm. A higher level of liquidity indicates firm development and shows that the firm has a strong position in the market, as well as indicating its achievement. According to Fang et al. (2009), a higher level of liquidity helps to manage the agency problems, which tends to lead towards informed share prices. Thus, a positive relationship between liquidity and firm performance can be predicted. Liquidity is termed as the ability of a firm to meet its short-term obligations and financial distress.

Liquidity is extracted from the balance sheet figures. Keeping in mind the established literature (Fang et al., 2009; Jose et al., 1996), the present study measured the liquidity of firms through current ratio (CR). This ratio is calculated through current assets (CA) and current liabilities (CL). Liquidity ratio shows that a firm with higher liquidity can manage any external shock related to financing and liabilities. This fact also helps to minimise financial distress. Additionally, a higher level of liquidity indicates an alarming situation where the firm has to bear higher opportunity cost because, when a firm keeps in hand a higher level of current assets, it reduces its ability to invest greater profits (Shahid and Abbas, 2019).



3.3.3 Corporate governance variables

3.3.3.1 Board size (BSE)

According to Ahmed et al. (2006) and Bennedsen et al. (2008), board size (considered as BSIZE in this study) is defined as the number of directors who are on the board. Previous studies have shown mixed findings regarding the relationship between board size and accounting performance of the firm. Hermalin and Weisbach, (1991), as well as Jensen (1993), found that small boards bring better performance for firms. The increased agency cost results in a lower performance for firms. Similarly, boards with a large number of members lose their ability to execute proper monitor and control functions; this provides managers with space and they tend to follow their self-interests and forego the interests of shareholders (El-Faitouri, 2014).

Hence, in large boards the CEO tends to control the board, which also has a negative impact on the financial performance of the firm. However, in spite of these arguments, some researchers have argued in favour of large board size and reported a positive relationship among firm performance and large board size (Dalton et al., 1998; Lehn et al., 2009). This view consists of the resource dependency theory because a larger board size improves external linkages (Hillman and Dalziel, 2003). Moreover, larger boards provide an opportunity for directors regarding the exchange of more highly qualified counsel through external linkages. Larger boards also ensure access to resources through improved external ties. Such resources can be new and advanced technologies, new markets and raw materials. Larger boards facilitate diversity among board members and ensure integration of knowledge and skills, which leads to improved and informed decision making, which might result in a higher level of firm performance (Lehn et al., 2009; Razzaque, Ali and Mather, 2020).

Hence, there is a lack of consensus among previous researchers regarding the impact of board size on firm performance. Therefore, the existing study explores the relationship between board size and financial performance of



the firm. In this study, number of board directors have been extracted from the Orbits database.

3.3.3.2 CEO duality (CEOD)

Proponents of agency theory (Berle and Means, 1932; Jensen and Meckling, 1976 and Eisenhardt, 1989) recommend that, to minimise the agency problem, ownership and control in the firm should be separated. This will improve firm performance. Hence, agency theory supports the argument that the roles of CEO and chairman should be separated to increase the independence of the board and management and to ensure improved monitoring and overseeing. This situation (theoretically) will result in improved firm performance (Jensen, 1993). Contrary to this, stewardship theory recommends duality on the board, i.e. roles of CEO and chairman should not be separated. The dual role of CEO and chairman will ensure effective management because it will bring unity of command. If the responsibilities and decisions are restricted to one person, it will result in improved performance through positive impact (Dalton and Kesner, 1987; Arosa et al., 2012). In addition to this, duality on the board tends to reduce misconceptions, because the chairman and CEO is the same person, which will reduce inconsistencies (Brickley et al., 1988; Algatan, Chbib and Hussainey, 2019).

As per recommendations of the JCGC (2006), the CEO and Chairman should be assigned separate responsibilities and duties to avoid any conflict of interest. This will ensure adequate supervision and monitoring. Many researchers in past studies have investigated the roles of Chairman and CEO under duality through a dummy variable (Abor, 2007; Bozec, 2005; Arosa et al., 2012; Haniffa and Cooke, 2002; Sheikh et al., 2012; Akbar et al., 2016). Hence, to document the impact of separating the roles of chairman and CEO in Jordanian companies, this study used a dummy variable to measure the CEO duality, keeping in mind the previous research. If the chairman holds the designation of CEO, a code of one was allotted,



and otherwise zero was considered. This data was collected from the Orbits database.

3.3.3.3 Non-executive directors (NEDs)

According to the findings of Fama and Jensen (1983), internal managers hold the dominant position and their performance is increased in the case when they initiate decisions and exert maximum control. However, in the modern day where firms have to compete on a day-to-day basis, it is difficult to handle between decision management and decision control. This situation recommends that NEDs should be appointed in firms to ensure board independence. This will ensure the separation of control and management functions. Moreover, conflict among internal managers can be mediated through NEDs. NEDS can also help to improve the relationships among stakeholders, which will enable them to better perform their monitoring function as compared to executive directors (Tricker, 2019).

In addition to this, NEDs can constructively criticise the firm's policies due to their independence (Jensen 1993), and they can impart their opinion without any fear. Also, NEDS can help to minimise information asymmetry among the stakeholders. This situation will lead to increased firm performance due to a reduction in agency problems, because independent directors can improve the flow of information through their nexus among stakeholders (Pfeffer and Salancik 1978). NEDs can foster knowledge sharing, and they can provide their expert advice regarding the strategic plan, which can reduce uncertainty (Mallin, 2019).

Contrary to this, some researchers (proponents of stewardship theory) argue that NEDS are only part-time and therefore they have less information regarding the firm's day-to-day activities, which can hamper their ability to perform their monitoring function (Baysinger and Hookisson, 1990; Agrawal and Knoeber, 1996). Hence, executive directors (insider directors) can perform the monitoring function more effectively (Baysinger and Hoskinsson, 1990; Tricker, 2019).



Thus, this study considered the three perspectives to explore the impact of NEDs on firm financial performance (agency, resource dependence and stewardship theories). Previous researchers (Arosa et al., 2012; Serrasqueiro and Nunes, 2008) have investigated the role of NEDs in predicting firm financial performance by considering their percentage in board membership. Consistent with the previous studies, this study measured NEDs based on their percentage according to total directors in BODs. This information was obtained from the annual reports of the sample firms (Akbar, et al., 2016).

3.3.3.4 Presence of board sub-committees (BSCS)

The existing body of literature recommends that board sub-committees can increase the effectiveness of BODs in more efficient ways (Harrison et al., 1994; Jiraporn et al., 2009; Chen and Jaggi, 2001). Such committees ensure genuine audit and nomination of qualified directors as well as payment of fair compensation to the directors (Chhaochharia and Grinstein, 2009; Jiraporn et al., 2009; Al-Ahdal et al., 2020). Globally, most of the codes pertaining to the CG mechanism recommend establishing board sub-committees (e.g. the Cadbury Report 1999 in the UK and the Blue-Ribbon Committee 1999 in the US).

During the last two decades, the presence of board sub-committees has increased dramatically. As per the contents of the Cadbury Report (1992), the majority of UK firms had a well-functioning remuneration committee while 48.6% firms had a nominating committee. Besides recommendations, empirical studies have documented mixed impacts of board sub-committees on the firm performance. In addition to this small size of the board, sub-committees enable committee members to hold meetings more frequently, which ensures timely decision making (Karamanou and Vafeas, 2005). More specifically, sub-committees perform specialised tasks, which increases firm performance. For instance, audit committees keep tight control over the reporting and internal control systems.



In contrast, nomination committees perform their duties in nominating the qualified members who increase the firm's performance through their studies expertise. However. some have also reported negative consequences of board sub-committees on firm performance. For example, firms have to bear the extra cost in establishing board sub-committees, while remuneration offered to sub-committee members can also be an additional cost for the firm (Vafeas, 1999). According to Conger et al. (1998); Braendle (2019), sub-committees may impose excessive monitoring on the activities of executive directors, which can limit their initiatives in enhancing firm performance. In addition to this, conflict among the members of committees can arise due to diverse expertise and knowledge of board members inducing conflicting situations.

The JCGC (2006) requires that all the firms listed on the ASE should have to establish nomination, remuneration and audit committees. All these committees should have independent non-executive directors as their chairman. This code also requires firms must establish remuneration committees with a minimum of three independent non-executive directors. For nomination committees, this code requires that more than half of the members should be independent NEDs. Keeping in mind the suggestions/recommendations of the JCGC (2006), it can be assumed that board sub-committees can influence the financial performance of Jordanian firms in a positive way.

3.3.4 Corporate governance index (CGI)

To assess the impact of CG on firm performance (accounting or market-based performance), the corporate governance index (CGI) is structured and considered as the predictor. In this study, the CGI has aggregated the provisions of the JCGC (2012). The JCGC (2012) is comprised of three parts. These parts are the introduction, board of directors/management committee and control environment. The ASE had adopted this code as a requirement for all Jordanian listed companies to comply with the provisions of the JCGC (2012). In case firms do not comply with these provisions, then



they are bound to provide justification in this regard. All of the previous studies which have documented the impact of CG on firm performance have followed national or internal CG codes and thus, keeping in mind the previous studies, this study followed the JCGC (2012) and its provisions (Schulze et al., 2003; Shabbir, 2008; El-Faitouri, 2014; Akbar et al., 2016).

The CGI tends to investigate the impact of CG as a whole in relation to the firm's financial performance measured through accounting-based measures. Most of the previous studies have considered only one aspect at one time, such as Gompers et al. (2003) and Cremers and Nair (2005) followed only shareholder rights, while Yermack (1996) considered the board of directors' size in his study. However, this study constructed a GI which provides sufficient room and allows the relationship between CG and performance of the firms to be measured.

Previous literature provides two methods to measure the quality of CG mechanisms (Lang et al., 2004). The first method recommends using subjective analyst disclosure quality rankings. Most of the professional institutes use this method, which deals in ranking the CG codes of nations, e.g. Standard & Poor's (S&P). However, the second method, which is more common, uses researcher-constructed disclosure indices; this method is based on the calculation of disclosure of governance aspects and finding out the quality of the disclosure by using the proxy for it. This type of disclosure is found in firms' annual reports (Chen et al., 2007; Alqatan, Chbib and Hussainey, 2019). Regarding CG disclosure, both methods have their pros and cons.

Above illustrated method of subjective analyst disclosure of quality, rankings are criticised usually. First, this method does not consider disclosure in other sources such as media and interim reports. In contrast, researcher constructed-disclosure indices can consider these types of disclosures (Long and Lundholm, 1993). But some researchers have recommended using the subjective analyst disclosure of quality rankings because this method could be more convincing than the researcher-constructed disclosure indices



method. After all, the latter method depends upon the knowledge and skills of the professional researchers who have expertise in their field (Sanda et al., 2005; Tricker, 2019).

Moreover, the researcher-constructed disclosure indices are subject to errors and bias of personal choices by researchers, leading to false conclusions (Connelly and Limpaphayom, 2004). Second, it is more labour intensive, so it is more convenient and reasonable for researchers to use a small sample size of firms that have fewer observations than larger companies using researcher-constructed disclosure indices (Sanda et al., 2005), which can produce misleading results. Finally, Marston and Shrives (1991) suggested using the current index, which has an essential advantage, which is that it can be compared with prior studies that have used this method in their research.

Despite the criticism of researcher-constructed disclosure indices, this study uses this method due to several factors. First, the subjective analyst disclosure of quality rankings has been used to rank the CG in a specific country (US). Thus, the standards of listing US companies may not be suitable to other nations due to differences in corporate governance regulations and approaches. Second, the subjective analyst disclosure of quality rankings may be out of date or suspended (Sanda et al., 2005). In these circumstances, there is no corporate governance ranking for Jordanian listed companies. Third, analysts' ratings usually focus on the largest companies that are influential in their industry, and they are unlikely to produce adequate variation in terms of corporate governance disclosure (Rosser, 2003). Finally, so far there is no published Jordanian study that examines the relationship between a corporate governance index and corporate performance by using a panel dataset with the generalised method of regression models (Solomon, 2014).

Following prior studies (for example, Gompers et al., 2003; Black et al., 2006; Arcot and Bruno, 2007; Henry, 2008; Shabbir, 2008; Morey et al., 2009; El-Faitouri, 2014; Akbar et al., 2016), this study uses a dummy coding



scheme to evaluate the compliance of Jordanian listed firms with the code. This method of rating includes giving a value of 1 if a company complies with a particular provision of the JCGC and 0 otherwise. This study selects the JCGC (2012) because it is the most recent version.

Table 3-3 shows the structure of the CGI and the provisions of the JCGC (2012), which have been consulted to develop the index. From the provisions of the JCGC (2012), structures of BODs and their sub-committees have been recommended. Additionally, the following table depicts definitions of the index coding measurement of the variables used in this study. Further, elements of the corporate governance index structure in this study are among the most inclusive indices in comparison to already conducted studies through the development of a governance index. For example, Padgett and Shabbir (2005) developed a non-compliance index of 12 provisions based on the recommendations of the UK CG code of 1998. In the same way, eight provisions from the UK CG code were used by Arcot and Bruno (2007) in their study. Hence, the governance index also has some limitations in measuring the provisions of CG codes, and practically difficult provisions are ignored by researchers while developing their index of corporate governance.



Table 3-3 Composition of corporate governance index

Corporate governance variables	Measurement
Board of directors	
Chairman and CEO	A dummy variable equal to 1 if the roles of chairman and CEO are combined, 0 otherwise.
Board structure	A dummy variable equal to 1 if half or more of directors on board of directors are independent non-executive directors, 0 otherwise.
Chairman	A dummy variable equal to 1 if the board chairman is independent non-executive director, 0 otherwise.
Board sub-committees	
Remuneration committee	ee e
Presence	A dummy variable equal to 1 if the company has a remuneration committee, 0 otherwise.
Structure	A dummy variable equal to 1 if the remuneration committee has three independent non-executive directors or more, 0 otherwise.
Chairman of remuneration committee	A dummy variable equal to 1 if the chairman of the remuneration committee is independent, 0 otherwise.
Audit committee	
Presence	A dummy variable equal to 1 if the company has an audit committee, 0 otherwise.
Structure	A dummy variable equal to 1 if the audit committee has three independent non-executive directors or more, 0 otherwise.
Chairman	A dummy variable equal to 1 if the chairman of the remuneration committee is independent, 0 otherwise.
Financial expert	A dummy variable equal to 1 if the audit committee has at least one financial expert, 0 otherwise.
Nomination committee	
Presence	A dummy variable equal to 1 if the company has a nomination committee, 0 otherwise.
Structure	A dummy variable equal to 1 if more than half of the nomination committee are independent non-executive directors or more, 0 otherwise.
Chairman	A dummy variable equal to 1 if the chairman of the nomination committee is independent, 0 otherwise.

Source: Self-generated



3.4 Data analysis

The researcher will use various statistical tools to analyse the data for non-financial Jordanian listed companies. The tools that will be used for analysing the data are descriptive analysis, correlation and multiple regression. Descriptive analysis will use to provide the summary of the entire data that is collected for 95 non-financial Jordanian listed companies from 2012 to 2018. Mean, variance, standard deviation, skewness, range, minimum and maximum descriptive analysis tools will be used to provide the summary of the large data set.

Correlation will also be used to explore how different variables are moving together with each other. The range of correlation between the variables is between -1 to +1, where -1 is highlighting the perfect negative correlation between the two variables and +1 is highlighting the perfect positive correlation between the two variables. Correlation will be used to explore the initial correlation between the variables of the current study.

Multiple regression analysis will be used to explore the impact of independent variables on dependent variable of the study. Independent variables are the variables that are not depending on other variables and have direct effect on dependent variable. Dependent variable is the variable that is being measured and dependent on independent variables and the researchers are looking for the possible effect of the independent variables on dependent variable. General form of the model is as follows:

$$FP_{it} = \beta_o + \Sigma_i^n \beta_i X_{it} + \varepsilon$$
 Equation 1

FPit = measure of firm performance of a firm i at time t

 β_o = intercept of the equation

 β_i = change co-efficient for X_{it} variables

 X_{it} = different independent variables for firm performance of a firm i at time t



i = the number of the firms i.e. $i = 1, 2, 3 \dots N$ (in this study N= 95 firms)

t = the time period i.e. t = 1, 2, 3 ... T (in this study T = 7 years)

For the current study the dependent variable is firm performance and it will be measured through three different ratios including return on assets and return on equity (book value measure) and Tobin's Q (market value measure). Various corporate governance mechanisms (board size, CEO duality, NEDs, audit committee, remuneration committee, nomination committee and corporate governance index) and control variables (sales growth, capital expenditure, leverage, firm size, research and development expenditures and liquidity) are used as independent variables to check their effect on firm performance that is dependent variable.

Researcher will run regression twice. Firstly, it will run to investigate the impact of various corporate governance mechanisms on firm performance in a standalone environment. Secondly, it will run to investigate the impact of corporate governance index on the firm performance. After converting equation 1 into our specified variables, the two equations will be

$$FP_{it} = \beta_0 + \beta_1(SGH_{it}) + \beta_2(FSE_{it}) + \beta_3(CAPX_{it}) + \beta_4(LEE_{it}) + \beta_5(R\&D_{it}) + \beta_6$$

$$(LIY_{it}) + \beta_7(BSE_{it}) + \beta_8(NED_{it}) + \beta_9(CEOD_{it}) + \beta_{10}(ACE_{it}) + \beta_{11}(NCE_{it}) + \beta_{12}$$

$$(RCE_{it}) + \epsilon \dots Equation 2$$

FP_{it} =
$$\beta_0$$
 + β_1 (SGH_{it}) + β_2 (FSE_{it}) + β_3 (CAPX_{it}) + β_4 (LEE_{it}) + β_5 (R&D_{it}) + β_6 (LIY_{it}) + β_7 (CGI_{it}) + ϵ Equation 3

The abbreviation are provided below

FP = Financial performance

SGH = Sales growth

FSE = Firm size

CAPX = Capital expenditure



LEE = Leverage

R&D = Research & development

LIY = Liquidity

BSE = Board size

NED = Non-executive directors

CEOD = CEO duality

ACE = Audit committee

NCE = Nomination committee

RCE = Remuneration committee

CGI = Corporate governance index



3.5 Summary

This chapter elaborates material and methods applied due to availability of pre-tested theories and hypotheses developed. For answering of research questions, multiple regression analysis was selected as major tool of the study. "In order to capture the effects of firm and time specific heterogeneities panel data models can be specified as fixed effects or random effects". Moreover, this chapter seeks to investigate some specification tests which might have tendency to influence corporate governance variables which can result in statistical problems related to predictors in regression model.

Also, this chapter elaborates on the methodology followed to conduct the study. This study used a deductive positivism approach in which already established theories are used to construct study hypotheses. Keeping in mind the complexity of relationships, a multiple regression analysis has been considered to be an important data analysis tool in this study. Fixed and random effects models have been used in this study to assess the heterogeneities in panel data models before running the multiple regressions. Additionally, this chapter has also discussed the underconsideration control variables which might have the potential to influence independent variables of this study in predicting firm performance. Moreover, this chapter has investigated some specification tests which might influence corporate governance variables, which can result in statistical problems related to predictors in the regression model.

This chapter has also shed light on the data and measurement scheme followed in this study regarding sample size, and data selection criteria from different databases and annual reports. The study considers data from three perspectives: one is from the perspective of firm financial performance (measured based on ROA, ROE and Tobin's Q), while the other two aspects consider corporate governance variables (board size, CEO duality, presence of NEDs and sub-committees) as well as control variables (firm sales, growth



size, leverage and R&D expenditure along with liquidity). Data for 95 Jordanian non-financial firms is collected. This study has consulted three sources for data collection: one is the Orbis database, the second is the annual reports of the selected firms, and the third is the ASE website.



Chapter 4 Analysis and discussion

As elaborated above in the methodology chapter, a framework was developed to check the impact of CG on firm performance of non-financial firms operating in the Jordanian context, and results are presented here. This chapter encompasses descriptive statistics and different data reliability, as well as validity tests, correlation and multiple regression results, and provides a discussion on the findings. The second section of the chapter elaborates on the descriptive statistics of the statistical data which has been used for analysis in this study. The third section of this chapter is based on regression results, model specification test results, control variables results, and provides the results and a discussion of CG variables on the CGI. The last section provides a summary of the chapter.

4.1 Results and discussion

The following section shows the results of the study based on which inference has been drawn. The current chapter is comprised of five sections. The first section highlights the results of the descriptive statistics, while the second section deals with the results of the statistical test. In contrast, the third section deals with the results of the control variables. Similarly, the fourth section deals with the results of corporate governance variables, and the fifth and last section discusses the results of the corporate governance index.

Dividing the results into various sections simply has been carried out to facilitate the presentation of the results and to make it easier to understand the results by keeping in mind the effect of each type of variable. Significance of the results has been set at three levels, i.e. .01 indicates highly significant results, 0.05 indicates significant results and .10 indicate marginally significant results. These significant levels are applied to all the tables and results presented in this chapter. Coefficients and other necessary information are reported at suitable places with appropriate tables



derived from the original one. At the same time, overall results are presented at the end of the thesis in Appendix 1.

4.1.1 Descriptive statistics

The following section shows various indicators of descriptive statistics of the statistical data used for analysis purposes in this study. Basic features of descriptive statistics are reported through central tendency and dispersion. Here, the mean, max and min values of each variable are discussed as measures of central tendency while standard deviation (SD) is reported as a measure of data dispersion. Complete results of the descriptive statistics are reported in Appendix 1.

Table 4-1 shows descriptive statistics for the firm performance, which is the dependent variable (DV) in this study and has been measured through ROA, ROE and Tobin's Q. This table indicates that the average range for ROA in this study is a minimum of -1.96 to a maximum of 0.39. At the same time, the average value of ROA is 0.020, with S.D of 0.137. Similarly, ROE ranges from a minimum of -11.21 to a maximum of 0.51 with an average of -0.005 and dispersion of 0.569. Further, TOQ ranges from a minimum of 0.10 to a maximum of 8.36 with an average of 0.823 and dispersion of 0.742.

Table 4-1 Descriptive statistics of ROA, ROE and Tobin's Q

Despt	Rang	Minimu	Maximu	Mea	S.	Varianc	Skewnes
•	е	m	m	n	Dev	е	S
ROA	2.34	-1.96	0.39	0.02	7	0.019	-6.885
ROE	11.72	-11.21	0.51	-0.00	0.56 9	0.323	-16.79
TOQ	8.26	0.1	8.36	0.82	0.74 2	0.551	3.937



Table 4-2 shows the descriptive statistics for the control variables. The table illustrates that the minimum value of SGH in this data was -1.00, while the maximum value was 9.56, with an average of 0.163 and dispersion of 1.173. Further, a minimum of FSE was 12.68 while the maximum value was 21.31 with an average of 17.09 and dispersion of 1.5114. Similarly, the minimum value of CAPX is 0.00 with a maximum value of 0.41 and average of 0.031. Dispersion (standard deviation) for CAPX is 0.046. LEE holds a minimum value of 0.00 with a maximum value of 0.90 and average as 0.28, while the standard deviation for LEE is 0.284. The minimum value of R&D in this data was 0.00 while 0.04 was the maximum value with an average 0.002 and dispersion of 0.006. Finally, minimum, maximum, average and standard deviation values of LIY are 0.01, 1027.00, 9.784 and 80.89, respectively.

Table 4-2 Descriptive statistics for control variables

Dept.	Range	Minimum	Maximum	Mean	S. Dev	Variance	Skewness
SGH	10.56	-1.00	9.56	0.163	1.173	1.377	4.700
FSE	8.63	12.68	21.31	17.09	1.5114	2.284	0.225
CAPX	0.41	0.00	0.41	0.031	0.046	0.002	3.835
LEE	0.9	0.00	0.9	0.28	0.284	0.081	0.696
R&D	0.04	0.00	0.04	0.002	0.006	0.000	4.068
LIY	1026.9	0.01	1027	9.784	80.89	6542.7	11.971

The following table, Table 4-3, shows descriptive statistics for the board governance variables used in this study, i.e. board size, non-executive directors and corporate governance index. First, statistics for the board size illustrate that the mean value of board size in this study/data is 12.47, which can be considered as 13, whereas the minimum value is 7 and the maximum value is 29. These statistics are for the total sample of 95 non-financial firms listed on the ASE. These statistics indicate that, on average, 13 directors work in each company while the minimum number of directors is seven and the maximum number is 29. By exploring the highest number of directors working in a firm, it was observed that only one firm has 29 directors on the



board. This situation indicates that, on average, non-financial firms listed on the ASE in Jordan meet the requirements of the CG Code 2006 and the Company Law (1997). These requirements are also in agreement with the recommendations of Jensen (1993) and Lipton and Lorsch (1992). As per their suggestions, the average number of directors in a firm should be eight to , while 10 should be the maximum number. Such a small sample size can affect the decision-making process on a board. However, literature provides necessary support in favour of the small sample size for firms operating in the context of developing countries. Average board size differs according to the requirements and economic conditions. For instance, the average board size is eight in Egypt and Malaysia (Elsayed, 2007; Haniffa and Hudaib, 2006; Akbar et al., 2016). Similarly, the average board size in the USA is 12.25 (Yermack, 1996). Besides these, a smaller board size on average has been observed in Australia, which is 6.6 (Kiel and Nicholson, 2003).

 Table 4-3
 Descriptive statistics for corporate governance variables

Dept.	Range	Minimum	Maximum	Mean	S. Dev	Variance	Skewness
BSE	22	7	29	12.47	4.36	19.005	1.297
NED	0.85	0.04	0.89	0.173	0.163	0.026	1.463

Descriptive statistics also indicate that, on average, 17% of board members are NEDs, with a range of 10 to 37%. Past studies state that the presence of more NEDs on a board ensures that the board has more independence. This independence might be the outcome of reduced information asymmetry among the agents and principals (Black et al., 2006). According to the findings of Brickley et al. (1997), the performance of the board increases when NED performs on behalf of the agent (owners) with their improved monitoring and advisory function. In the case of Jordan, the presence of NEDs on boards is relatively small when it is compared to other nations; for example, in the USA the mean is 54%, (Yermack, 1996) whilst in Malaysia it is 50% (Haniffia and Hudaib, 2006; El-Faitouri, 2014).



Further, the JCGC of 2007 states that at least 33% of board members should be non-executive directors to ensure independence. However, in our sample, in the case of Jordan, the average composition of non-executive directors is less than the requirements laid down by the JCGC (2012). Nepotism might be the possible reason behind this result. A different dummy variable is also used for the analysis, such as CEO duality, presence of varying board subcommittees, e.g. remuneration committee, audit committee and nomination committee, and CGI.

Descriptive statistics also reveal that, on average (89%), it was observed that most of the firms have the same CEO and board chairman. This situation indicates that board/members are significantly influenced by the CEO/Chairman. In the case of Jordan and family-owned businesses, it has been observed that both roles (CEO and chairman) are played by a single entity, who is mainly the founder of that business/firm. However, under such circumstances in which a single person is playing both roles, it can be assumed that non-financial firms listed on the ASE in Jordan are not following the CG requirements as laid down by the Cadbury Report (1992) and the Jordanian CGC (2006). According to these, the two roles should be separate.

The other dummy variables such as the existence of different board committees, e.g. remuneration committee, audit committee and nomination committee, and CGI. 0 and 1 are recorded for the four dummy variables that are included in the analysis. As 0 and 1 are recorded for the four dummy variables, these variables are not used for the descriptive statistics.

4.1.2 Specification test results

In this study, panel data has been used to find out the effect of corporate governance on financial performance/market performance of the firms. Using panel data increases the need to address the issues of econometric models which are specified in panel data. This study used Breusch and Pagan's



(1980) Lagrang multiplier test. The results of this test are highly significant, which endorses the use of panel regression. According to the assumptions of Breusch and Pagan (1980) Lagrang multiplier tests if it has a null probability (Gujarati, 2003; Saunders et al., 2019), the null hypothesis is rejected and using panel regression is recommended.

Further, panel data models are described/divided into fixed effect and random effect models. These two categories help to assess the impact of firm and time-specific heterogeneities in the sample. However, choosing random results against fixed effect and additional specification is required in the shape of the Hausman test. If the Hausman test is statistically significant (Table 4-4), then using random effect is prohibited, and it is recommended to use the fixed effect model.

Table 4-4 Panel model test

Tests	Results	
Prougeh and Pagan I M Tost	Chi ²	71.42
Breusch and Pagan LM Test	Р	0.000
Hausman Test	Chi ²	8.11
Hausillan Test	P	0.004

The first issue addressed in this section is multicollinearity. Multicollinearity exists when two or more than two predictors are highly correlated with each other (Hair et al., 2009) because their higher correlation can affect the estimation through the regression. Multicollinearity is identified through various methods; the fundamental approach is to look at the values of correlations among predictors. If a correlation of 0.70 or higher exists, it indicates multicollinearity. In this study, no higher correlation (\geq .70) was observed in cases of all predictors (Table 4-8), control variables (table 4-7) or even dependent variables (Table 4-6). This indicates that all the values are

less than the threshold value of 10, and thus it can be inferred that there is no issue of multicollinearity in this model.

Secondly, panel data is usually biased due to the presence of serial correlation, which is necessary to assess and eradicate because it can lead to improper estimation/forecasting. This study followed the approach of the Wooldridge serial correlation test (2002) to determine the serial correlation. Results given below, in Table 4-5, indicate that all the variables are serially correlated. In addition to this, heteroscedasticity must be tested when using panel data analysis, and the Breusch-Pagan test has been used to do this. Results of this test are given below, in Table 4-5, which indicates the presence of a heteroscedasticity issue. Hence, this issue can be resolved by using a cluster-robust standard error estimator (Rogers, 1993; Bell, Bryman and Harley, 2018).

Table 4-5 Specification test results

Tests	Dependent Variable				
16313	ROA	ROE	TOQ		
Wald (Chi-square)	6932.32	3445.7	5346.75		
P	0.000	0.000	0.000		
Breusch-Pagan/Cook-Weisberg test	4408.78	532.34	3748.03		
Heteroscedasticity P-value	0.000	0.000	0.000		
Wooldrige-test for autocorrelation	5.114	16.703	7.1034		
Р	0.030	0.000	0.000		

4.1.3 Correlation

Correlation is a statistical tool that is used to explore how different variables are moving together. The result of correlation always falls between -1 and +1, which shows both negative as well as positive extremes. Values of correlation \pm .70 indicate strong correlation while values higher than \pm .70 and



less than ± 1 indicate a near-perfect correlation. Similarly, values near 0 indicate weak correlation while the 0 itself indicates no relationship among the constructs. Values near to \pm . 10 are considered weak correlations while those near to \pm .30 are regarded as small correlations, and values of correlations near to \pm .50 are regarded as a moderate relationship (Pallant, 2020). The complete correlation table is provided in the appendix 1, whilst Table 4-6 highlights the correlation among different dependent variables (ROA, ROE and TOQ) that are used for the current analysis.

Table 4-6 Correlation for dependent variables

Description	ROA	ROE	TOQ
ROA	-		_
ROE	0.355	-	
TOQ	-0.106	-0.431	-

The above table illustrates that the value of the correlation between ROA and ROE is .355. First, the positive sign shows a positive relationship between these two variables, implying that both will move in the same direction if one changes accordingly. The second instance about this relationship is the presence of small correlation between the two. However, this value of correlation is slightly above weak correlation and moving towards a moderate level of relationship (r=.50). In the case of the relationship between ROA and ROE, a negative correlation has been observed, indicating a negative correlation. In this case, increase in one variable will decrease the other variable. However, the level of correlation is small, i.e. correlation -.106. Similarly, in the case of the relationship of TOQ and ROE, the association is also negative, indicating that, if ROE is moving in an upward direction, the TOQ will move in a downward direction. However, the level of correlation is near to a moderate level, i.e. near to 0.50.



Table 4-7 Correlation for control variables

Description	SGH	FSE	CAPX	LEE	R&D	LIY
SGH	-					
FSE	-0.097	-				
CAPX	-0.035	0.090	-			
LEE	0.049	-0.317	-0.046	-		
R&D	-0.023	0.026	0.021	-0.004	-	
LIY	0.012	-0.200	-0.049	0.152	-0.027	-

The above table illustrates the nature and direction of relationships among the control variables. First, the correlation between the SGH and FSE has been found to be negative at a small level, indicating that an increase in FSE will decrease the SGH at a modest level. Similarly, the relationship of CAPX and SGH has also been found to be negative, which indicates that an increase in CAPX will decrease the SGH. The value of this correlation is very weak, i.e. -.035. Similarly, the correlation of LEE and SGH is also weak but positive, indicating that both LEE and SGH will move in the same direction if one variable is changed. R&D and SGH also have negative but weak correlation showing that, if R&D moves in an upward trend, the SGH will move in a downward direction; however, the level of correlation is very weak. LIY and SGH also have very weak but positive correlation, indicating that both will move in the same direction but with a very weak relationship. In the case of CAPX and FSE, the correlation is also small but positive, implying that both variables will move in the same direction if one of them is changed.

However, LEE and FSE have been found to have a negative relationship which is at a small level, i.e. -.317, indicating that an increase in LEE will decrease the FSE. R&D and FSE have positive correlation but it is very weak. Similarly, the relationship between LIY and FSE is at a small level with a negative sign, indicating that both will move in the opposite direction if one

of them is changed. Further relationship between LEE and CAPX has been found negative with little correlation. A very weak but positive correlation is observed in the case of R&D and CAPX. In the case of R&D and LEE, the correlation is minute, -.004, indicating almost no relationship among both R&D and LEE. In case of LIY and LEE have been found in positive correlation but at a small level. Finally, the relationship between LIY and R&D is negative and weak.

 Table 4-8
 Correlation for corporate governance variables

Description	BSE	NED	CGI
BSE	-		
NED	-0.19	-	
CGI	-0.027	0.208	-

A weak negative relationship exists between BSE and NED and CGI, while a weak positive relationship exists between NED and CGI. The above correlation analysis also highlights that a weak positive or negative relationship exists between all the dependent, controlled and CG variables. Weak correlation among all the variables highlights that no multicollinearity issue exists among the variables. However, the value of the correlation between CGI and NED is positive at a small correlation. In contrast, the correlation between BSE and NED indicates a negative relationship between them, meaning that if the board size increases then the number of NEDs will decrease at a weak level.

4.1.4 Results of multiple regression analysis

Regression analysis helps to forecast a dependent variable on a given value of the independent variable. Alternatively, it can be stated that it is used to explore the impact of an independent variable on the dependent variable. For



the current analysis, firm performance is the dependent variable, which is measured through three different accounting ratios: ROA, ROE and TOQ. ROA and ROE are the accounting measures, while TOQ is the market value measure. For the current study, independent variables are divided into different control variables (SGH, FSE, CAPX, LEE, R&D and LIY), corporate governance variables (BSE, NED) and dummy variables (CEOD, RCE, ACE, NCE and CGI).

Regression analysis in this study has been divided into two sections. Firstly, the researcher will conduct the regression analysis by using all the corporate governance variables in a standalone manner. The researcher will measure the firm performance with the help of three accounting ratios (ROA, ROE and TOQ) so the regression equation will be run three times to measure the impact of different corporate governance variables in isolation on firm performance. After conducting the regression analysis in a standalone manner, the researcher will check the compliance of all the selected firms against the corporate governance index that is based on 13 different corporate governance requirements of the JCGC. The researcher will run the regression equation three separate times using the three different accounting ratios that are used to measure the firm performance. In this way, the second part of the regression analysis will assist the researcher in measuring the impact of the corporate governance index on the firm performance of listed non-financial Jordanian firms. The complete regression analysis tables are provided in Appendix 1. For ease of discussion, the researcher will break down the regression tables into small tables that are provided in the following section.

Before presenting the results of each independent variable, the researcher will provide the results of F-statistics and R-square because these two measures provide information about the entire model that is used for the analysis. F-statistics are used to check the significance of the entire model, while R-square provides information on how much variation in the dependent variable is explained by independent variables included in the model. Table 4-9 provides the results for F-statistics and R-square.



Table 4-9 F-statistics and R-square

Description	ROA	ROE	TOQ
F-statistics	9.25	4.68	6.96
Significance	0.00	0.00	0.00
R	0.69	0.51	0.58
R square	0.48	0.26	0.37

Table 4-9 highlights that the F-statistics are highly significant for all three models used for the analysis. The R square is highest for ROA (69%) while it is lowest for ROE (51%), which means that the highest variation for firm performance is explained by predictors of governance when firm performance is measured using the ROA. The results indicate that more independent variables should be included in the model to obtain more information about the movement in firm performance.

4.1.4.1 Results of control variables

Results of control variables have been found to be different across the performance variables which have been measured in terms of financial and market performance (ROA, ROE and TOQ).

4.1.4.1.1 Sales growth (SGH)

According to the perception of Durnev and Kim (2005), firms which have greater opportunities are more likely to grow faster as compared to other firms. Theoretically, market valuation firms can provide a good picture of growth opportunities for the firms (Klapper and Love, 2004). In addition to this, the growing firms need significant financial assistance, which drives them to follow a better governance practice to maintain the investors' interest (Beiner et at., 2006); this also helps them to reduce costs. Past researchers have documented a positive relationship between firm performance and



growth opportunities measured in terms of sales growth (SGH) (Gompers et at., 2003; Drobetz et at., 2004). In line with the previous researchers, this study followed sales growth on a year-to-year basis as an indicator of growth opportunities (Cui et at., 2008; Henry, 2008; Cifci et al., 2019).

Table 4-10 Impact of sales growth on firm performance

Description	ROA	ROE	TOQ
Sales growth (SGH)	0.005	0.023	-0.038
Significance	0.032	0.028	0.017

The above table shows that sales growth has a positive relationship with the firm performance for the three-measurement basis that is used for firm performance, ROA, ROE and TOQ. Here, it can be noted that coefficients of sales growth (SGH) as a measure of growth opportunities are smaller for all the three indicators of firm performance. However, these results are statistically significant at 5%. In the case of ROA and ROE, the impact of sales growth is positive, while in the case of TOQ (market performance), the effect is negative. These results are in line with previous empirical results (Drobetz et at., 2004; Cui et at., 2008; Henry, 2008; Akbar et al., 2016).

4.1.4.1.2 Firm size (FSE)

ك للاستشارات

Firm size is measured by considering the figure of total assets, as shown earlier in the methodology chapter. Keeping in mind the recommendations of previous researchers, the total assets of firms in this study were transformed into logs. This factor helped to minimise the skewness and kurtosis as well as helping to overcome the effect of outliers' data points. Statistics reported in Table 4-11 below illustrate a positive impact of firm size on firm performance. Further, these effects are statistically significant in terms of its effect on three indicators of performance (financial and market performance), ROA, ROE and TOQ. The positive relationship between the variables

indicates that firms with a large assets base could achieve economies of scale.

Further, firms with larger assets can use their economies of scale to introduce an efficient production process, which can increase their performance. Hence, firms with larger assets have the potential to secure/attract finance. Firms with larger assets can also generate funds through both sources, either internal or external (Short and Keasey, 1999). According to Meek et al. (1995), larger firms tend to be more complex and more diversified when considering the market development and business risk because larger firms have more access to information as compared to small firms. Additionally, based on positive effects reported in this study, it can be assumed that firms with larger assets tend to involve themselves in broader activities and value creation process. Further, larger firms have a varied product range through their diversified resources and influence on the market. It can be implied that larger firms are in a position to borrow on better conditions due to their large asset base, which can be used as collateral (Alqatan, Chbib and Hussainey, 2019).

Table 4-11 Impact of firm size on firm performance

Description	ROA	ROE	TOQ
Firm size (FSE)	0.029	0.102	0.133
Significance	0.00	0.00	0.00

4.1.4.1.3 Capital expenditure (CAPX)

According to Jermias (2007) and Brown et al. (2009), firms which deal in technology-related products tend to invest in innovative initiatives. This factor helps them to gain a competitive advantage by introducing unique products and new services. Later, these unique products and service help firms to achieve a quasi-monopoly, which results in premium prices as well as

ensures improved profitability in the long run (Jermias, 2007; Braendle, 2019).

However, investing in unique and innovate methods is not an easy task. There is a need to protect the intangible assets of firms which are dealing in the technology sector because it is quite easy to steal intangible assets as compared to fixed assets. Hence, corporate governance can protect the firms from misuse of their intangible assets (Durnev and Kim, 2005; Black et at., 2006; Brown et at., 2009; Al-Ahdal et al., 2020).

Table 4-12 Impact of capital expenditure on firm performance

Description		ROA	ROE	TOQ
Capital (CAPX)	expenditure	-0.177	-0.112	-1.104
Significance		0.17	0.84	0.12

The above table illustrates that the impact of CAPX is negative on all the indicators of performance, both financial and market performance, for the three-measurement basis used for firm performance, ROA, ROE and TOQ. Here, the impact of CAPX has been found to be more influential on the TOQ (market performance) of the firm as compared to the other two indicators. However, these results are not statistically significant even at 5%, so no relationship can be ascertained between CAPX and firm performance.

4.1.4.1.4 Leverage (LEE)

This study used the percentage of long-term debt to total assets as a measure of leverage, keeping in mind the alignment of previous studies. Statistics given in Table 4-13 indicate that leverage has a negative impact on TOQ with a significance level of .01. This factor suggests that higher levels of debt will decrease the performance of the firm. At a higher debt ratio there will be lower ROA, ROE and Tobin's Q. However, these results are only



significant for Tobin's Q (market performance) at a 1% confidence level, whereas for ROA and ROE they are not significant. It can be argued that an increase in debt ratio can increase the firm's operating costs, which might undermine its ability to meet financial obligations regarding higher interest rates (Dechow et al., 1996).

In the same context, a higher level of debt can limit the firm in generating new funds through loans, which can restrict the firm's potential to invest in valuable investment opportunities. Simply, it can be concluded that a higher level of debt further hampers the amount of dividend. The reason is that a firm with a higher debt ratio tends to pay lower dividends to avoid an external source of financing, which is a burden on the firm. Moreover, the presence of a higher level of debt can lead firms towards financial distress; also, it can cause financial constraints and would become quite difficult for the firms to borrow, as even banks become more cautious in such circumstances. On the other hand, confidence of current as well as potential shareholders is undermined (Chen and Jaggi, 2001; Stulz, 1988; Detthamrong, Chancharate and Vithessonthi, 2017).

Table 4-13 Impact of leverage on firm performance

Description	ROA	ROE	TOQ
Leverage (LEE)	-0.025	-0.093	-0.33
Significance	0.26	0.33	0.01

4.1.4.1.5 Research and development (R&D)

Firms usually tend to invest in new products to gain a competitive advantage through the introduction of new products or services. Once these new products or services become available, companies can demand higher prices and generate profitability in the long term (Barney, 1991, Calantone et al., 2002). Additionally, a new invention can also be considered as a barrier



for rivals, preventing them from entering the market and attracting new clients (Golder and Tellis, 1993). Although research and development (R&D) expenditure helps to ensure success in the economic environment as well as in the market, previous researchers have reported mixed results while documenting the impact of R&D expenditure on firm performance. For instance, Lev and Sougiannis (1996) reported that for each dollar increase R&D expenditure generates almost 1.7 to 2.6 USD in future earnings.

Similarly, Bublitz and Ettredge (1989) documented the positive effect of R&D expenditures on future cash flows. Accordingly, Chan et al. (2010) stated that every announcement related to the increase in R&D increases the share prices of firms positively. However, other empirical studies have not found a significant relationship between R&D expenditure and corporate performance (Johnson, 1967; Hall and Bagchi-Sen, 2002; El-Faitouri, 2014). Following prior corporate governance studies (see, for example, Vafeas and Theodorou, 1998; Dahya and McConnell, 2007; Akbar et al., 2016), this study uses R&D as a control variable, measured through dividing the figure of total R&D expenditure by the book value of the firm's total assets.

Table 4-14 Impact of research & development on firm performance

Description	ROA	ROE	TOQ
Research & Development (R&D)	4.111	7.809	10.933
Significance	0.00	0.03	0.04

The above shows the results of R&D expenditure on the financial and market performance of the firm. These results are significant at 1% for ROA and 5% for ROE and TOQ. The table highlights that R&D has a positive effect on the three-measurement basis of firm performance in both directions, financial and market performance, such as ROA, ROE and TOQ. In the case of ROA, the coefficient is 4.111 at 1% significance level, for ROE it is 7.809 at 5%



significance level, while for TOQ it is 10.933 significant at 5% level of confidence. However, a positive impact is found in all three indicators of firm performance, but the more significant effect is observed in the case of ROE as compared to ROA. Similarly, in the case of TOQ, the impact is more significant as compared to the other two indicators of the financial performance of the firm. The results are also statistically significant at 5%, so it can be stated that a positive relationship exists between CAPX and firm performance for Jordanian non-financial firms listed on the ASE.

4.1.4.1.6 Liquidity (LIY)

In this study, the impact of liquidity on the financial and market performance of the firm has been found to be insignificant. However, in the case of liquidity (LIY), ROE has an insignificant p-value. The impact of LIY on TOQ (market performance) of the firm was found to be highly insignificant. Similarly, results for ROA were also insignificant. It can be argued here that firms with a high liquidity ratio have the potential to absorb external shocks very comfortably. Firms with a higher level of liquidity can handle economic crises very easily, and such firms can reduce their financial distress. Additionally, firms having a higher level of liquidity can enjoy the benefits of investing through greater opportunities as compared to those firms which have a lower level of liquidity.

Table 4-15 Impact of liquidity on firm performance

Description	ROA	ROE	TOQ
Liquidity (LIY)	7.012	0.000	2.476
Significance	0.35	0.27	0.95

The above table illustrates that liquidity has a positive effect on the firm performance, measured either through the market or financial performance (ROA, ROE and TOQ). Here, to measure firm performance, a three-



measurement basis is used. In the case of the impact of liquidity on ROE, a minute coefficient has been observed which is insignificant too. In the case of TOQ, the coefficient was more significant than for ROE; however, a large coefficient was observed in the case of ROA as compared to the other indicators of firm performance measured in this study. The results are not statistically significant even at 5%, so no relationship can be ascertained between liquidity and firm performance for Jordanian non-financial firms listed on the ASE.

4.1.4.2 Results of corporate governance variables

4.1.4.2.1 Board size (BSE)

In the case of board size (BSE), a negative impact was observed in all the three indicators of firm performance (Table 4-16). In the case of effects of board size (BSE) on ROA, the coefficient was very weak with a negative sign, indicating a negative impact on ROA. Similarly, in the case of the impact of board size on ROE, the coefficient was minute with a negative sign, showing a negative effect on ROE. Similar results were observed in the case of the effect of board size on TOQ: a negative impact was observed. Impact for all three measures of performance, ROA, ROW and TOQ, was significant at 1% level of confidence, implying that with the increase in board size the performance of the firm decreases.

These findings are in line with the recommendations of many previous researchers (e.g. Lipton and Lorsch 1992; Jensen 1993; Yermack 1996; Jacoby et al., 2019) that larger board size can result in poor coordination and communications. With the increase in board size, the issue of coordination and communication can be more complex, which leads to agency problems through management and control issues (Eisenberg et al., 1998). The presence of a large number of directors on a board promotes various opinions, which results in inefficient decision making due to the number of views. Simply, it can be stated that, due to differing views, it becomes quite difficult for the board to exert their control over the management. On the



other hand, a small board develops more cohesiveness, and the members can easily form a shared opinion, which boosts efficient decision making and more tight and effective control and monitoring of managers/agents.

However, some past researchers have provided opinions in favour of larger boards (e.g. Miller, 2003; Dalton et al., 1999; Lehn et al. 2009). According to these researchers, larger boards perform very well in managing firms' affairs. In contrast, in small boards, the CEO enjoys a position of power and usually overrides decisions to protect his/her interests, which may result in agency problems. This increased agency problem might result in lower performance of the firm (Miller, 2003).

Further, as per the statement of resource dependency theory, larger boards have more significant networks, which enables them to strengthen their linkages with external resources and thus attract expertise. This fact helps to generate funds as well as helps to seize the benefits from the capital. In addition to this, larger boards also tend to facilitate the directors to exchange highly qualified counsel, which boosts the positive linkage with externals. Moreover, larger boards facilitate decision making, and this improves the quality of idea sharing and contributions. This factor increased the board performance in every aspect (Lehn et al., 2009; Shahid and Abbas, 2019). Hence, it can be assumed that larger boards can minimise the conflicts among the board members, principals and shareholders, which increases the shareholders' wealth and increases the firm's performance as well.

Table 4-16 Impact of board size on firm performance

Description	ROA	ROE	TOQ
Board Size (BSE)	-0.004	-0.020	-0.024
Significance	0.01	0.00	0.00



4.1.4.2.2 Non-executive directors (NEDs)

As depicted by Table 4-17, NEDs have a significant impact on all the indicators of firm performance, both financial and market performance, measured in terms of ROA, ROE and TOQ. The effect was observed to be significant with negative coefficients. The impact of NEDs on ROA and TOQ was equal in terms of magnitude while it was more significant in size in the case of ROE. The negative sign shows that NEDs tend to decrease the firm's performance, whether measured in terms of financial performance or market performance. These findings are not in alignment with the monitoring hypothesis of agency theory. This hypothesis holds that independence of the board increases with the presence of a larger number of NEDs on the board because NEDs increase value addition in the board through independent judgements (Cadbury Report, 1992; Chhaochharia and Grinstein, 2009; Tricker, 2019).

Further, NEDs, being an external source on the board, can boost the experience and enhance effective monitoring of the firm's affairs. As part of the external environment, NEDS also play their role in developing a positive reputation for the firm through monitoring services (Baranchuk and Dybvig, 2009). Besides these arguments of this study, the results are also consistent with the recommendations of past researchers (e.g. Weir and Laing, 2003; Yermack, 1996). These researchers have reported that those companies with a higher number of NEDs are most likely to experience lower performance. The reason for this might be the presence of NEDs as part-time workers who are not familiar with the environment of the firm. Thus, they face complications in monitoring the issues and affairs of the firm, which might result in the firm's poor performance. Possible reasons behind these findings are given as:

 NEDs are mainly part-time workers who work for firms on a part-time basis and are even considered as ceremonial functionaries. Due to their part-time nature, they do not have sufficient information and knowledge regarding the daily activities of the firm; these deficiencies



undermine the potential of the NEDs to manage the affairs of the firm properly, and they are unable to perform their duties effectively. The incentives on offer might be an important consideration for the NEDs, and, in the case of fewer incentives, their motivation to serve the firm may be hampered. They can overlook their responsibilities, which may result in lower/poor performance of the firm.

- Due to their part-time nature, they have commitments with other stakeholders which might affect their devotion to the firm, hence lowering their monitoring. For instance, a NED in one firm can be an executive director in another firm. Thus, role conflict can undermine his/her motivation and incentive to perform his/her task correctly and effectively in both organisations.
- NEDs can be unfamiliar with the affairs of the firm and its day-to-day activities. This fact makes them unable to assess the complications and difficulties in arranging the firm's affairs, thus lowering the firm's performance. Additionally, the lack of technical knowledge and expertise among NEDs can result in the firm's poor performance.
- Furthermore, it is possible that the CEO and NEDs can indulge their relationships or may develop their nexus to promote their own interests. In such circumstances, NEDs and the CEO will not work in the best interests of shareholders. Hence, it can be postulated that NEDs in a firm cannot perform an active and productive role in monitoring the activities of the firm or execute their duties properly.

Despite inconsistencies in the findings, the results mentioned above are very interesting in relation to NEDs. Simply, the negative association among the NEDs and financial and market performance of the firm cannot overturn the recommendations of corporate governance codes which are in practice across the globe, nor can it be assumed that the recommendations of these CG codes are wrong in the case of NEDs. Literature provides sufficient



support that, in most developing nations, firms are controlled and managed by the family business; this is even observed in the case of MENA countries (ROSC, 2004). These operating families hold significant shares in the firms, due to which they dominate the board. This dominance on the board can result in lower performance because such members do not have the necessary and sufficient knowledge due to the technicities involved in such firms. However, in spite of these difficulties, NEDs in every firm do not need to have the proper independence to perform their role effectively and efficiently. In some cases, NEDs can even compromise on many issues due to their close social circles or relationships with the managers, CEOs or even with other board members, and this situation might hamper their ability to interfere with management decisions independently (Mallin, 2019).

Table 4-17 Impact of non-executive directors on firm performance

Description		ROA	ROE	TOQ
Non-executive (NEDs)	directors	-0.09	-0.196	-0.097
Significance		0.02	0.02	0.02

4.1.4.2.3 CEO duality (CEOD)

With regard to the impact of CEO duality (CEOD) on the performance of the firm from both financial and market perspectives, a positive effect has been observed (Table 4-18). These findings are highly significant in terms of ROA, ROE and TOQ. Impact of CEO duality (CEOD) on ROE was smaller as compared to the two other measures of firm performance. At the same time, the effect of CEOD on TOQ was more significant as compared to the other two indicators of financial performance. This positive impact on the performance of the firm consists of the stewardship theory, and these results

endorse this perspective of stewardship theory. The performance of the firm increases when the two roles of CEO and chairman are held by a single person. The above quoted positive relationship between CEO duality and firm performance shows that the CEO's knowledge regarding the firm can increase the investment opportunities as well as enhance the firm's strategic direction (Tricker, 2019).

Duality also allows optimum decision making, thus reducing the wastage of scarce resources in a firm. Hence, the CEO/Chairman holding two roles at the same time can extend his/her services by rendering his/her valuable knowledge to the other board directors. This fact can facilitate the advisory role in more effective and efficient ways. In addition to this, duality can help to minimise the conflicts among the board members. Further, when the CEO and Chairman are two separate roles, it can create tension and confusion regarding implementation of decisions as well as formulation of policies and procedures (Solomon, 2014).

Contrary to this, when the two roles are combined in a single role, it results in more consistent and effective management through cohesive decision making. However, if the role is separated into two categories, it can promote rivalry among the two key roles. On the other hand, combining the roles can help to avoid the potential conflict between two personalities. Simply, duality provides a single command chain and eliminates the tension of reporting by lower staff to 'high ups'. It also promotes unified leaderships which can facilitate enhanced understanding of firms' day-to-day affairs and decisions. Additionally, considering the case of the emerging markets of developing countries, combining the roles of CEO and chairman is common in small businesses; it offers cost-effectiveness and boosts organisational performance due to unified leadership and monitoring. In the case of family-owned businesses, it is common to combine the roles of CEO and chairman in a single role through duality when the resources are scarce, and firms have to manage within a competitive environment.



These findings are in alignment with a number of previous studies (Elsayed, 2007; Haniffa and Cooke, 2002; Haniffa and Hudaib, 2006; Weir et al., 2002; Tricker, 2019). These researchers have reported a positive association between CEO duality (CEOD) and firm performance. However, firms can face emerging challenges as well as being able to grab potential opportunities when they implement new strategies. In a case where both roles are confined in a single entity, the CEO/Chairman is more likely to face these challenges successfully due to their vast expertise and knowledge. The broad knowledge and expertise of the CEO/Chairman enable him/her to understand the issues more deeply and respond accordingly, as compared to NEDs (Weir et al., 2002; Mallin, 2019).

Furthermore, CEO duality provides the freedom to focus on long-run objectives. Freedom by one person holding both roles eradicates the tension at board level, and thus lower interference improves the firm's performance. Therefore, in the case of rapid decision making due to a dynamic business environment, a single entity holding both roles as chairman and CEO can respond rapidly. This can result in improved decision making (Haniffa and Hudaib, 2006). From the perspective of cost, duality helps to eradicate the tension regarding extra remunerations/compensations (Vafeas and Theodorou, 1998). In addition to these benefits, it is quite easy to hold a single person accountable. In the case of any poor performance, the CEO/Chairman can be blamed very easily and comfortably (Abor, 2007; Bozec, 2005; Shu and Chiang, 2020).

While these results are consistent with previous studies, somehow inconsistencies are also attached to them when the perspective of agency theory is considered. According to this perspective, the roles of CEO and chairman should be separated. According to this theory, duality in a firm indicates a problematic situation. Agency theory argues that CEO duality represents a problem because the same person will manage the performance as thoroughly as well as it will also evaluate the performance of the firm.



According to agency theory, CEO duality will result in less efficient and effective supervision and management of the firm's affairs. The reason is that, under such circumstances, management will become opportunistic, which will increase the agency problems. Hence, probably, under CEO duality, there is a chance that the CEO will not control the board properly, in order to protect his/her interests. This fact will also result in inadequate monitoring and come at the expense of shareholders. Alternatively, it can be argued that, under CEO duality, the role of CEO is rooted in the board, because it is the prime responsibility of the chairman to set the board agenda to manage the affairs of the firm through increased information (Braendle, 2019).

Due to these reasons, it can be concluded that duality leads to entrenchment of the executives as well as the CEO and, under such circumstances, the CEO as well as the other board members will become inefficient in performing their monitoring role. Hence, it can be stated that CEO duality hampers the functions of the board as well as reducing the performance of the board members. This dysfunction results in misalignment among the interests of shareholders, managers, board members and CEO. This situation leads to increased agency problems, thus reducing the maximisation of shareholder wealth.

The findings of this study are inconsistent with the Cadbury Report (1992) and OECD. Also, these findings are contradicted by the UK Combined Code and JCG Code (2006). All of these governing mechanisms state that the roles of CEO and Chairman should be separate to avoid agency conflict and to increase the performance of the firm. Furthermore, these findings are in contradiction to the findings of some previous studies (e.g. Chahine and Tohme, 2009; Dahya et al., 1996; Rechner and Dalton, 1991). These findings have found a negative association between duality and firm performance, and this study has observed a positive association between the duality and indicators of financial and market performance of the firm. The observed previous negative relationship by researchers between performance and duality might be due to the entrenchment and reduced

board monitoring function, which provides the freedom to the CEO to follow his/her interests at the expense of the shareholders' interests (Tricker, 2019). Keeping in mind the context of Jordan, where companies usually tend to be smaller, the presence of CEO duality can provide a meaningful advantage from the perspective of cost and management. Furthermore, in Jordanian firms, duality can provide improved supervision and cohesiveness among the board members. It can also provide unified leadership, which may result in improved firm performance. As Jordanian businesses are often family owned, the founder is more likely to be the CEO due to his/her vast experience and knowledge about the firm. In addition to this, most firms in developing nations are controlled by families where family members hold a majority share. The same case is observed in MENA countries which have the same structure of businesses as Jordan (ROSC, 2004). Hence, due to the generic nature of duality, it is not present in Jordanian firms (Mallin, 2019).

However, in mature markets, it can be observed that duality does not decrease the firm performance compared to in different cultures, and duality might have positive results too, especially in the West. Further, most Jordanian firms are operating in a simpler organisational environment as compared to firms in developed nations. Hence, CEO duality (OECD) might provide benefits in terms of cost, improved decision making, strategic alignment and speedy decision making. Further, it can improve communications among the various stakeholders of the firm, such as board members and managers.

Separating the roles of CEO and chairman in small firms might produce complications in terms of internal power conflicts, which could further extend among management. In a case when the firms are simpler and operate in smaller product lines, it can become a major cost for the firm. Furthermore, small firms lack resources, and such firms need to be able to make a speedy and rapid response to various market fluctuations. This speedy and quicker response helps small firms to minimise and control costs to become more



successful (Akbar et al., 2016). If a small firm has duality, it will be more difficult for it to respond rapidly.

Moreover, the financial cost associated with separating the roles of CEO and chairman might take away from the benefits in a small firm. A similar case can be observed for administrative cost. However, in large firms where operations are complex, separating CEO and chairman can provide fruitful benefits because such benefits will exceed the cost associated with the separation of the two roles (Qurashi, 2018).

Table 4-18 Impact of CEO duality on firm performance

Description	ROA	ROE	TOQ
CEO duality (CEOD)	0.381	0.214	0.598
Significance	0.00	0.01	0.00

4.1.4.2.4 Establishment of board committees

Board sub-committees are an important part of the CG mechanism. These sub-committees help the board and board directors to increase the firm's performance through efficiency and effectiveness. Such board committees help to reduce the impact of corporate managerial teams. Further, these committees also help to minimise the effect of large shareholders on the board. Past researchers have provided evidence that such sub-committees tend to enhance the quality of the CG mechanism of firms, which further increases their performance (e.g. Bizjak and Anderson, 2000; Ruigrok et al., 2006; Laplante and Tong, 2007; Anderson et al., 2020). This study tested the impact of various sub-committees, audit committee, remuneration committee, and nomination committee (Table 4-19). The results provided a clear picture regarding the impact of such committees on the firms' financial and market performance.



Here, the impact of RCE on ROA, ROE and TOQ was found to be positive and significant at 1% level of confidence. However, the effect of RCE on ROA was found to be more substantial as compared to the other two indicators of performance (ROE and TOQ). At the same time, the impact of RCE on ROE was observed to be smaller as compared to TOQ. Similarly, the effect of ACE on ROA, ROE and TOQ was also found to be positive and significant at 1% level. The impact of ACE on ROE was found to be more influential as compared to the other two indicators of firm performance (ROA and TOQ). At the same time, the impact of NCE was also found to be positive and significant on all three indicators of firm performance (financial and market performance).

Previous researchers have reported that the audit committee tends to ensure the quality of the internal audit, which increases the transparency within the firm. Similarly, the compensation committee ensures remuneration efficiency, which further boosts corporate transparency. In an attempt to align with the market reforms, most Jordanian firms have started to appoint audit, remuneration and nomination committees to improve transparency. This fact shows that firms in Jordan have considered increasing their transparency. Thus, it can be assumed that, in such circumstances, audit and remuneration committees improve the firms' performance (Alqatan, Chbib and Hussainey, 2019).

In comparison to the state-owned companies, privately-owned firms tend to follow the CG mechanism more deliberately and effectively, and these companies tend to strengthen their CG mechanism. Therefore, in private companies, the presence of audit and remuneration committees will bring increased performance (Kang and Stulz, 2008). Hence, in state-owned companies, the board sub-committees do not show any significant improvement in the performance of the firm. In large-size firms, shareholders and executives are powerful enough to manage the affairs of the firm. Usually, they control the management of the company due to their dominant positions (Wei and Geng, 2008). In such scenarios, the nomination



committee cannot perform very well so it can be assumed that in such firms the nomination committee cannot perform very well to increase the firms' financial performance (Tricker, 2019).

Table 4-19 Impact of board committees on firm performance

Description	ROA	ROE	TOQ
Remuneration committee (RCE)	0.683	0.452	0.625
Significance	0.017	0.021	0.018
Audit committee (ACE)	0.149	0.572	0.348
Significance	0.00	0.00	0.00
Nomination committee (NCE)	0.524	0.327	0.826
Significance	0.016	0.024	0.031

4.1.4.5 Results for multiple regression for corporate governance index

This section of the analysis chapter will present the results of multiple regression that are obtained by incorporating the corporate governance index (CGI) collectively instead of using different corporate governance variables in isolation. Table 4-20 presents the results of multiple regression generated by measuring firm performance with ROA.



Table 4-20 Multiple regression results by measuring firm performance by ROA

Description	Unstandardised B	Significant
Constant	-0.35	0.00
Sales Growth	0.09	0.03
Firm Size	0.02	0.00
Capital Expenditures	-0.20	0.13
Leverage	-0.03	0.19
Research & Development	3.79	0.00
Liquidity	6.58	0.38
Corporate Governance Index	0.01	0.01
R 0.65	R Square	0.423
F statistics 9.093	Significant	0.00

Table 4-20 highlights that the F-statistics are highly significant when firm performance is measured through ROA. The R square is approximately 42%, which means that 42% of the variation for firm performance is explained by the independent variables when ROA measures firm performance. The results indicate that more independent variables should be included in the model to obtain more information about the movement in firm performance.

The results of different control variables are in line with the results of previous regression equations where different corporate governance variables were included in the model separately to explore the impact of CG on firm performance (where firm performance is measured through ROA). The positive and significant relationship between firm performance and sales



growth, firm size, research and development expenditures, and corporate governance index has been witnessed. In contrast, insignificant results have been seen between firm performance and capital expenditure, leverage and liquidity.

Table 4-21 Multiple regression results by measuring firm performance by ROE

Description	Unstandardised B	Significant
Constant	-1.20	0.00
Sales Growth	0.03	0.04
Firm Size	0.08	0.00
Capital Expenditures	-0.01	0.98
Leverage	-0.11	0.25
Research & Development	6.15	0.04
Liquidity	0.07	0.31
Corporate Governance Index	0.03	0.04
R 0.33	R Square	0.109
F statistics 3.999	Significant	0.00

Table 4-21 highlights that the F-statistics are highly significant when firm performance is measured through ROE. The R square is approximately 11%, which means that 11% of the variation for firm performance is explained by the independent variables when ROE measures firm performance. The results indicate that more independent variables can be added to the model to obtain more information about the movement in firm performance.

The results of different control variables are in line with the results of previous regression equations where different corporate governance variables were included in the model separately to explore the impact of CG on performance (where firm performance is measured through ROE). The positive and significant relationship between firm performance and sales



growth, firm size, research and development expenditures, and corporate governance index has been witnessed. In contrast, insignificant results have been witnessed between firm performance and capital expenditure, leverage and liquidity.

Table 4-22 Multiple regression results by measuring firm performance by TOQ

Description	Unstandardised B	Significant
Constant	2.49	0.00
Sales Growth	-0.05	0.04
Firm Size	0.10	0.00
Capital Expenditures	-1.00	0.16
Leverage	-0.34	0.13
Research & Development	13.26	0.02
Liquidity	4.54	0.91
Corporate Governance Index	0.02	0.00
R 0.49	R Square	0.24
F statistics 6.557	Significant	0.00

Table 4-22 highlights that the F-statistics is highly significant when firm performance is measured through TOQ. The R square is approximately 24%, which means that 24% of the variation for firm performance is explained by the independent variables when TOQ measures firm performance. The results indicate that more independent variables should be included in the model to obtain more information about the movement in firm performance.

The results of different control variables are in line with the results of previous regression equations where different corporate governance variables were included in the model separately to explore the impact of CG on performance (where firm performance is measured through TOQ). The positive and significant relationship between firm performance and sales growth, firm size, research and development expenditures and corporate



governance index has been witnessed. In contrast, insignificant results have been witnessed between firm performance and capital expenditure, leverage and liquidity.



4.2 Summary

This chapter has presented and discussed the empirical results. The results have been reported in alignment with the study objectives and research questions to assess the effect of internal CG mechanisms on the financial and market performance of the firm. More specifically, this chapter elaborated on the findings and their discussion based on descriptive and regression results. Regression was run through control variables, sales growth, firm size, capital expenditure, leverage, research and development expenditures and liquidity.

In order to make the results more straightforward in accordance with the findings, the small tables were generated separately in accordance with the study objectives. However overall table containing total results of the descriptive analysis, correlation and multiple regression have been reported in the Appendix 1.



Chapter 5 Conclusion and recommendations

5.1 Introduction

This chapter provides information about the key results of the analysis, how the research aim and objectives have been achieved, limitations of the study, contributions of the study, scope for conducting future studies and relevant recommendations based on the analysis of the study.

5.2 Research findings

This study aimed to investigate the impact of various CG mechanisms on the performance of Jordanian non-financial companies listed on the ASC. The data was collected for 95 Jordanian non-financial companies from 2012 to 2018. In order to achieve the aim of the study and answering the research questions, various statistical tools were used for the analysis, such as descriptive analysis, correlation and multiple regression analysis.

The aim of the study was further sub-divided into five research objectives, such as conducting a critical literature review in the field of corporate governance and firm performance, collection of the secondary data for various CG mechanisms and firm performance for Jordanian non-financial firms from 2012 – 2018, data analysis by using various statistical tools (descriptive analysis, correlation and multiple regression analysis), presentation of the findings of the study and compare and contrast the results of the study with previous studies and providing a brief conclusion and recommendations in light of the analysis of this study.

To achieve the first objective, the researcher conducted a critical and systematic review of the existing literature in the field of corporate governance and firm performance. This critical review of the literature is provided in Chapter 2 of the study, which is divided into three sections. The first section provided a brief history of Jordan and the contribution of its various industrial sectors. The section also provided a short history of various



Jordanian regulators and their contributions in the preparation and revision of the country's multiple corporate governance codes. The second section of the literature review chapter provided the theoretical background of the study. Different corporate governance theories were reviewed, such as agency theory, stakeholder theory, stewardship theory and resource dependency theory. This section explained that corporate governance mechanisms have been developed to overcome the agency conflict between managers and shareholders. For this reason, the research questions were generated based on agency theory. The third section of this chapter compared and contrasted the results of various research studies that have been conducted in the field of corporate governance and firm performance.

The second objective of the study was to collect the secondary data for the Jordanian non-financial firms listed on the ASE from 2012 – 2018. The data was gathered for six financial years because the study is focused on the corporate governance mechanisms recommended by the 2012 corporate governance code of Jordan. The secondary data was collected from three primary sources, the official website of the ASE, annual reports of the non-financial companies and Orbits digital database. Secondary data was collected for the study because financial data has no reliability and validity issues as it is audited by external auditors.

The third objective of the study was to analyse the data using various statistical tools, such as descriptive analysis, correlation and regression analysis. The results of the analysis were provided in Chapter 4 of the study. The descriptive analysis tools (mean, median, mode, standard deviation, range, minimum and maximum) were used to provide a summary of the large data set. Correlation was used to explore how dependent and various independent variables were moving together. Regression analysis was used to investigate the relationship between the dependent variable (firm performance, which is measured by ROA, ROE and Tobin's Q) and various independent variables in a standalone manner (sales growth, firm size, capital expenditures, leverage, research and development expenditures, liquidity, the board size, non-executive directors, CEO duality, remuneration

committee, audit committee and nomination committee). Furthermore, regression analysis was also used to explore the impact of firm performance on the corporate governance index. The corporate governance index was generated by combining various corporate governance mechanisms in order to explore its effect on the performance of Jordanian non-financial firms.

F-statistics were used to check the effectiveness of various regression models. The results showed that all the models were highly significant. R-square demonstrated that the independent variables explained high variability on the dependent variable. The study examined the effect of various board-related CG mechanisms (for example, the board size, CEO duality and presence of NEDs). Multiple statistical tests showed that the fixed effect panel data model was valid, so the study used this model.

The firm performance was measured by using accounting-based measures (ROA, ROE) and market-based measures (Tobin Q). The result of the analysis showed that board size and non-executive directors were negatively correlated with firm performance. In contrast, CEO duality and board sub-committees were shown to have a positive relationship with firm performance. The results also showed that there is a positive relationship between corporate governance index and firm performance.

In the case of board size, the results showed a highly significant negative impact of board size on Jordanian companies' performance. The reason for these results might be because Jordanian companies' boards are generally controlled and dominated by large block owners, usually a family member or a family band. This fact might result in the appointment of managers and board members based on friendship or nepotism instead of experience and skill. Large block owners can use their power to influence management decisions to lead to undermining their monitoring function, and this might result in decreasing a company's performance.

In the case of non-executive directors, our results showed a negative and highly significant impact of NEDs on Jordanian firms' performance. The result



is inconsistent with JCGC (2012), which states that at least a third of the board of directors should be independent NEDs to ensure the independence of the board. Besides this, the result is inconsistent with the monitoring hypothesis of agency theory, which states that the presence of a more significant proportion of NEDs on the board adds value to the firm by providing the firm with independent decisions and judgements (Cadbury Report, 1992; Chhaochharia and Grinstein, 2009). The possible explanations for this finding might be: NEDs are part-time workers, who are unfamiliar with the operations and company business; this makes them unable to comprehend the complications and difficulties that face the company, as well as leads to undermining their ability to apply their monitoring functions.

In the case of CEO duality, our results showed a highly significant positive impact of CEOD on the performance of Jordanian non-financial listed companies. This result is inconsistent with the JCGC (2012), which recommends separation of the roles of CEO and chairman to avoid any conflicting interests between them, and also to maintain efficient supervision of management. This finding is also inconsistent with the agency theory that supports the notion of separation between CEO and chairman to minimise the agency problem. On the other hand, the results support the stewardship theory, which emphasises that having one person performing as both CEO and Chairman will increase the company's performance because the enterprise is monitored. It may be beneficial for Jordanian companies to be CEOD because they are friendly with the CEO and it provides proper management and monitoring, as well as providing more stability and durable control.

Furthermore, in Jordan, the president is generally the founder and organiser of the business, so it is more probable that the president will be the CEO due to his or her knowledge and experience about the firm's affairs. Besides, commonly, Jordanian companies have worked in a relatively more uncomplicated business environment, unlike large companies in the markets of developed nations. Therefore, CEO duality can be useful for Jordanian



companies because it could speed up the decision-making process and enhance communication and coordination among the board members.

In terms of establishment of board sub-committees, the study found a positive impact of three board sub-committees, the audit committee, remuneration committee and nomination committee, on firm performance for Jordanian non-financial listed companies. The results of the study are consistent with the JCGC (2012), which recommends establishing committees such as audit, remuneration and nomination committees, and are also consistent with the empirical results related to the board sub-committees.

As discussed before, the corporate governance index was generated based on the Jordanian corporate governance code requirements. Compliance with the corporate governance index was checked for the Jordanian non-financial firms. The findings indicated that the corporate governance index has a positive and highly significant impact on the firm performance of Jordanian companies. The results of the analysis are not consistent with previous studies where the researchers did not find any significant positive relationship between corporate governance index and firm performance (Padgett and Shabbir, 2005; El-Faitouri, 2014).

The last objective is achieved through this chapter, which provides a summary of the entire research study. This chapter also provides recommendations to the reader based on the analysis. The achievement of the five objectives has enabled the researcher to achieve the research aim of the study, which is to investigate the impact of various corporate governance mechanisms on the performance of Jordanian non-financial firms that are listed on the ASE.

5.3 Limitations of the study

The findings of research studies are important, but there are some limitations that researchers face while conducting their studies. In this study, the first



limitation is the sample size of the study that is composed of non-financial companies. Companies in the financial sector are controlled by different reporting regulations and rules as compared to financial companies (Abideal-Allah, 2011). For this reason, financial companies have been removed from the sample. Thus, the size of the sample was reduced from 276 companies to 95 non-financial companies.

The second limitation is to include only some board structure variables including board size, presence of NEDs, CEO duality and board subcommittees. A broader understanding of features of a board structure could also be drawn from the education level, gender and nationality of members. But only the objectively quantifiable variables were included in this study, to avoid bias within the findings. Hence, as the board of directors is considered to be an essential mechanism of the corporate governance that affects a company's performance, this study recommends that future studies determine the impact of some additional board characteristics such as education level, gender and nationality of members on company performance.

5.4 Research contribution

Corporate governance has become a quite important research area that focuses on various mechanisms that are used by the regulators to ensure a proper check and balance and monitor system to protect the stakes of all the stakeholders.

The literature review revealed this crucial importance and highlighted the issues with conflicts of interest between the principals (owners) and (agents) managers (Jensen and Meckling, 1976). Hence, effective corporate governance mechanisms need primarily to ensure shareholders' equity by ensuring that the company's resources are used properly and enhancing investors' confidence to attract more investments (Dennis and McConnell, 2003).



A proper corporate governance structure ensures better decision making, and effective management results in reducing the conflicts and improving the performance of a company. Most of the previous studies related to the various corporate governance mechanisms and their impact on company performance have been conducted by researchers in developed countries and markets, especially in the United States and the United Kingdom, but in developing countries relatively little is known about corporate governance and its mechanisms, particularly in the Middle East and North Africa (MENF), where there are different cultural and economic considerations.

This study makes a new contribution in that it is the first to examine the impact of various corporate governance mechanisms on the accounting performance of Jordanian industrial and services companies listed on the ASE. The data was extracted from different sources: the Orbits database, the ASE official website and annual reports of selected companies. The results of this study will enhance our understanding of corporate governance in terms of agency theory in developing countries, in particular Jordanian companies and companies in the MENA region, where the same cultural and economic circumstances prevail.

5.5 Further studies

There are many opportunities for future research and improvement. Firstly, to enhance the Jordanian banking system, the Central Bank of Jordan issued the Bank Corporate Governance Code in 2006. Therefore, the researcher recommends the need to study the impact of corporate governance on financial firms' performance. The sample of the study should be increased, and the results from such an investigation would improve understanding by providing another perspective of the effect on financial firms.

Secondly, additional study is recommended to examine the impact of the characteristics and features of the board on company performance, in particular the education level, experience, gender and age of the board members. It will effectively help to understand the characteristics and



features of the board of directors and its impact on the performance of the firms more efficiently. Filling the gaps in these areas will help to provide further understanding of the board practices and their effect on the performance of Jordanian companies.

Thirdly, an average economic growth rate of 8.1% over the first 10 years of the 21st century (2000-2010) made Jordan one of the faster-growing economies in the region. This fast growth was due to various reforms conducted by the Jordanian government for the purpose of improving the investment environment and attracting local and foreign investors to invest in the Jordanian market. Besides this, the US invasion of Iraq in 2003 promoted a large number of Iraqi investors to invest in Jordan due to stability in the investment environment. But Jordan's economic problem became appeared since 2011 due to the drain of the country economic resources, and higher state expenditure resulting from the presence of more than 600,000 refugees escape from Syria cause the Civil War, as well as the bad economic and political conditions of the neighbouring countries of Jordan because of the so-called Arab Spring revolutions. Such circumstances have put the brakes on the Jordanian economy where facing severe fiscal strains, these critical condition of Jordanian economic might affect the financial decisions of local and foreign investors alike. Hence, since the data collection of this study that had a cut-off in 2018, further study is recommended to explore the effect of CG on firm performance under different economic conditions.

The adoption of CG principles more than a concept or code that the companies should adopt. Establishing new companies and creating new positions for these companies to gather with in structure, does not mean that we followed of corporate governance mechanisms. Application of corporate governance mechanisms should ensure better decision- making policies, maximizing profits and reducing risk of human interference activities such as cheating. In addition, application and following of corporate governance should maintain the rights and interests of shareholders as well as provide the best measures for financial stability and management effectiveness. Moreover, application and following of corporate governance principles can

best help Jordan's economic interests. Hence, an urgent need has raised for Jordanian companies to properly implementation the corporate governance practice.

Implementation corporate governance principles will lead the better participation in the decision- making process improve the prevention of corruption cases and attract domestic and foreign investment, thus creating such a better investment environment will bring better investment as well as offer more employment opportunities. The concept of corporate governance has many direct and indirect references in many legal clauses and items; to name but a few, there is the companies Act 22 for the year of 1997 and its amendments, and the securities Act No 28 for the year 2000, and the law that regulates accounting for legal professionals (Law No. 73 of 2003). These legal references allows to use of corporate governance principles in Jordanian companies.

Implementation of corporate governance principles will attract domestic and foreign investment, thus creating a better investment environment that will bring more employment opportunities. Despite efforts made by the Jordanian governance to ensure the implementation of corporate governance standards, it can still be observed that numerous companies are not implementing the corporate governance mechanisms effectively to protect the stakes of all the stakeholder groups. Therefore, the researcher recommends further cooperation between the public and private sectors to intensify their efforts to implement the JCGC in a more effective manner t in order to improve the firm performance listed on the ASE and thus attract more local and foreign investors to invest in the Jordanian market.



5.6 Summary

This chapter closed the research cycle that was open in Chapter 1 by explaining how to achieve the research objectives presented in Chapter 1, and also presents the key results of the research based on the statistical analysis, . This chapter also provides information about the limitations that the researcher faced while conducting the study. Moreover, this chapter introduces the study contributions and the scope of further studies in the future and relevant recommendations based on the results of the study analysis.



References

Abed, S., Al-Attar, A., & Suwaidan, M. (2012), Corporate Governance and Earnings Management: Jordanian Evidence, International Business Research, Vol. 5 (1), p. 216

Abor, J. (2007), Corporate Governance and Financing Decisions of Ghanaian Listed Firms, Corporate Governance, The International Journal of Effective Board Performance, Vol. 7(1), pp. 83 - 92

Adams, R. B., Almeida, H., & Ferreira, D. (2007), Powerful CEOs and their Impact on Corporate Performance, Review of Financial Studies, Vol. 18 (4), pp. 1403 - 1432

Adams, R., & Mehran, H. (2003), Is Corporate Governance Different for Bank Holding Companies? Economic Policy Review (19320426), Vol. 9 (1), p. 123

Adams, R., & Mehran, H. (2005), Corporate Performance, Board Structure and its Determinants in the Banking Industry, EFA 2005 Moscow Meetings

Adjaoud, F., Zeghal, D., & Andaleeb, S. (2007), The Effect of Board's Quality on Performance: A Study of Canadian Firms, Corporate Governance, An International Review, Vol. 15 (4), pp. 623 - 635, DOI: 10.1111/j.1467-8683.2007.00592.x

Aggarwal, R., Erel, I., Ferreira, M., & Matos, P. (2011), Does Governance Travel Around The World? Evidence from Institutional Investors, Journal of Financial Economics, Vol. 100 (1), pp. 154 - 181

Aggarwal, R., Klapper, L., & Wysocki, P. D. (2005), Portfolio Preferences of Foreign Institutional Investors, Journal of Banking & Finance, Vol. 29 (12), pp. 2919 - 2946



Agrawal, A., & Knoeber, C. R. (1996), Firm Performance and Mechanisms to Control Agency Problems between Managers and Shareholders, Journal of Financial and Quantitative Analysis, Vol. 31 (03), pp. 377 - 397

Aguilera, R. V., & Cuervo-Cazurra, A. (2009), Codes of Good Governance, Corporate Governance: An International Review, Vol. 17 (3), pp. 376 - 387

Ahmed, K., Hossain, M., & Adams, M. B. (2006), The Effects of Board Composition and Board Size on the Informativeness of Annual Accounting Earnings, Corporate Governance An International Review, Vol. 14 (5), pp. 418 - 431. DOI: 10.1111/j.1467-8683.2006.00515.x

Akbar, S., El-Faitouri, R., Hughes, J. P., and Shah, S. Z. A. (2016), More on the Relationship between Corporate Governance and Firm Performance in the UK: Evidence from the Application of Generalized Method of Moments Estimation, Research in International Business and Finance, https://www.researchgate.net/publication/299489401 Accessed on 25 April 2020, Available at

Al-Ahdal, W. M., Alsamhi, M. H., Tabash, M. I. and Farhan, N. H. S. (2020), The Impact of Corporate Governance on Financial Performance of Indian and GCC Listed Firms: An Empirical Investigation, Research in International Business and Finance, Vol. 51, pp. 1 – 13

Al-Akra, M., Jahangir Ali, M., & Marashdeh, O. (2009), Development of Accounting Regulation in Jordan, The International Journal of Accounting, Vol. 44 (2), pp. 163 - 186

Al-Basheer, M. (2003), Corporate Governance and Auditor, Jordan Association of Certified Public Accountants, 5th Professional Conference, 24-25 September 2003, Amman

Aldamen, H., Duncan, K., Kelly, S., McNamara, R., & Nagel, S. (2012), Audit Committee Characteristics and Firm Performance during the Global Financial



Crisis, Accounting & Finance, Vol. 52 (4), pp. 971 - 1000: 10.1111/j.1467-629X.2011.00447.x

Alexander, D., Britton, A., & Jorissen, A. (2007), International Financial Reporting and Analysis: Cengage Learning, Emea

Al-Fanik, F. (2005) Corporate Governance in Jordan, Al Rai Newspaper, 12575 [Arabic]

Al-Fayoumi, N., Abuzayed, B., & Alexander, D. (2010), Ownership Structure and Earnings Management in Emerging Markets: The Case of Jordan, International Research Journal of Finance and Economics, Vol. 38, pp. 28 - 47

http://www.aljazylaw.com/arabic/pdf/hawkamat alsherkat2.pdf,

Al-Jazi, O. (2007), Corporate Governance in Jordan (Arabic), Accessed on 11 January 2019, Available online at:

http://www.aljazylaw.com/arabic/pdf/hawkamat_alsherkat2.pdf.

Aljifri, K., & Moustafa, M. (2007), The Impact of Corporate Governance Mechanisms on the Performance of UAE Firms: An Empirical Analysis, Journal of Economic and Administrative Sciences, Vol. 23 (2), pp. 71 – 93, 10.1108/10264116200700008

Al-Khouri, R. (2006), Corporate Governance and Firms Value in Emerging Markets: The Case of Jordan, Journal of Transnational Management, Vol. 12 (1), pp. 25 – 49, DOI: 10.1300/J482v12n01_03

Allen, F. (2005), Corporate Governance in Emerging Economies, Oxford Review of Economic Policy, Vol. 21 (2), pp. 164 - 177

Al-Matari, E. M., Al-Swidi, A. K., Fadzil, F. H., & Al-Matari, Y. A. (2012), The Impact of Board Characteristics on Firm Performance: Evidence from Non-



financial Listed Companies in Kuwaiti Stock Exchange, International Journal of Accounting and Financial Reporting, Vol. 2 (2), pp.310 - 332

Al-Matari, Y. A., Al-Swidi, A. K., Fadzil, & Al-Matari, E. M. (2012), Board of Directors, Audit Committee Characteristics and the Performance of Saudi Arabia Listed Companies, International Review of Management and Marketing, Vol. 2 (4), pp. 241 - 251

Almeida, H., & Campello, M. (2007), Financial Constraints, Asset Tangibility and Corporate Investment, Review of Financial Studies, Vol. 20 (5), pp. 1429 – 1460

Almustafa, H. H. (2017), National Governance, Corporate Governance and Firm Performance: Empirical Evidence from Two MENA Countries – Jordan and UAE, PhD Dissertation, University of Salford Manchester

Al-Muhtaseb, B. (2010), The Impact of Foreign Direct Investment on the Economic Growth of Jordan (1990-2006), Administrative Sciences, Vol. 36 (2)

Al-Najjar, B., & Taylor, P. (2008), The Relationship between Capital Structure and Ownership Structure: New Evidence from Jordanian Panel Data, Managerial Finance, Vol. 34 (12), pp. 919 – 933

Alqatan, A., Chbib, I. and Hussainey, K. (2019), How Does Board Structure Impact on Firm Performance in the UK? Corporate Board: Role, Duties of Composition, Vol. 15 (2), pp. 18 – 27

Amman Stock Exchange (2020), Corporate Profile, Accessed on 13 June 2020, Available at

https://www.ase.com.jo/en/Corporate-Profile/About-Us/About-ASE



Amman Stock Exchange (2020), Investor Guide, Accessed on 13 June 2020, Available at

https://www.ase.com.jo/en/Investors/Investor-Corner/Investor-Guide

Ameer, R. (2014), Financial Constraints and Corporate Investment in Asian Countries, Journal of Asian Economics, Vol. 33, pp. 44 – 55 www.ase.com.jo.

Amman Stock Exchange, Available online at www.ase.com.jo.

Anderson, R. C., Bates, T. W., Bizjak, J. M. and Lemmon, M. L. (2020) Corporate Governance & Firm Diversification, Financial Management, Vol. 29 (1), pp. 5 – 22, Accessed on 22 May 2020, Available at https://www.jstor.org/stable/3666358

Anderson, R. C., & Bizjak, D. M. (2003), Founding-Family Ownership and Firm Performance: Evidence from the S&P 500, The Journal of Finance, Vol. 58 (3), pp. 1301 - 1327

Anderson, R. C., Mansi, S. A., & Reeb, D. M. (2004), Board Characteristics, Accounting Report Integrity, and the Cost of Debt, Journal of Accounting and Economics, Vol. 37 (3), pp. 315 - 342

Andrade, G., & Kaplan, S. N. (1998), How Costly Is Financial (Not Economic) Distress? Evidence from Highly Leveraged Transactions That Became Distressed, The Journal of Finance, Vol. 53 (5), pp. 1443 - 149

Andres, C. (2008), Large Shareholders and Firm Performance: An Empirical Examination of Founding-Family Ownership, Journal of Corporate Finance, Vol. 14 (4), pp. 431 - 445



Andres, P. D., & Vallelado, E. (2008), Corporate Governance in Banking: The Role of the Board of Directors, Journal of Banking & Finance, Vol. 32 (12), pp. 2570 - 2580

Angelopoulou, E., & Gibson, H. D. (2009), The Balance Sheet Channel of Monetary Policy Transmission: Evidence from the United Kingdom, Economica, Vol. 76 (304), pp. 675 – 703

Ararat, M., & Dallas, G. (2011), Corporate Governance in Emerging Markets: Why it Matters to Investors and What They Can Do About It Ardalan, K. (2012), On the Role of Paradigms in Finance: Ashgate Publishing, Ltd.

Arnold, B., & De Lange, P. (2004), Enron: An Examination of Agency Problems, Critical Perspectives on Accounting, Vol. 15 (6), pp. 751 - 765

Arosa, B., Iturralde, T., & Maseda, A. (2010), Ownership Structure and Firm Performance in Non-Listed Firms: Evidence from Spain, Journal of Family Business Strategy, Vol. 1 (2), pp. 88 - 96

Arosa, B., Iturralde, T., & Maseda, A. (2012), The Board Structure and Firm Performance in SMEs: Evidence from Spain, Investigaciones Europeas De Dirección Y Economía De La Empresa

Azam, M., & Shah, S. A. (2011), Internal Financial Constraints, External Financial Constraints and Investment Choice: Evidence from Pakistani Firms, Australian Journal of Business and Management Research, Vol. 1 (8), pp. 1018 – 1037

Babbie, E. (2012), The Practice of Social Research: Cengage Brain

Baek, J.-S., Kang, J.-K., & Suh Park, K. (2004), Corporate Governance and Firm Value: Evidence from the Korean Financial Crisis, Journal of Financial Economics, Vol. 71 (2), pp. 265 - 313



Baliga, B., Moyer, R. C., & Rao, R. S. (1996), CEO Duality and Firm Performance: What's the Fuss? Strategic Management Journal, Vol. 17 (1), pp. 41 - 53

Baltagi, B. H., & Giles, M. D. (1998), Panel Data Methods, Statistics Textbooks and Monographs, Vol. 155, pp. 291 - 324

Baranchuk, N., & Dybvig, P. H. (2009), Consensus in Diverse Corporate Boards. Review of Financial Studies, Vol. 22 (2), pp. 715 - 747

Barton, D., & Wong, S. C. (2006), Improving Board Performance in Emerging Markets, Mckinsey Quarterly, Vol. 1, pp. 74

Bauer, R., Guenster, N., & Otten, R. (2004), Empirical Evidence on Corporate Governance in Europe: The Effect on Stock Returns, Firm Value and Performance, Journal of Asset Management, Vol. 5 (2), pp. 91 - 104

Baum, C. F., Schäfer, D., & Talavera, O. (2011). The Impact of the Financial System's Structure on Firms' Financial Constraints, Journal of International Money and Finance, Vol. 30 (4), pp. 678 – 691

Bavarsad, B., Sinaei, H., & Delavaripour, J. (2013), Study on the Relationship Between Financial Constraints and Stock Return in Tehran Stock Exchange

Baysinger, B., & Hoskisson, R. E. (1990), The Composition of Boards of Directors and Strategic Control: Effects on Corporate Strategy, Academy of Management Review, Vol. 15 (1), pp. 72 - 87

Becht, M., Bolton, P., & Röell, A. (2003), Corporate Governance and Control, Handbook of The Economics of Finance, Vol. 1, pp. 1 - 109

Beck, T., Degryse, H., De Haas, R., & Van Horen, N. (2017), When Arm's Length is too Far: Relationship Banking over the Business Cycle, Journal of Financial Economics, Forthcoming



Beck, T., Demirgüç-Kunt, A., Laeven, L., & Maksimovic, V. (2006), The Determinants of Financing Obstacles, Journal of International Money and Finance, Vol. 25 (6), pp. 932 – 952

Beiner, S., Drobetz, W., Schmid, M. M., & Zimmermann, H. (2006), An Integrated Framework of Corporate Governance and Firm Valuation, European Financial Management, Vol. 12 (2), pp. 249 - 283

Bekaert, G., Harvey, C. R., & Lundblad, C. (2007), Liquidity and Expected Returns: Lessons from Emerging Markets, Review of Financial Studies, Vol. 20 (6), pp. 1783 - 1831

Bell, E., Bryman, A. and Harley, B. (2018), Business Research Methods, 5th edition, Oxford: Oxford University Press

Benito, A. (2005), Financial Pressure, Monetary Policy Effects and Inventories: Firm-Level Evidence from a Market-Based and a Bank-Based Financial System, Economica, Vol. 72 (2), 201 – 224

Bennedsen, M., Kongsted, H. C., & Nielsen, K. M. (2008), The Causal Effect of Board Size in The Performance of Small and Medium-Sized Firms, Journal of Banking & Finance, Vol. 32 (6), pp. 1098 - 1109

Berg, B. L. (2004), Qualitative Research Methods for the Social Sciences, 5th Edition, Boston, MA: Pearson

Berle, A. A., & Means, G. C. (1932), The Modern Corporation and Private Property: Transaction Books

Bhagat, S., & Bolton, B. (2008), Corporate Governance and Firm Performance, Journal of Corporate Finance, Vol. 14 (3), pp. 257 - 273



Bhaumik, S., Driffield, N., Gaur, A., Mickiewicz, T. and Vaaler, P. (2019), Corporate Governance and MNE Strategies in Emerging Economies, Journal of World Business, Vol. 54, pp. 234 – 243

Bhaumik, S. K., & Gregoriou, A. (2010), Family ownership, Tunnelling and Earnings Management: A Review of the Literature, Journal of Economic Surveys, Vol. 24 (4), pp. 705 - 730

Black, B. (2001), The Corporate Governance Behaviour and Market Value of Russian Firms, Emerging Markets Review, Vol. 2 (2), pp. 89 - 108

Black, B. S., Jang, H., & Kim, W. (2006), Does Corporate Governance Predict Firm's Market Values? Evidence from Korea, Journal of Law, Economics, and Organization, Vol. 22 (2), pp. 366 - 413

Black, B. S., Love, I., & Rachinsky, A. (2006), Corporate Governance Indices and Firm's Market Values: Time Series Evidence from Russia, Emerging Markets Review, Vol. 7 (4), pp. 361 - 379

Bloom, N., & Van Reenen, J. (2007), Measuring and Explaining Management Practices Across Firms and Countries, The Quarterly Journal of Economics, Vol. 122 (4), pp. 1351 - 1408

Boone, A. L., Casares Field, L., Karpoff, J. M., & Raheja, C. G. (2007), The Determinants of Corporate Board Size and Composition: An Empirical Analysis, Journal of Financial Economics, Vol. 85 (1), pp. 66 - 101

Booth, J. R., & Deli, D. N. (1996), Factors Affecting The Number of Outside Directorships Held By CEOs, Journal of Financial Economics, Vol. 40 (1), pp. 81 - 104

Booth, J. R., Cornett, M. M., & Tehranian, H. (2002), Boards of Directors, Ownership, and Regulation, Journal of Banking & Finance, Vol. 26 (10), pp. 1973 - 1996



Bond, S., Harhoff, D., & Van Reenen, J. (1999), Investment, R&D and Financial Constraints in Britain and Germany (No. W99/05), Institute for Fiscal Studies, London

Borghesi, R., Houston, J., & Naranjo, A. (2007), Value, Survival, and the Evolution of Firm Organizational Structure, Financial Management, Vol. 36 (3), pp. 5 - 31

Boyd, B. K. (1995), CEO Duality and Firm Performance: A Contingency Model, Strategic Management Journal, Vol. 16 (4), pp. 301 - 312

Bozec, R. (2005), Boards of Directors, Market Discipline and Firm Performance, Journal of Business Finance & Accounting, Vol. 32 (10), pp. 1921 - 1960

Braendle, U. (2019), Corporate Governance Code Compliance and Financial Performance: The Case of Austrian Stock Listed Companies, Investment Management and Financial Innovations, Vol. 16 (3), pp. 131 – 141

Breusch, T. S., & Pagan, A. R. (1980), The Lagrange Multiplier Test and Its Applications to Model Specification in Econometrics, The Review of Economic Studies, Vol. 47 (1), pp. 239 - 253

Brickley, J. A., Coles, J. L., & Jarrell, G. (1997), Leadership Structure: Separating the CEO and Chairman of the Board, Journal of Corporate Finance, Vol. 3 (3), pp. 189 - 220

Brickley, J. A., Lease, R. C., & Smith, C. W. (1988), Ownership Structure and Voting on Antitakeover Amendments, Journal of Financial Economics, Vol. 20, pp. 267 - 291

Bruner, R. F., Conroy, R. M., Estrada, J., Kritzman, M., & Li, W. (2002), Introduction to Valuation in Emerging Markets Review, Vol. 3 (4), pp. 310 -



Bryman, A. (2016), Social Research Methods, 5th Edition, Oxford: Oxford University Press

Buch, C. M., Kesternich, I., Lipponer, A., & Schnitzer, M. (2014), Financial Constraints and Foreign Direct Investment: Firm-Level Evidence, Review of World Economics, Vol. 150 (2), pp. 393 – 420

Burrell, G., & Morgan, G. (2005), Sociological Paradigms and Organisational Analysis, Ashgate Publishing Company, England

Cadbury, A. (1999), Foreword in World Bank Report, Corporate Governance: A Framework for Implementation Overview, Washington, DC: World Bank

Cadbury, A., Butler, J., Lipworth, S., Macdonald, N., Smith, A. H., Brown, S., Collum, H. (1992), Committee on the Financial Aspects of Corporate Governance Gee, London

Campello, M., Graham, J. R., & Harvey, C. R. (2010), The Real Effects of Financial Constraints: Evidence from a Financial Crisis, Journal of Financial Economics, Vol. 97 (3), 470 – 487

Canton, E., Grilo, I., Monteagudo, J., & Van der Zwan, P. (2013), Perceived Credit Constraints in the European Union, Small Business Economics, Vol. 41 (3), pp. 701 – 715

Caplan, D. (1999), Internal Controls and the Detection of Management Fraud, Journal of Accounting Research, pp. 101 - 117

Carpenter, R. E., Fazzari, S. M., Petersen, B. C., Kashyap, A. K., & Friedman, B. M. (1994), Inventory Investment, Internal-Finance Fluctuations, and the Business Cycle, Brookings Papers on Economic Activity, Vol. 4 (2), pp. 75 – 138



Carpenter, R. E., & Petersen, B. C. (2002), Capital Market Imperfections, High-Tech Investment, and new Equity Financing, The Economic Journal, Vol. 12 (4), pp. 54 – 72

Carter, D. A., Simkins, B. J., & Simpson, W. G. (2003), Corporate Governance, Board Diversity, and Firm Value, Financial Review, Vol. 38 (1), pp. 33 - 53

Cascio, W. F. (2004), Board Governance: A Social Systems Perspective, The Academy of Management Executive, Vol. 18 (1), pp. 97 - 100

Chahine, S., & Tohmé, N. S. (2009), Is CEO Duality Always Negative? An Exploration of CEO Duality and Ownership Structure in the Arab IPO Context, Corporate Governance: An International Review, Vol. 17 (2), pp. 123 - 141

Chan, H., Chang, X., Faff, R., & Wong, G. (2010), Financial Constraints and Stock Returns - Evidence from Australia, Pacific-Basin Finance Journal, Vol. 18 (3), pp. 306 – 318

Chen, C. J., & Jaggi, B. (2001), Association between Independent Non-Executive Directors, Family Control and Financial Disclosures in Hong Kong, Journal of Accounting and Public Policy, Vol. 19 (4), pp. 285 - 310

Chen, X., Harford, J., & Li, K. (2007), Monitoring: Which Institutions Matter? Journal of Financial Economics, Vol. 86 (2), pp. 279 - 305



Cheng, S., Evans III, J. H., & Nagarajan, N. J. (2008), Board Size and Firm Performance: The Moderating Effects of the Market for Corporate Control, Review of Quantitative Finance and Accounting, Vol. 31 (2), pp. 121 - 145

Chenhall, R. H., & Moers, F. (2007), The Issue of Endogeneity within Theory-Based, Quantitative Management Accounting Research, European Accounting Review, Vol. 16 (1), pp. 173 - 196

Chhaochharia, V., & Grinstein, Y. (2009), CEO Compensation and Board Structure, The Journal of Finance, Vol. 64 (1), pp. 231 - 261

Choi, H. M., Sul, W., & Min, S. K. (2012), Foreign Board Membership and Firm Value in Korea, Management Decision, Vol. 50 (2), pp. 207 - 233

Choi, J. J., Park, S. W., & Yoo, S. S. (2007), The Value of Outside Directors: Evidence from Corporate Governance Reform in Korea, Journal of Financial and Quantitative Analysis, Vol. 42 (4), pp. 941- 958

Cifci, I., Tatoglu, E., Wood, G., Demirbag, M. and Zaim, S. (2019), Corporate Governance and Firm Performance in Emerging Markets: Evidence from Turkey, International Business Review, Vol, 28, pp. 90 – 103

Claessens, S. (2006), Corporate Governance and Development, The World Bank Research Observer, Vol. 21 (1), pp. 91 - 122

Claessens, S., & Yurtoglu, B. B. (2013), Corporate Governance in Emerging Markets: A Survey, Emerging Markets Review, Vol. 15, pp. 1 - 33

Clarke, T. (2004), Theories of Corporate Governance, New York, NY: Routledge

Clayman, M. R., Fridson, M. S., & Troughton, G. H. (2011), Corporate Finance: A Practical Approach, Hoboken, NJ: John Wiley & Sons



Cleary, S. (1999), The Relationship between Firm Investment and Financial Status, The Journal of Finance, Vol. 54 (2), pp. 673 – 692

Cochran, P. L., & Wood, R. A. (1984), Corporate Social Responsibility and Financial Performance, Academy of Management Journal, Vol. 27 (1), pp. 42 - 56

Coffee Jr, J. C. (1999), Future as History: The Prospects for Global Convergence in Corporate Governance and its Implications, Nw. UL Rev., Vol.93, pp. 641

Cohen, L., Manion, L. and Morrison, K. (2017), Research Methods in Education, 8th edition, London: Taylor and Francis

Coles, J. L., Daniel, N. D., & Naveen, L. (2008), Boards: Does One Size Fit All? Journal of Financial Economics, Vol. 87 (2), pp. 329 - 356

Collis, J., & Hussey, R. (2009), Business Research: A Practical Guide for Undergraduate and Postgraduate Students, London: Palgrave Macmillan

Conger, J. A., Finegold, D., & Lawler, E. (1998), Appraising Boardroom Performance, Harvard Business Review, Vol. 76, pp. 136 – 164

Connelly, J. T., & Limpaphayom, P. (2004), Board Characteristics and Firm Performance: Evidence from the Life Insurance Industry in Thailand, Chulalongkorn Journal of Economics, Vol. 16 (2), pp. 101 - 124

Cremers, K., & Nair, V. B. (2005), Governance Mechanisms and Equity Prices, The Journal of Finance, Vol. 60 (6), pp. 2859 - 2894

D'Souza, J., Megginson, W., & Nash, R. (2005), Effect of Institutional and Firm-Specific Characteristics on Post-Privatization Performance: Evidence from Developed Countries, Journal of Corporate Finance, Vol. 11 (5), pp.



Dahlquist, M., Pinkowitz, L., Stulz, R. M., & Williamson, R. (2003), Corporate Governance and the Home Bias, Journal of Financial and Quantitative Analysis, Vol. 38 (1), pp. 87 - 110

Dahya, J., Lonie, A., & Power, D. (1996), The Case for Separating the Roles of Chairman and CEO: An Analysis of Stock Market and Accounting Data, Corporate Governance: An International Review, Vol. 4 (2), pp. 71 - 77

Dahya, J., McConnell, J. J., & Travlos, N. G. (2007), The Cadbury Committee, Corporate Performance, and Top Management Turnover, The Journal of Finance, Vol. 57 (1), pp. 461 -483

Daily, C. M., & Dalton, D. R. (1993), Board of Directors Leadership and Structure: Control and Performance Implications, Entrepreneurship Theory and Practice, Vol. 17, pp. 65 – 69

Daily, C. M., Dalton, D. R., & Cannella, A. A. (2003), Corporate Governance: Decades of Dialogue and Data, Academy of Management Review, Vol. 28 (3), pp. 371 - 382

Daines, R., & Klausner, M. (2001), Do IPO Charters Maximize Firm Value? Antitakeover Protection in IPOs, Journal of Law, Economics, and Organization, Vol. 17 (1), pp. 83 - 120

Dalton, D. R., Daily, C. M., Ellstrand, A. E., & Johnson, J. L. (1998), Meta-Analytic Reviews of Board Composition, Leadership Structure, and Financial Performance, Strategic Management Journal, Vol. 19 (3), pp. 269 - 290

Dalton, D., & Kesner, I. (1987), Composition and CEO Duality in Boards of Directors: An International Perspective, Journal of International Business Studies, Vol. 18 (3), pp. 33 - 42



Davis, E. P., & Steil, B. (2001), Institutional Investors, Cambridge, MA: MIT Press

Davis, J. H., Schoorman, F. D., & Donaldson, L. (1997), Toward A Stewardship Theory of Management, Academy of Management Review, Vol. 22 (1), pp. 20 - 47

De Andres, P., Azofra, V., & Lopez, F. (2005), Corporate Boards in OECD Countries: Size, Composition, Functioning and Effectiveness, Corporate Governance: An International Review, Vol. 13 (2), pp. 197 - 210

Dechow, P. M., Sloan, R. G., & Sweeney, A. P. (1996), Causes and Consequences of Earnings Manipulation: An Analysis of Firms Subject to Enforcement Actions by the Security Contemporary Accounting Research, Vol. 13 (1), pp. 1 - 36

Defond, M. L., & Hung, M. (2004), Investor Protection and Corporate Governance: Evidence from Worldwide CEO Turnover, Journal of Accounting Research, Vol. 42 (2), pp. 269 - 312

Demsetz, H. (1983), Structure of Ownership and the Theory of the Firm, The Journal of Economics, Vol. 26, pp. 375 – 394

Demsetz, H., & Lehn, K. (1985), The Structure of Corporate Ownership: Causes and Consequences, The Journal of Political Economy, Vol. 93 (6), pp. 1155 - 1177

Denis, D. J., & Denis, D. K. (1994), Majority Owner-Managers and Organizational Efficiency, Journal of Corporate Finance, Vol. 1 (1), pp. 91 - 118

Denis, D. K., & McConnell, J. J. (2003), International Corporate Governance, Journal of Financial and Quantitative Analysis, Vol. 38 (1), pp. 1 - 36



Detthamrong, U., Chancharat, N. and Vithessonthi, C. (2017), Corporate Governance, Capital Structure and Firm Performance: Evidence from Thailand, Research in International Business and Finance, Vol. 42, pp. 689 – 709

DeZoort, F. (1998), An Analysis of Experience Effects on Audit Committee Members Oversight Judgments, Accounting, Organizations and Society, Vol. 23 (1), pp. 1 - 21

Dhumale, R. (1998), Earnings Retention as a Specification Mechanism in Logistic Bankruptcy Models: A Test of the Free Cash Flow Theory, Journal of Business Finance & Accounting, Vol. 25 (7), pp. 1005 – 1023

Doidge, C., Andrew Karolyi, G., & Stulz, R. M. (2007), Why Do Countries Matter so Much for Corporate Governance? Journal of Financial Economics, Vol. 86 (1), pp. 1 - 39

Donaldson, L., & Davis, J. H. (1991), Stewardship Theory or Agency Theory: CEO Governance and Shareholder Returns, Australian Journal of Management, Vol. 16 (1), pp. 49 - 64

Donaldson, L., & Davis, J. H. (1994), Boards and Company Performance Research Challenges the Conventional Wisdom, Corporate Governance: An International Review, Vol. 2 (3), pp. 151 - 160

Dong, M., & Ozkan, A. (2008), Institutional Investors and Director Pay: An Empirical Study of UK Companies, Journal of Multinational Financial Management, Vol. 18 (1), pp. 16 – 29

Douma, S., George, R., & Kabir, R. (2006), Foreign and Domestic Ownership, Business Groups, and Firm Performance: Evidence from a Large Emerging Market, Strategic Management Journal, Vol. 27 (7), pp. 637 - 657



Durisin, B., & Puzone, F. (2009), Maturation of Corporate Governance Research, 1993–2007: An Assessment, Corporate Governance: An International Review, Vol. 17 (3), pp. 266 - 291

Durnev, A., & Kim, E. (2005), To Steal or Not to Steal: Firm Attributes, Legal Environment, and Valuation, The Journal of Finance, Vol. 60 (3), pp. 1461 - 1493

Dyck, A., & Zingales, L. (2004), Private Benefits of Control: An international comparison, The Journal of Finance, Vol. 59 (2), pp. 537 - 600

Ebaid, I. E. S. (2011), Corporate Governance Practices and Auditor's Client Acceptance Decision: Empirical Evidence from Egypt, Corporate Governance, Vol. 11 (2), pp. 171 - 183

Ehikioya, B. I. (2009), Corporate Governance Structure and Firm Performance in Developing Economies: Evidence from Nigeria, Corporate Governance, Vol. 9 (3), pp. 231 - 243

Eisenberg, T., Sundgren, S., & Wells, M. T. (1998), Larger Board Size and Decreasing Firm Value in Small Firms, Journal of Financial Economics, Vol. 48 (1), pp. 35 - 54

Eisenhardt, K. M. (1989), Agency Theory: An Assessment and Review. Academy of Management Review, Vol. 14 (1), pp. 57 - 74

El-Faitouri, R. (2014), Board of Directors and Tobin's Q: Evidence from UK Firms, Journal of Finance and Accounting, Vol. 2(4), pp. 82 – 99

Elsayed, K. (2007), Does CEO Duality Really Affect Corporate Performance? Corporate Governance: An International Review, Vol. 15 (6), pp. 1203 - 1214

Eng, L. L., & Mak, Y. T. (2003), Corporate Governance and Voluntary Disclosure, Journal of Accounting and Public Policy, Vol. 22 (4), pp. 325 -



Erhardt, N. L., Werbel, J. D., & Shrader, C. B. (2003), Board of Director Diversity and Firm Financial Performance, Corporate Governance: An International Review, Vol. 11 (2), pp. 102 - 111

Erkens, D. H., Hung, M., & Matos, P. (2012), Corporate Governance in the 2007–2008 Financial Crises: Evidence from Financial Institutions Worldwide, Journal of Corporate Finance, Vol. 18 (2), pp. 389 - 411

Estrin, S., Hanousek, J., Kočenda, E., & Svejnar, J. (2009), The Effects of Privatization and Ownership in Transition Economies, Journal of Economic Literature, pp. 699 - 728

European Corporate Governance Institute (2020), List of Published Codes, Accessed on 30 June 2020, Available at https://ecgi.global/content/codes?page=7

Fama, E. F. (1980), Agency Problems and the Theory of the Firm, The Journal of Political Economy, pp. 288 - 307

Fama, E. F., & Jensen, M. C. (1983), Agency Problems and Residual Claims, Journal of Law and Economics, Vol. 26 (2), pp. 327 - 349

Fama, E. F., & Jensen, M. C. (1983), Separation of Ownership and Control, Journal of Law and Economics, Vol. 26 (2), pp. 301 - 325

Fan, J. P., Wei, K., & Xu, X. (2011), Corporate Finance and Governance in Emerging Markets: A Selective Review and an Agenda for Future Research, Journal of Corporate Finance, Vol. 17 (2), pp. 207 - 214

Fang, V. W., Noe, T. H., & Tice, S. (2009), Stock Market Liquidity and Firm Value, Journal of Financial Economics, Vol. 94 (1), pp. 150 - 169



Fazzari, S. M., Hubbard, R. G., Petersen, B. C., Blinder, A. S., & Poterba, J. M. (1988), Financing Constraints and Corporate Investment, Brookings papers on Economic Activity, Vol. 8 (1), pp. 141 – 206

Financial Reporting Council (2016), UK Corporate Governance Code, The Combined Code on Corporate Governance

Foroughi, M., & Fooladi, M. (2011), Corporate Ownership Structure and Firm Performance: Evidence from Listed Firms in Iran, International Proceedings of Economics Development & Research, Vol. 20

Franks, J., & Mayer, C. (2001), Ownership and Control of German Corporations, Review of Financial Studies, Vol. 14 (4), pp. 943 - 977

Fraser, D. R., Zhang, H., & Derashid, C. (2006), Capital Structure and Political Patronage: The Case of Malaysia, Journal of Banking & Finance, Vol. 30 (4), pp. 1291 - 1308

Fukuyama, F. (1992), The End of History and the Last Man, New York, NY: Free Press

Gabrielsson, J. (2007), Correlates of Board Empowerment in Small Companies, Entrepreneurship Theory and Practice, Vol. 31 (5), pp. 687 - 711

Gabrielsson, J., & Winlund, H. (2000), Boards of Directors in Small and Medium-Sized Industrial Firms: Examining the Effects of the Board's Working Style on Board Task Performance, Entrepreneurship & Regional Development, Vol. 12 (4), pp. 311 - 330

García Olalla, M., & García Ramos, R. (2010), Family Ownership, Structure and Board of Directors Effectiveness: Empirical Evidence from European Firms. 9th annual IFERA Conference



Garen, J. E. (1994), Executive Compensation and Principal-Agent Theory, Journal of Political Economy, pp. 1175 - 1199

Gaur, A. S., & Delios, A. (2006), Business Group Affiliation and Firm Performance During Institutional Transition, Paper presented at the Academy of Management Proceedings

Ghazali, A. (2010), Analyzing the Relationship between Foreign Direct Investment Domestic Investment and Economic Growth for Pakistan, International Research Journal of Finance and Economics, Vol. 47, pp. 123 - 131

Giannetti, M., & Simonov, A. (2006), Which Investors Fear Expropriation? Evidence from Investors Portfolio Choices, The Journal of Finance, Vol. 61 (3), pp. 1507 - 1547

Gibson, M. S. (2003), Is Corporate Governance Ineffective in Emerging Markets? Journal of Financial and Quantitative Analysis, Vol. 38 (1), pp. 231 - 250

Gill, A., & Mathur, N. (2011), Board Size, CEO Duality, and the Value of Canadian Manufacturing Firms, Journal of Applied Finance and Banking, Vol. 1 (3), pp. 1 – 13

Gillan, S. L. (2006), Recent Developments in Corporate Governance: An Overview, Journal of Corporate Finance, Vol. 12 (3), pp. 381 - 402

Gillan, S. L., & Starks, L. T. (2000), Corporate Governance Proposals and Shareholder Activism: The Role of Institutional Investors, Journal of Financial Economics, Vol. 57 (2), pp. 275 - 305

Gillan, S., & Starks, L. (1998), A Survey of Shareholder Activism: Motivation and Empirical Evidence, Contemporary Finance Digest, Vol. 2 (3), pp. 10 -



Gillan, S., & Starks, L. (2003), Corporate Governance, Corporate Ownership, and the Role of Institutional Investors: A Global Perspective, Weinberg Centre For Corporate Governance Working Paper

Glaeser, E., Johnson, S., & Shleifer, A. (2001), Coase versus the Coasians, The Quarterly Journal of Economics, Vol. 116 (3), pp. 853 - 899

Gompers, P., Ishii, J., & Metrick, A. (2003), Corporate Governance and Equity Prices, The Quarterly Journal of Economics, Vol. 118 (1), pp. 107 - 156

Goodstein, J., Gautam, K., & Boeker, W. (1994), The Effects of Board Size and Diversity on Strategic Change, Strategic Management Journal, Vol. 15 (3), pp. 241 - 250

Gordini, N. (2012), The Impact of Outsiders on Small Family Firm Performance: Evidence from Italy, Vol. 4 (2), pp. 14 – 35

Greene, W. H. (2012), Econometric Analysis, 5th Edition, London: Pearson Education

Greene, W. H. (2008), The Econometric Approach to Efficiency Analysis, The Measurement of Productive Efficiency and Productivity Growth, pp. 92 - 250

Gregory, B. T., Rutherford, M. W., Oswald, S., & Gardiner, L. (2005), An Empirical Investigation of the Growth Cycle Theory of Small Firm Financing, Journal of Small Business Management, Vol. 43 (4), pp. 382 – 392

Gregory, H. J. (2002), Comparative Matrix of Corporate Governance Codes Relevant to the European Union and its Member States, Weil Gotshal and Manges LLP



Grossman, W., & Hoskisson, R. E. (1998), CEO Pay at the Crossroads of Wall Street and Main: Toward the Strategic Design of Executive Compensation, The Academy of Management Executive, Vol. 12 (1), pp. 43-57

Grout, P. A., & Stevens, M. (2003), The Assessment: Financing and Managing Public Services, Oxford Review of Economic Policy, Vol. 19 (2), pp. 215 - 234

Guariglia, A. (1999), The Effects of Financial Constraints on Inventory Investment: Evidence from a Panel of UK Firms, Economica, Vol. 66 (1), pp. 43 – 62

Guariglia, A. (2000), Inventory Investment and Capital Market Imperfections: A Generalization of the Linear Quadratic Inventory Model, Oxford Bulletin of Economics and Statistics, Vol. 62 (2), pp. 223 – 242

Guariglia, A. (2008), Internal Financial Constraints, External Financial Constraints, and Investment Choice: Evidence from a Panel of UK firms, Journal of Banking & Finance, Vol. 32 (9), pp. 1795 – 1809

Guest, P. M. (2008), The Determinants of Board Size and Composition: Evidence from the UK, Journal of Corporate Finance, Vol. 14 (1), pp. 51 – 72

Gujarati, D. N. (2003), Basic Econometrics, 4th Edition, New York, NY: McGraw Hill

Gulbrandsen, T. (2009), Family Businesses and Trade Unions in Norway, Economic and Industrial Democracy, Vol. 30 (4), pp. 592 – 613

Gutterman, A. S. (2019), Corporate Governance: An Introduction to Theory and Practice, London: Sustainable Entrepreneurship Project



Hair, J. F. (2009), Research Methods for Business, Austin, TX: Leyh Publishing

Hakim, C. (1987), Research Design: Strategies and Choices in the Design of Social Research

Hall, E. and Wall, K. (2019), Research Methods for Understanding Professional Learning, London: Bloomsbury Research Methods for Education, Bloomsbury

Haniffa, R. M., & Cooke, T. E. (2002), Culture, Corporate Governance and Disclosure in Malaysian Corporations, Abacus, Vol. 38 (3), pp. 317 – 349

Haniffa, R., & Hudaib, M. (2006), Corporate Governance Structure and Performance of Malaysian Listed Companies, Journal of Business Finance & Accounting, Vol. 33 (7), pp. 1034 – 1062

Hansmann, H., & Kraakman, R. (2000), End of History for Corporate Law, The Law Journal, Vol. 89, pp. 439 – 461

Harris, D., & Helfat, C. E. (1998), CEO Duality, Succession, Capabilities and Agency Theory: Commentary and Research Agenda, Strategic Management Journal, Vol. 19 (9), pp. 901 – 904

Harris, M., & Raviv, A. (1988), Corporate Governance: Voting Rights and Majority Rules, Journal of Financial Economics, Vol. 20, pp. 203 – 235

Harrison, J. R. (1987), The Strategic Use of Corporate Board Committees, California Management Review, Vol. 30 (1), pp. 109 – 125

Harrison, J. R., Schiantarelli, F., & Siregar, M. G. (1994), The Effect of Financial Liberalization on the Capital Structure and Investment Decisions of Indonesian Manufacturing Establishments, The World Bank Economic Review, Vol. 8 (1), pp. 17 – 47



Hart, O. (1995), Firms, Contracts and Financial Structure, Oxford: Oxford University Press

Hermalin, B. E., & Weisbach, M. S. (1988), The Determinants of Board Composition, The RAND Journal of Economics, pp. 589 – 606

Hermalin, B. E., & Weisbach, M. S. (1991), The Effects of Board Composition and Direct Incentives on Firm Performance, Financial Management, pp. 101 – 112

Hermalin, B. E., & Weisbach, M. S. (1998), Boards of Directors as an Endogenously Determined Institution: A Survey of the Economic Literature, National Bureau of Economic Research

Higgs, D., & Britain, G. (2003), Review of the Role and Effectiveness of Non-Executive Directors, London: HMSO Stationery Office

Hillman, A. J., & Dalziel, T. (2003), Boards of Directors and Firm Performance: Integrating Agency and Resource Dependence Perspectives, Academy of Management Review, Vol. 28 (3), pp. 383 - 396

Hillman, A. J., Cannella, A. A., & Paetzold, R. L. (2000), The Resource Dependence Role of Corporate Directors: Strategic Adaptation of Board Composition in Response to Environmental Change, Journal of Management Studies, Vol. 37 (2), pp. 235 – 256

Hiraki, T., Inoue, H., Ito, A., Kuroki, F., & Masuda, H. (2003), Corporate Governance and Firm Value in Japan: Evidence from 1985 to 1998, Pacific-Basin Finance Journal, Vol. 11 (3), pp. 239 – 265

Hitt, M. A., Dacin, M. T., Levitas, E., Arregle, J. L., & Borza, A. (2000), Partner Selection in Emerging and Developed Market Contexts: Resource-Based and Organizational Learning Perspectives, Academy of Management Journal, Vol. 43 (3), pp. 449 – 467



Ho, C. K. (2005), Corporate Governance and Corporate Competitiveness: An International Analysis, Corporate Governance: An International Review, Vol. 13 (2), pp. 211 – 253

Ho, C. A., & Williams, S. M. (2003), International Comparative Analysis of the Association between Board Structure and the Efficiency of Value Added By a Firm from its Physical Capital and Intellectual Capital Resources, The International Journal of Accounting, Vol. 38 (4), pp. 465 – 491

Hsu, W. H. L., Wang, G. Y., & Hsu, Y. P. (2010), Testing Mediator and Moderator Effects of Independent Director on Firm Performance

Hussey, J., & Hussey, R. (2009), Business Research: A Practical Guide for Undergraduate and Postgraduate Students, Basingstoke: MacMillan Press

Ingley, C., & Van Der Walt, N. (2002), Board Dynamics and the Politics of Appraisal, Corporate Governance, An International Review, Vol. 10 (3), pp. 163 – 174

Ingley, C., & Van der Walt, N. (2003), Board Configuration: Building Better Boards, Corporate Governance, Vol. 3 (4), pp. 5 – 17

International Labour Office (2013), Independent Evaluation of the ILO's Strategy to Promote Decent Work in the Arab Region: A Cluster Evaluation of Jordan, Lebanon and the Occupied Palestinian Territory, Accessed on 2nd March 2018, Available online at

http://www.ilo.org/wcmsp5/groups/public/ed_mas/eval/documents/publication/wcms_226356.pdf.

Ittner, C. D., & Larcker, D. F. (2003), Coming Up Short on Nonfinancial Performance Measurement, Harvard Business Review, Vol. 81 (11), pp. 88 - 95



Jacoby, G., Liu, M., Wang, Y., Wu, Z. and Zhang, Y. (2019), Corporate Governance, External Control, and Environmental Information Transparency: Evidence from Emerging Markets, Journal of International Finance, Markets, Institutional Money, Vol. 58, pp. 269 – 283

Jensen, M. C. (1986), Agency Costs of Free Cash Flow, Corporate Finance and Takeovers, The American Economic Review, Vol. 76 (2), pp. 323 – 329

Jensen, M. C. (1993), The Modern Industrial Revolution, Exit, and the Failure of Internal Control Systems, The Journal of Finance, Vol. 48 (3), pp. 831 – 880

Jensen, M. C., & Meckling, W. H. (1976), Theory of the Firm: Managerial Behaviour, Agency Costs and Ownership Structure, Journal of Financial Economics, Vol. 3 (4), pp. 305 – 360

Jensen, M. C., & Murphy, K. J. (1990), Performance Pay and Top-Management Incentives: Division of Research, Harvard Business School

Jiraporn, P., Singh, M., & Lee, C. I. (2009), Ineffective Corporate Governance: Director Busyness and Board Committee Memberships, Journal of Banking & Finance, Vol. 33 (5), pp. 819 – 828

Joh, S. W. (2003), Corporate Governance and Firm Profitability: Evidence from Korea before the Economic Crisis, Journal of Financial Economics, Vol. 68 (2), pp. 287 – 322

Johnson, J. L., Daily, C. M., & Ellstrand, A. E. (1996), Boards of Directors: A Review and Research Agenda, Journal of Management, Vol. 22 (3), pp. 409 – 438

Johnson, S., & Mitton, T. (2003), Cronyism and Capital Controls: Evidence from Malaysia, Journal of Financial Economics, Vol. 67 (2), pp. 351 – 382



Jordan Companies Law (1997), Accessed on 17th July 2018, Available online at

http://www.mit.gov.jo/portals/0/tabid/502/Companies%20Law.aspx.

Jordan Corporate Governance Code (2012), Accessed on 12th April 2017, Available online at www.ccd.gov.jo/uploads/CG%20Code%English.pdf.

Jordan Investment Board (2013), Industrial Zones, Accessed on 15th April 2017, Available online at

http://www.jordaninvestment.com/BusinessandInvestment/WheretoInvest/IndustrialZ.

Jordan Media and Advertising (2013), Accessed on 12th April 2017, Available online at

http://www.irex.org/system/files/MSIMENA09_Jordan.pdf.

Jordanian Ministry of Energy and Mineral Resources (2013), Accessed on 2nd April 2018, Available online at

http://www.memr.gov.jo/Default.aspx?tabid=242

Jordanian Ministry of Industry and Trade (2017), Accessed on 7th June 2017, Available online at

www.mit.gov.jo/tabid/36/default.aspx.

Jordanian Ministry of Transport, Accessed on 19 November 2017, Available online at

http://www.mot.gov.jo/en

Jordan Securities Commission (2017), Accessed on 18th April 2017, Available online at www.jsc.gov.jo.

Jordan Securities Commission (2020a), A Brief History, Accessed on 12 June 2020, Available at

https://jsc.gov.jo/page/en/a_brief_history



Jordan Securities Commission (2020b), Decisions of the board of directors, Accessed on 12 June 2020, Available at

https://www.jsc.gov.jo/NewsGroup/en/board_decisions

Jose, M. L., Lancaster, C., & Stevens, J. L. (1996), Corporate Returns and Cash Conversion Cycles, Journal of Economics and Finance, Vol. 20 (1), pp. 33 – 46

Kabariti, N. (2019), Jordan "A Safe Haven" of Opportunities: Kabariti, Alghad, Accessed on 17 July 2020, Available at https://alghad.com/jordan-a-safe-haven-of-opportunities-kabariti/

Kakabadse, A., Kakabadse, N. K., & Kouzmin, A. (2003), Reinventing the Democratic Governance Project through Information Technology?, A Growing Agenda for Debate, Public Administration Review, Vol. 63 (1), pp. 44 – 60

Kang, J.-K., & Stulz, R. (2008), Why Is there a Home Bias? An Analysis of Foreign Portfolio Equity Ownership in Japan, Journal of Financial Economics, Vol. 46 (1), pp. 3 - 28

Kaplan, S. N., & Minton, B. A. (1994), Appointments of Outsiders to Japanese Boards: Determinants and Implications for Managers, Journal of Financial Economics, Vol. 36 (2), pp. 225 – 258

Kaplan, S. N., & Zingales, L. (1997), Do Investment Cash Flow Sensitivities Provide Useful Measures of Financing Constraints?, The Quarterly Journal of Economics, pp. 169 – 215

Kapopoulos, P., & Lazaretou, S. (2007), Corporate Ownership Structure and Firm Performance: Evidence from Greek Firms, Corporate Governance: An International Review, Vol. 15 (2), pp. 144 – 158



Karamanou, I., & Vafeas, N. (2005), The Association between Corporate Boards, Audit Committees, and Management Earnings Forecasts: An Empirical Analysis, Journal of Accounting Research, Vol. 43 (3), pp. 453 – 486

Kashyap, A. K., Stein, J. C., & Lamont, O. (1994), Credit Conditions and the Cyclical Behavior of Inventories, Quarterly Journal of Economics, pp. 109

Kaymak, T., & Bektas, E. (2008), East Meets West? Board Characteristics in an Emerging Market: Evidence from Turkish Banks, Corporate Governance An International Review, Vol. 16 (6), pp. 550 – 561

Kearney, C. (2005), Emerging markets research: Trends, Issues and Future Directions, Emerging Markets Review, Vol. 13 (2), pp. 159 – 183

Khan, A., & Awan, S. (2012), Effect of Board Composition on Firm's Performance: A Case of Pakistani Listed Companies, Interdisciplinary Journal of Contemporary Research in Business, Vol. 3 (10), pp. 853 – 863

Khanna, T., & Palepu, K. (2000), Is Group Affiliation Profitable in Emerging Markets? An Analysis of Diversified Indian Business Groups, The Journal of Finance, Vol. 55 (2), pp. 867 – 891

Kholeif, A. (2008), CEO Duality and Accounting-Based Performance in Egyptian Listed Companies: A Re-Examination of Agency Theory Predictions, Research in Accounting in Emerging Economies, Vol. 8, pp. 65 – 96

Kiel, G. Nicholson. G. (2003), Board Composition and Corporate Performance: How the Australian Experience Informs Contrasting Theories of Corporate Governance, Corporate Governance An International Review, Vol. 11 (3), pp. 189 – 205



Kim, E. (2006), The Impact of Family Ownership and Capital Structures on Productivity Performance of Korean Manufacturing Firms, Corporate Governance and The Chaebol Problem, Journal of the Japanese and International Economies, Vol. 20 (2), pp. 209 – 233

Kirkpatrick, C., Parker, D., & Zhang, Y. F. (2006), Foreign Direct Investment in Infrastructure in Developing Countries: Does Regulation Make a Difference?, Transnational Corporations, Vol. 15 (1), pp. 143

Klapper, L. F., & Love, I. (2004), Corporate Governance, Investor Protection, and Performance in Emerging Markets, Journal of Corporate Finance, Vol. 10 (5), pp. 703 – 728

Klein, A. (1998), Firm Performance and Board Committee Structure, The Journal of Law and Economics, Vol. 41 (1), pp. 275 – 304

Kohler, U., & Kreuter, F. (2005), Data Analysis Using Stata, College Station, TX: Stata Press

Kollewe, J. (2019), Accountancy Profession 'Complicit in Thomas Cook Failure', The Guardian, 22 October 2019, Accessed on 29 April 2020, Available at

https://www.theguardian.com/business/2019/oct/22/accountancy-profession-complicit-in-thomas-cook-failure

Korajczyk, R. A., & Levy, A. (2003), Capital Structure Choice: Macroeconomic Conditions and Financial Constraints, Journal of Financial Economics, Vol. 68 (1), pp. 75 – 109

Kounouwewa, J., & Chao, D. (2011), Financing Constraints Determinants in African Countries, The International Journal of Applied Economics and Finance, Vol. 5 (1), pp. 30 – 45



Krivogorsky, V. (2006), Ownership, Board Structure, and Performance in Continental Europe, The International Journal of Accounting, Vol. 41 (2), pp. 176 – 197

Kumar, J. (2004), Agency Theory and Firm Value in India, Does Ownership Structure Influence Value

Kumar, N., & Singh, J. (2012), Outside Directors, Corporate Governance and Firm Performance: Empirical Evidence from India, Asian Journal of Finance & Accounting, Vol. 4 (2), pp. 39 – 55

La Porta, R., Lopez de Silanes, F., Shleifer, A., & Vishny, R. (1999), Investor Protection and Corporate Valuation, Nber Working Paper Series, 7403

Lamont, O., Polk, C., & Saaá-Requejo, J. (2001), Financial Constraints and Stock Returns, Review of Financial Studies, Vol. 14 (2), pp. 529 – 554

Lang, M. H., Lins, K. V., & Miller, D. P. (2004), Concentrated Control, Analyst Following, and Valuation: Do Analysts Matter Most when Investors are Protected Least?, Journal of Accounting Research, Vol. 42 (3), pp. 589 – 623

Larmou, S., & Vafeas, N. (2010), The Relation between Board Size and Firm Performance in Firms with a History of Poor Operating Performance, Journal of Management & Governance, Vol. 14 (1), pp. 61 – 85

Lasfer, M. A. (2006), The Interrelationship between Managerial Ownership and Board Structure, Journal of Business Finance & Accounting, Vol. 33 (7), pp. 1006 – 1033

Lawrence, J., & Stapledon, G. (1999), Do Independent Directors Add Value?, Centre for Corporate Law and Securities Regulation, University of Melbourne



Lehn, K. M., Patro, S., & Zhao, M. (2009), Determinants of the Size and Composition of US Corporate Boards 1935-2000, Financial Management, Vol. 38 (4), pp. 747 – 780

Letza, S., Sun, X., & Kirkbride, J. (2004), Shareholding Versus Stakeholding: A Critical Review of Corporate Governance, Corporate Governance An International Review, Vol. 12 (3), pp. 242 – 262

Leuz, C., Lins, K. V., & Warnock, F. E. (2010), Do Foreigners Invest Less in Poorly Governed Firms?, Review of Financial Studies, Vol. 23 (3), pp. 3245 – 3285

Lev, B., & Sunder, S. (1979), Methodological Issues in the Use of Financial Ratios, Journal of Accounting and Economics, Vol. 1 (3), pp. 187 – 210

Li, D. (2011), Financial Constraints, R&D Investment and Stock Returns, Review of Financial Studies, Vol. 24 (9), pp. 2974 – 3007

Lim, S., Matolcsy, Z., & Chow, D. (2007), The Association between Board Composition and Different Types of Voluntary Disclosure, European Accounting Review, Vol. 16 (3), pp. 555 – 583

Linck, J. S., Netter, J. M., & Yang, T. (2008), The Determinants of Board Structure, Journal of Financial Economics, Vol. 87 (2), pp. 308 – 328

Lipton, M., & Lorsch, J. W. (1992), A Modest Proposal for Improved Corporate Governance, The Business Lawyer, pp. 59 – 77

Liu, H., & Fong, M. W. (2010), Board Characteristics of Medium and Large Chinese Companies, Corporate Governance, Vol. 10 (2), pp. 163 – 175

Livdan, D., Sapriza, H., & Zhang, L. (2009), Financially Constrained Stock Returns, The Journal of Finance, Vol. 64 (4), pp. 1827 – 1862



Lund, M., & Wright, J. (1999), The Financing of Small Firms in the United Kingdom, Bank of England, Quarterly Bulletin, Vol. 39 (2), pp. 195

Luo, Y. (2007), Global Dimensions of Corporate Governance, Malden, MA: Blackwell Pub

MacAvoy, P. W., & Millstein, I. M. (2004), The Recurrent Crisis in Corporate Governance, Palo Alto, CA: Stanford University Press

MacAulay, K., Dutta, S., Oxner, M. and Hynes, T. (2020), The Impact of a Change in Corporate Governance Regulations on Firms in Canada, Journal of Finance and Accounting, Vol. 48 (4), pp. 29 – 52

Machold, S., Huse, M., Minichilli, A., & Nordqvist, M. (2011), Board Leadership and Strategy Involvement in Small Firms: A Team Production Approach, Corporate Governance An International Review, Vol. 19 (4), pp. 368 – 383

Main, B. G., & Johnston, J. (1993), Remuneration Committees and Corporate Governance, Accounting and Business Research, Vol. 23 (1), pp. 351 – 362

Mallette, P., & Fowler, K. L. (1992), Effects of Board Composition and Stock Ownership on the Adoption of Poison Pills, Academy of Management Journal, Vol. 35 (5), pp. 1010 – 1035

Mallin, C. A. (2014), Corporate Governance, 4th Edition, Oxford: Oxford University Press

Mallin, C. A. (2019), Corporate Governance, 6th edition, Oxford University Press, Oxford



Mandacı, P., & Gumus, G. (2010), Ownership Concentration, Managerial Ownership and Firm Performance: Evidence from Turkey, South East European Journal of Economics and Business, Vol. 5 (1), pp. 57 – 66

Mangena, M., & Chamisa, E. (2008), Corporate Governance and Incidences of Listing Suspension by the JSE Securities Exchange of South Africa: An Empirical Analysis, The International Journal of Accounting, Vol. 43 (1), pp. 28 – 44

Mangena, M., & Tauringana, V. (2007), Disclosure, Corporate Governance and Foreign Share Ownership on the Zimbabwe Stock Exchange, Journal of International Financial Management & Accounting, Vol. 18 (2), pp. 53 – 85

Mangena, M., Tauringana, V., & Chamisa, E. (2008), Corporate Boards, Ownership Structure and Firm Performance in an Environment of Severe Political and Economic Uncertainty: Bradford University, School of Management

Mansur, Y. (2008), Overcoming Barriers to Foreign Direct Investment in Jordan, International Research Foundation of Oman Discussion Paper

Markets, C. E. (2000), The Tide's Gone Out: Who's Swimming Naked, Credit Lyonnais Securities Asia, Hong Kong

McKinsey & Company (2002), Global Investor Opinion Survey: Key Findings, McKinsey & Company

McVey, H., & Draho, J. (2005), US Family-Run Companies They May Be Better Than You Think, Journal of Applied Corporate Finance, Vol. 17 (4), pp. 134 – 143

Meek, G. K., Roberts, C. B., & Gray, S. J. (1995), Factors Influencing Voluntary Annual Report Disclosures by US, UK and Continental European



Multinational Corporations, Journal of International Business Studies, pp. 555 – 572

Megginson, W. L., & Netter, J. M. (2001), From State to Market: A Survey of Empirical Studies on Privatization, Journal of Economic Literature, pp. 321 – 389

Mehran, H. (1995), Executive Compensation Structure, Ownership, and Firm Performance, Journal of Financial Economics, Vol. 38 (2), pp. 163 – 184

Melville, A. (2019), International Financial Reporting: A Practical Guide, 7th Edition, Pearson, London

Mkheimer, I. M. (2018), Corporate Governance in Jordan and Boardroom Diversity: A Critical Review of Literature, European Scientific Journal, Vol 4 (10), pp. 1857 - 1878

Miller, D., Le Breton-Miller, I., Lester, R. H., & Cannella Jr, A. A. (2007), Are Family Firms Really Superior Performers?, Journal of Corporate Finance, Vol. 13 (5), pp. 829 – 858

Miller-Millesen, J. L. (2003), Understanding the Behaviour of Non-Profit Boards of Directors: A Theory-Based Approach, Non-profit and Voluntary Sector Quarterly, Vol. 32 (4), pp. 521 – 547

Mintzberg, H. (1983), Power in and Around Organizations, Englewood Cliffs, NJ: Prentice-Hall

Mitton, T. (2002), A Cross-Firm Analysis of the Impact of Corporate Governance on the East Asian Financial Crisis, Journal of Financial Economics, Vol. 64 (2), pp. 215 – 241

Modigliani, F., & Miller, M. H. (1958), The Cost of Capital, Corporation Finance and the Theory of Investment, The American, pp. 1-3



Modigliani, F., & Miller, M. H. (1963), Corporate Income Taxes and the Cost of Capital: A Correction, The American Economic Review, Vol. 53 (3), pp. 433 – 443

Mohamed, S. E., & Sidiropoulos, M. G. (2010), Another Look at the Determinants of Foreign Direct Investment in MENA Countries: An Empirical Investigation, Journal of Economic Development, Vol. 35 (2), pp. 75 – 95

Monks, R. A. (2001), Redesigning Corporate Governance Structures and Systems for the Twenty First Century, Corporate Governance An International Review, Vol. 9 (3), pp. 142 – 147

Monks, R. A., & Minnow, N. (1995), Corporate Governance for the 21st Century: Watching the Watchers, Cambridge, MA: Blackwell Publishers

Morck, R., Shleifer, A., & Vishny, R. W. (1988), Management Ownership and Market Valuation: An Empirical Analysis, Journal of Financial Economics, Vol. 20, pp. 293 – 315

Morey, M., Gottesman, A., Baker, E., & Godridge, B. (2009), Does Better Corporate Governance Result in Higher Valuations in Emerging Markets?, Another Examination Using a New Data Set, Journal of Banking & Finance, Vol. 33 (2), pp. 254 – 262

Musso, P., & Schiavo, S. (2008), The Impact of Financial Constraints on Firm Survival and Growth, Journal of Evolutionary Economics, Vol. 18 (2), pp. 135 – 149

Muth, M., & Donaldson, L. (1998), Stewardship Theory and Board Structure: A Contingency Approach, Corporate Governance An International Review, Vol. 6 (1), pp. 5 – 28

Myers, S. C. (1977), Determinants of Corporate Borrowing, Journal of Financial Economics, Vol. 5 (2), pp. 147 – 175



Najid, N. A., & Abdul Rahman, R. (2011), Government Ownership and Performance of Malaysian. Government-Linked Companies, International Research Journal of Finance and Economics, Vol. 61, pp. 42 – 56

Nenova, T. (2003), The Value of Corporate Voting Rights and Control: A Cross-Country Analysis, Journal of Financial Economics, Vol. 68 (3), pp. 325 – 351

Nenova, T. (2009), A Corporate Governance Agenda for Developing Countries, Contaduría Y Administración, 217

Nicholson, G. J., & Kiel, G. C. (2007), Can Directors Impact Performance? A Case-Based Test of Three Theories of Corporate Governance, Corporate Governance An International Review, Vol. 15 (4), pp. 585 – 608

Nwibo, S. U., & Okorie, A. (2013), Constraints To Entrepreneurship And Investment Decisions Among Agribusiness Investors In Southeast, Nigeria, International Journal of Small Business and Entrepreneurship Research, Vol. 1 (4), pp. 30 – 42

Oman, C. (2001), Corporate Governance and National Development

Oman, C., Fries, S., & Buiter, W. (2004), Corporate Governance in Developing, Transition and Emerging-market Economies, OECD Publishing

Orden, O., & Garmendia, A. (2005), Does it Matter Ownership Structure? Performance in Spanish Companies, Journal of European Financial Management, pp. 1 – 40

Organisation for Economic Cooperation and Development (1999), Principles of Corporate Governance, OECD Publishing

Organisation for Economic Cooperation and Development (2004), Principles of Corporate Governance, OECD Publishing



Organisation for Economic Cooperation and Development (2006), Principles of Corporate Governance, OECD Publishing

Organisation for Economic Cooperation and Development (2013), Jordan – Investment Policy Review – OECD, Accessed on 20 March 2018, Available online at

http://www.oecd.org/countries/jordan/jordan-investment-policy.htm.

Oxelheim, L., & Randøy, T. (2003), The Impact of Foreign Board Membership on Firm Value, Journal of Banking & Finance, Vol. 27 (12), pp. 2369 – 2392

Padgett, C., & Shabbir, A. (2005), The UK Code of Corporate Governance: Link between Compliance and Firm Performance, ICMA Centre Discussion Papers in Finance DP 17

Pallant, J. (2020), SPSS Survival Manual: A Step by Step Guide to Data Analysis Using IBM SPSS, 7th edition, London: Open University Press

Patibandla, M. (2006), Equity Pattern, Corporate Governance and Performance: A Study of India's Corporate Sector, Journal of Economic Behaviour & Organization, Vol. 59 (1), pp. 29 – 44

Pearce, J. A., & Zahra, S. A. (1992), Board Composition from a Strategic Contingency Perspective, Journal of Management Studies, Vol. 29 (4), pp. 411 – 438

Peng, M. W., Li, Y., Xie, E., & Su, Z. (2010), CEO Duality, Organizational Slack, and Firm Performance in China, Asia Pacific Journal of Management, Vol. 27 (4), pp. 611 – 624

Peoples, K. (2020), How to Write a Phenomenological Dissertation: A Stepby-Step Guide, Qualitative Research Methods, London: Sage



Pfeffer, J. (1972), Size and Composition of Corporate Boards of Directors: The Organization and Its Environment, Administrative Science Quarterly, pp. 218 – 228

Pfeffer, J. (1973), Size, Composition, and Function of Hospital Boards of Directors: A Study of Organization-Environment Linkage, Administrative Science Quarterly, pp. 349 – 364

Pfeffer, J., & Salancik, G. R. (1978), The External Control of Organizations: A Resource Dependence Approach, New York, NY: Harper and Row Publishers

Pike, R. and Neale, B. (2009), Corporate Finance and Investment: Decision and Strategies, 6th Edition, London: FT Prentice Hall

Pike, R., Neale, B., Akbar, S and Linsley, P. (2018), Corporate Finance and Investment: Decisions and Strategies, 9th edition, London: Pearson

Pradhan, J. P. (2011), Firm Performance during Global Economic Slowdown: A View from India, Economics, Management, and Financial Markets, Vol. 1, pp. 73 – 97

Punch, K. (1998), Introduction to Social Research: Quantitative and Qualitative Methods, London: Sage

Qurashi, M. H. (2018), Corporate Governance Code Comparison for South Asian Emerging Economies, International Journal of Law and Management, Vol. 60 (2), pp. 250 – 266, Accessed on 20 June 2020, Available at https://doi.org/10.1108/IJLMA-05-2017-0115

Raheja, C. G. (2005), Determinants of Board Size and Composition: A Theory of Corporate Boards, Journal of Financial and Quantitative Analysis, Vol. 40 (2), pp. 283 – 306



Rajagopalan, N., & Zhang, Y. (2008), Corporate Governance Reforms in China and India: Challenges and Opportunities, Business Horizons, Vol. 51 (1), pp. 55 – 64

Razzaque, R. M. R., Ali, M. J. and Mather, P. (2020), Corporate Governance Reform and Family Firms: Evidence from an Emerging Economy, Pacific-Basin Finance Journal, Vol. 59, pp. 251 – 260

Rechner, P. L., & Dalton, D. R. (1991), CEO Duality and Organizational Performance: A Longitudinal Analysis, Strategic Management Journal, Vol. 12 (2), pp. 155 – 160

Reed, D. (2002), Corporate Governance Reforms in Developing Countries, Journal of Business Ethics, Vol. 37 (3), pp. 223 – 247

Roe, M. J. (2003), Political Determinants of Corporate Governance: Political Context, Corporate Impact, Oxford: Oxford University Press

ROSC (2004), Report on the Observance of Standards and Cods, Corporate Governance Country Assessment, Jordan, Accessed on 23rd October 2018, Available online at www.worldbank.org/ifa/jor rosc cg.pdf

Ross, S. A., Westerfield, R., & Jordan, B. D. (2008), Fundamentals of Corporate Finance, New York, NY: McGraw-Hill Education

Rosser, A. (2003), Coalitions, Convergence and Corporate Governance Reform in Indonesia, Third World Quarterly, Vol; 24 (2), pp. 319 – 337

Sanda, A., Mikailu, A. S., & Garba, T. (2005), Corporate Governance Mechanisms and Firm Financial Performance in Nigeria, African Economic Research Consortium

Saunders, M., Lewis, P., & Thornhill, A. (2016), Research Methods for Business Students, 6th Edition, London: Pearson Education



Saunders, M., Lewis, P. and Thornhill, A. (2019), Research Methods for Business Students, 8th edition, London: Pearson

Schulze, W. S., Lubatkin, M. H., & Dino, R. N. (2003), Exploring the Agency Consequences of Ownership Dispersion among the Directors of Private Family Firms, Academy of Management Journal, Vol. 46 (2), pp. 179 – 194

Securities Depository Centre (2020), Establishment and Responsibilities, Accessed on 15 June 2020, Available at https://www.sdc.com.jo/english/index.php?option=com_content&task=view&id=35&Itemid=62

Serrasqueiro, Z. S., & Nunes, P. M. (2008), Performance and Size: Empirical Evidence from Portuguese SMEs, Small Business Economics, Vol. 31 (2), pp. 195 – 217

Shabbir, A. and Padgett, C. (2008), The UK Code of Corporate Governance: Link Between Compliance and Firm Performance, Research Paper Series, 02/08, Cranfield University School of Management

Shanikat, M., & Abbadi, S. S. (2011), Assessment of Corporate Governance in Jordan: An Empirical Study, Australasian Accounting Business and Finance Journal, Vol. 5 (3), pp. 93 – 106

Sharar, Z, (2006), A Comparative Analysis of the Corporate Governance Legislative Frameworks in Australia and Jordan Measured against the OECD Principles of Corporate Governance 2004 as an International Benchmark, PhD dissertation, Queensland, Australia: Bond University

Shehata, N. F. (2015), Development of Corporate Governance Codes in the GCC: An Overview Corporate Governance, Vol. 15 (3), pp. 315 – 338



Sheikh, N. A., & Wang, Z. (2012), Effects of Corporate Governance on Capital Structure: Empirical Evidence from Pakistan, Corporate Governance, Vol. 12 (5), pp. 629 – 641

Sheu, H. J., & Yang, C. Y. (2005), Insider Ownership and Firm Performance in Taiwan's Electronics Industry: A Technical Efficiency Perspective, Managerial and Decision Economics, Vol. 26 (5), pp. 307 – 318

Shu, P. G. and Chiang, Sue-Jane, (2020), The Impact of Corporate Governance on Corporate Social Performance: Cases from Listed Firms in Taiwan, Pacific-Basin Finance Journal, Vol. 61, pp. 1 – 20

Singh, A. (2003), Corporate Governance, Corporate Finance and Stock Markets in Emerging Countries, Journal of Corporate Studies, pp. 3 – 41

Singh, A., & Zammit, A. (2006), Corporate Governance, Crony Capitalism and Economic Crises: Should the US Business Model Replace the Asian Way of Doing Business?, Corporate Governance An International Review, Vol. 14 (4), pp. 220 – 233

Solomon, J. (2014), Corporate Governance and Accountability, London: Wiley and Sons

Sonnenfeld, J. A. (2002), What Makes Great Boards Great, Harvard Business Review, Vol. 80 (9), pp. 106 – 113

Spaliara, M.-E. (2011), Financial Frictions and the K/L Ratio in UK Manufacturing, Economics Letters, Vol. 112 (1), pp. 23 – 25

Sternberg, E. (2004), Corporate Governance: Accountability in the Market Place, London: The Institute of Economic Affairs

Stulz, R. M. (2005), The Limits of Financial Globalization, The Journal of Finance, Vol. 60 (4), pp. 1595 – 1638



Sulong, Z., & Nor, F. M. (2010), Corporate Governance Mechanisms and Firm Valuation in Malaysian Listed Firms: A Panel Data Analysis, Journal of Modern Accounting and Auditing, Vol. 6 (1), pp. 1 – 18

Sun, J., & Cahan, S. (2009), The Effect of Compensation Committee Quality on the Association between CEO Cash Compensation and Accounting Performance, Corporate Governance An International Review, Vol. 17 (2), pp. 193 – 207

Suto, M. (2003), Capital Structure and Investment Behaviour of Malaysian Firms in the 1990s: A Study of Corporate Governance before the Crises, An International Review, Vol. 11(1), pp. 25 – 39

Tam, O. K., & Tan, M. G. S. (2007), Ownership, Governance and Firm Performance in Malaysia, Corporate Governance An International Review, Vol. 15 (2), pp. 208 – 222

Tan, H., Wang, S., & Welker, M. (2011), Analyst Following and Forecast Accuracy After Mandated IFRS Adoptions, Journal of Accounting Research, Vol. 49 (5), pp. 1307 – 1357

Tariff, J. (2006), Corporate Governance in the Middle East and North Africa (MENA) Region, Arab Bank Review, Vol. 8 (1), pp. 31 – 36

Taufil-Mohd, K. N., Md-Rus, R., & Musallam, S. R. (2013), The Effect of Ownership Structure on Firm Performance in Malaysia, International Journal of Finance and Accounting, Vol. 2 (2), pp. 75 – 81

The Sarbanes-Oxley Act (2002), Sarbanes-Oxley Act USA, Accessed on 2nd March 2018, Available online at

http://fl1.findlaw.com/news.findlaw.com/cnn/docs/gwbush/sarbanesoxley072 302.pdf



The World Bank (2020), GDP per capita (current US\$) – Jordan, Accessed on 23 June 2020, Available on

https://data.worldbank.org/indicator/NY.GDP.PCAP.CD?locations=JO

Topak, M. (2011), The Effect of Board Size on Firm Performance: Evidence From Turkey, Middle Eastern Finance and Economics, Vol. 14, pp.119 – 127

Townsend, K. and Saunders, M. (2018), How to Keep Your Research Project on Track: Insights from When Things Go Wrong, Cheltenham, Glocs.: Edward Elgar Publishing Ltd

Tricker, B. (2019), Corporate Governance: Principles, Policies and Practices, 4th edition, Oxford: Oxford University Press

Tsamenyi, M., Enninful-Adu, E., & Onumah, J. (2007), Disclosure and Corporate Governance in Developing Countries: Evidence from Ghana, Managerial Auditing Journal, Vol. 22 (3), pp. 319 – 334

United Nations Development Programme (2013), United Nations Development Programme in Jordan, Accessed on 19th November 2018, Available online at

http://www.jo.undp.org/jordan/en/home.html.

Van den Berghe, L., & Levrau, A. (2004), Evaluating Boards of Directors: What Constitutes a Good Corporate Board?, Corporate Governance An International Review, Vol. 12 (4), pp. 461 – 478

Villalonga, B., & Amit, R. (2006), How do Family Ownership, Control and Management Affect Firm Value?, Journal of Financial Economics, Vol. 80 (2), pp. 385 – 417

Walker, D. (2005), Restoring Trust after Recent Accountability Failures, Governing the Corporation: Regulation and Corporate Governance in an Age of Scandal and Global Markets, pp. 21 – 34



Watson, D. and Head, A. (2016), Corporate Finance: Principles and Practice, 7th Edition, London: Pearson

Weir, C., Laing, D., & McKnight, P. J. (2002), Internal and External Governance Mechanisms: Their Impact on the Performance of Large UK Public Companies, Journal of Business Finance & Accounting, Vol. 29 (5), pp. 579 – 611

Wooldridge, J. M. (2002), Econometric Analysis of Cross Section and Panel Data, Cambridge, MA: The MIT Press

World Bank (2015), Jordan Overview, Accessed on 23rd March 2018, Available online at

http://www.worldbank.org/en/country/jordan/overview

Wulf, J. (2007), Authority, Risk, and Performance Incentives: Evidence from Division Manager Positions Inside Firms, The Journal of Industrial Economics, Vol. 55 (1), pp. 169 – 196

Yawson, A. (2006), Evaluating the Characteristics of Corporate Boards Associated with Layoff Decisions, Corporate Governance An International Review, Vol. 14 (2), pp. 75 – 84

Young, M. N., Peng, M. W., Ahlstrom, D., Bruton, G. D., & Jiang, Y. (2008), Corporate Governance in Emerging Economies: A Review of The Principal–Principal Perspective, Journal of Management Studies, Vol. 45 (1), pp. 196–220



Zu'ubi, F. (2013), The Process of Changing Jordan's Telecommunications Market, Accessed on 20th July 2018, Available online at http://www.oecd.org/sti/broadband/1806507.pdf



Appendix 1 Tables

Table 1 Descriptive statistics

	Description									
Variable	N	Range	Minimum	Maximum	Mean	S. Dev	Variance	Skewness	Kurtosis	
ROA	475	2.34	-1.96	0.39	0.02	0.137	0.019	-6.885	94.226	
ROE	475	11.72	-11.21	0.51	-0.005	0.569	0.323	-16.79	322.61	
TOQ	475	8.26	0.1	8.36	0.823	0.742	0.551	3.937	28.05	
SGH	475	10.56	-1	9.56	0.163	1.173	1.377	4.7	26.535	
FSE	475	8.63	12.68	21.31	17.09	1.5114	2.284	0.225	0.725	
CAPX	475	0.41	0	0.41	0.031	0.046	0.002	3.835	21.361	
LEE	475	0.9	0	0.9	0.28	0.284	0.081	0.696	-1.124	
R&D	475	0.04	0	0.04	0.002	0.006	0	4.068	17.82	
LIY	475	1026.9	0.01	1027	9.784	80.89	6542.7	11.971	145.07	
BSE	475	22	700	29	12.47	4.36	19.005	1.297	2.124	
NED	475	0.85	0.04	0.89	0.239	0.163	0.026	1.463	2.857	
CGI	475	4	4	8	4.442	0.897	0.804	2.847	8.429	



Table 2 Correlation

Variable	ROA	ROE	TOQ	SGH	FSE	CAPX	LEE	R&D	LIY	BSE	NED	CGI
ROA	-											
Sig												
ROE	0.36	-										
Sig	0.02											
TOQ	-0.11	-0.43	-									
Sig	0.15	0.00										
SGH	0.00	0.02	-0.04	-								
Sig	0.19	0.05	0.00									
FSE	0.28	0.21	-0.23	-0.10	-							
Sig	0.00	0.00	0.01	0.17								
CAPX	0.09	0.01	0.04	-0.04	0.09	-						
Sig	0.02	0.00	0.39	0.01	0.66							
LEE	-0.14	-0.11	0.19	0.05	-0.32	-0.05	-					
Sig	0.96	0.03	0.05	0.09	0.01	0.05						
R&D	0.17	0.07	0.10	-0.02	0.03	0.02	0.00	-				
Sig	0.05	0.76	0.38	0.05	0.00	0.02	0.00					
LIY	-0.03	0.00	0.06	0.01	-0.20	-0.05	0.15	-0.03	-			
Sig	0.32	0.17	0.07	0.58	0.64	0.97	0.19	0.79				



BSE	0.04	-0.02	0.03	-0.04	0.43	-0.01	-0.10	0.13	-0.07	-		
Sig	0.29	0.00	0.36	0.69	0.48	0.34	0.00	0.85	0.15			
NED	-0.11	-0.05	-0.02	0.09	-0.10	0.04	0.05	-0.02	-0.01	-0.19	-	
Sig	0.25	0.66	0.06	0.06	0.45	0.19	0.39	0.66	0.23	0.92		
CGI	-0.06	-0.04	-0.02	0.06	0.00	0.08	-0.02	0.02	-0.05	-0.03	0.82	-
Sig	0.10	0.97	0.00	0.05	0.75	0.00	0.66	0.02	0.00	0.39	0.03	



Table 3 Multiple regression results for ROA

Descrip	tion	Unstandardised B	Significant
Constant		-0.40	0.00
Sales Growth		0.01	0.03
Firm Size		0.03	0.00
Capital Expendit	ures	-0.18	0.17
Leverage		-0.03	0.26
Research & Dev	elopment	4.11	0.00
Liquidity		7.01	0.35
Board Size		0.00	0.01
Non-Executive D	Directors	-0.09	0.02
CEO Duality		0.38	0.00
Remuneration C	ommittee	0.68	0.02
Audit Committee	•	0.15	0.00
Nomination Com	nmittee	0.52	0.02
R	0.69	R Square	0.48
F statistics	9.245	Significant	0.00



Table 4 Multiple regression results for ROE

Desci	ription	Unstandardised B	Significant
	триоп	Onstandardised B	Olgrimodrit
Cons	stant	-1.44	0.00
Sales	Growth	0.02	0.03
Firm	Size	0.10	0.00
Capital Ex	penditures	-0.11	0.84
Leve	erage	-0.09	0.33
Research &	Development	7.81	0.03
Liqu	idity	0.00	0.27
Board	d Size	-0.02	0.00
Non-Executi	ive Directors	-0.20	0.02
CEO I	Duality	0.21	0.01
Remuneratio	n Committee	0.45	0.02
Audit Co	ommittee	0.57	0.00
Nomination	Committee	0.33	0.02
R	0.51	R Square	0.26
F statistics	4.676	Significant	0.00



Table 5 Multiple regression results for TOQ

Descript	tion	Unstandardised B	Significant		
Constant		2.682	0.000		
Sales Growth		-0.038	0.017		
Firm Size		0.133	0.000		
Capital Expenditu	res	-1.104	0.118		
Leverage		-0.330	0.007		
Research & Deve	lopment	10.933	0.035		
Liquidity		2.476	0.952		
Board Size		-0.024	0.004		
Non-Executive Dir	rectors	-0.097	0.016		
CEO Duality		0.598	0.000		
Remuneration Co	mmittee	0.625	0.018		
Audit Committee		0.348	0.000		
Nomination Comn	nittee	0.826	0.031		
R	0.58	R Square	0.370		
F statistics	6.962	Significant	0.000		



ProQuest Number: 28367255

All rights reserved

INFORMATION TO ALL USERS

The quality of this reproduction is dependent on the quality of the copy submitted.

In the unlikely event that the author did not send a complete manuscript and there are missing pages, these will be noted. Also, if material had to be removed, a note will indicate the deletion.



ProQuest 28367255

Published by ProQuest LLC (2021). Copyright of the Dissertation is held by the Author.

All Rights Reserved.

This work is protected against unauthorized copying under Title 17, United States Code Microform Edition © ProQuest LLC.

ProQuest LLC 789 East Eisenhower Parkway P.O. Box 1346 Ann Arbor, MI 48106 - 1346

